

Newfoundland & Labrador

COLLECTIVE AGREEMENT

BETWEEN

HIS MAJESTY THE KING IN RIGHT OF NEWFOUNDLAND AND LABRADOR (REPRESENTED HEREIN BY THE TREASURY BOARD)

ON BEHALF OF ITS MEMBER HOSPITALS AND AGENCIES AS

LISTED IN SCHEDULE "C"

AND

REGISTERED NURSES' UNION NEWFOUNDLAND AND LABRADOR

REPRESENTING THE REGISTERED AND UNREGISTERED NURSES EMPLOYED BY CERTAIN OF THE HOSPITALS

AND INSTITUTIONS IN THE PROVINCE LISTED IN SCHEDULE "C",

IN ACCORDANCE WITH THE RESPECTIVE CERTIFICATION ORDERS

SIGNED: AUGUST 4, 2023 EXPIRES: JUNE 30, 2026

BETWEEN		MAJESTY RADOR	THE	KING	IN	RIGHT	OF	NEWFOUNDLAND	AND
	repres	sented herein	by the	Treasury	y Boa	ırd		of the o	ne part

THIS AGREEMENT made this 4th day of August Anno Domini, Two Thousand and Twenty-Three

AND

REGISTERED NURSES' UNION NEWFOUNDLAND AND LABRADOR representing the registered and unregistered nurses employed by certain of the hospitals and institutions in the Province listed in Schedule "C" in accordance with the respective certification orders

of the other part;

THIS AGREEMENT WITNESSETH that for and in consideration of the premises and covenants, conditions, stipulations and provisos herein contained, the parties hereto agree as follows:

TABLE OF CONTENTS

(Alphabetical index at the end of collective agreement)

<u>Article No</u> .	<u>Title</u>	Page No.	
1	Purpose of Agreement	1	
2	Interpretations and Definitions	1	
3	Recognition	4	
4	Management Rights	5	
5	Employee Rights	5	
6	Union Representation	6	
7	Union Dues	7	
8	Hours of Work	8	
9	Overtime	15	
10	Standby	17	
11	Callback	18	
12	Escort Duty	18	
13	Strikes - Lockouts	20	
14	Dispute Resolution	21	
15	Arbitration	24	
16	Vacation	26	
17	Holidays	30	
18	Sick Leave	32	
19	Maternity/Adoption/Parental Leave	34	
20	Position Protection	35	
21	Compassionate Leave	36	
22	Special Leave	37	
23	Seniority, Layoff and Recall	38	
24	Promotions and Staff Changes	41	
25	Salaries	44	
26	Pension Plan	44	
27	Staff Development	45	
28	Health Program	46	
29	Termination of Employment	47	
30	Discipline	47	
31	Dressing Rooms and Parking Facilities	48	
32	Part-Time Employees	49	
33	Casual Employees	49	
34	Nurse Practitioners	53	
35	Special Allowance	53	
36	Temporary Assignment	53	
37	Use of Automobile	55	
38	Workers' Compensation	56	
39	Transfers and Portability	57	
40	Technological Change or New Method of Operations	58	
41	Complete Agreement	59	
42	Leave for Attendance – CRNNL Annual Meeting	59	
43	Schools of Nursing	59	

Article No.	<u>Title</u>	Page No
44	Educational Differential	61
45	Contracting Out	61
46	Family Leave	62
47	Workplace Violence and Sexual and Personal Harassment	63
48	Family Violence Leave	65
49	Professional Practice Committee	65
50	Retroactivity	67
51	Terms of Agreement	67
Schedules		
Schedule	A - Step Progression and Contact Allowance – Salaries	69
Schedule	B - Experience Credits	87
Schedule	e C - Employers Covered by this Agreement	88
Schedule	D - Classifications of Nurses Covered by this Agreement	89
	E - Vacation Leave	90
Schedule	F - Classification Review and Appeal Process	91
	e G - Summary of Group Insurance Benefits	96
	e H - Labrador Benefits Agreement	105
	e I - Deferred Salary Leave Plan	118
	e J - Job Sharing Guidelines	125
	e K - Seniority Numbers	127
	e L - Transition Agreements	130
	e M - Alternate Scheduling	175
	e N – Layoff and Recall	180
	e O – MOA of November 2015	185
Letters of Unde	erstanding	
	Leave and Vacation	189
Loss of S		190
	Holiday and Weekends Off	191
Hiring Pi	•	192
•	ation of Nursing Work Studies	193
BN 2000		195
	Employees Working Additional Shifts	196
	and District Continuing Care Program	150
	nd St. John's Home Care Program	198
	of Nursing	199
	Workplace Pilot	200
	Paid Sick Leave (Article 18.01)	201
	d Senior Level Collaboration between RNUNL,	201
	Government and NL Health Services	202
	Adjustment	203
	Hours of Work Community Health	203
Nurse U		204
	of Relief Staffing	203
Sick Lea	<u> </u>	207
	al Workload Measurement Committee	207
1 TOVINCIA	at workidau ivicasurcincin Cullillitte	200

Other Post-Employment Benefits ("OPEB") Eligibility	211
Annual Vacation and Holiday Provisions for Employees Employed by GNL	213
Provincial Nursing Network	214
Job Postings	215
Severance Pay	216
Statutory Holiday Replacement	217
Other Post-Employment Benefits ("OPEB")	219
Temporary Overtime Rate of Pay	220
Portability of Seniority	221
Conversion of Casual and Part-time Employees	222
Permanent Full-time (PFT) Incentive Pilot Project	223
Health Sector Safety Council	224
Preceptorship/Mentorship	225
Ten Hour Shifts	227
Flexible 12 Hour Shifts	228
Expiration of Paid Sick Leave	229
Casual Employees	230
Nurses' Automobile Expenses	231
Pilot Project	
Granting of Leave of Less Than One week after Schedule has been Posted	232
Provincial Temporary Employees	233
Memorandum of Understanding	
Agreement on Pensions	234
Classification Plan	235
Layoffs During the Term of the Collective Agreement	236
Arbitration	237
Dispute Resolution	238
Professional Practice Committee	239
Bachelor of Nursing in Community Health	240
Memorandum of Agreement re: Kilometer Rate Adjustment Formula (RNUNL)	241
Memorandum of Agreement re: Nursing Practice and Leadership Premium	245
Index	258

Clauses changed from the previous Collective Agreement are noted with an asterisk (*)

Article 1 - Purpose of Agreement

1.01 The purpose of this Agreement is to maintain harmonious and mutually beneficial relationships among the Employer, the employees and the Union and to set forth certain terms and conditions of employment relating to remuneration, hours of work, employee benefits and general working conditions affecting employees covered by this Agreement.

Article 2 - Interpretations and Definitions

- *2.01 For the purpose of this Agreement:
 - (a) "Casual Employee" means any employee who works on an occasional or intermittent basis.
 - (b) "Chief Executive Officer" means the highest management officer of the hospital or their designated representative.
 - (c) "Chief Operating Officer/Vice President or designate" means the Chief Operating Officer/Vice President of Nursing Service and/or Education, or person acting in either capacity.
 - (d) "**Day**" means a calendar day, e.g. Sunday, Monday, etc. unless otherwise specified in the Agreement.
 - (e) "Day of Rest" means a calendar day on which an employee is not ordinarily required to perform the duties of their position other than:
 - (i) a designated holiday;
 - (ii) a calendar day on which the employee is on leave of absence.
 - (f) **"Demotion"** means a change from one classification to another classification for which a lower salary is paid.
 - (g) "**Employee**" means any person included in the bargaining unit who is employed by the Employer for remuneration including all full-time, part-time, temporary, and casual employees.
 - (h) "Employer" means the Treasury Board or a hospital listed in Schedule "C".
 - (i) "Experience" means time worked as a registered or graduate nurse where such work is acceptable for the purposes of obtaining and maintaining an active licence to practice nursing. In order for an employee to receive credit for any past experience, they must make every reasonable effort to provide written documentation from their previous Employer(s) to their Employer outlining the length and type of experience worked. Credit for past experience shall be applied from fourteen days prior to the date the Employer receives written documentation of same.
 - (j) **"Fiscal Year"** means the period extending from the first (1st) day of April in one year to the thirty first (31st) day of March in the succeeding year.

- (k) "**Grievance**" is a difference rising out of the interpretation, application, administration or the alleged violation of the provisions of this Agreement.
- (l) "**Holiday**" means the twenty-four (24) hour period commencing at 0001 hours of a calendar day designated as a holiday in this Agreement.
- (m) "Hospital" means a hospital, home or agency listed in Schedule "C".
- (n) "Month" is a calendar month e.g. January, February, etc.
- (o) "Month of Service" means a calendar month in which an employee is in receipt of full salary or wages in respect of the prescribed number of working hours in each working day in the month and includes a calendar month in which an employee is absent on special leave without pay not in excess of one hundred and fifty (150) working hours.
- *(p) "Nurse" means a registered or unregistered nurse or nurse practitioner, who is employed by the Employer in one of the classifications listed in Schedule "D".
- *(q) "Part-time Employee" means any employee who has completed their probationary period of six hundred (600) working hours (exclusive of any orientation period up to a maximum of 150 hours) and who shall be regularly scheduled to work the number of shifts each week as specified in their letter of appointment unless otherwise mutually agreed between the Employer and the Employee. The letter of appointment shall be provided within thirty (30) days of their appointment to the part-time position. Notwithstanding the above, the hours of work shall be less than those applicable to a full-time employee in their classification.
- *(r) "Probationary Period" means a period of continuous employment of six hundred (600) hours of work (exclusive of any orientation period up to a maximum of one hundred and fifty (150) hours) from the date of last hiring. For the purpose of this Clause, any time off from work excluding statutory holidays, annual leave, or leave in lieu of overtime pay shall not be considered as time worked.
- (s) **"Promotion"** means a change from one classification to another classification for which a higher salary is paid.
- (t) "Registered Nurse" means a registered nurse as defined in the *Newfoundland and Labrador Registered Nurses' Act.* 1990.
- (u) "Service" means any period of employment either before or after the date of signing of this Agreement in respect of which an employee is in receipt of salary or wages from the Employer and includes periods of special leave without pay not exceeding one hundred fifty (150) working hours in the aggregate in any year unless otherwise specified in this Agreement.

(v) "Shift"

(i) Eight (8) Hour Shifts

"Shift" means the normal consecutive working hours scheduled for each employee which occur in any twenty-four (24) hour period. In each twenty-four (24) hour period there will normally be three (3) shifts, viz, day, evening, and night. The first shift of each day shall commence at 0001 hours.

(ii) Twelve Hour Shifts

"Shift" means the normal consecutive working hours scheduled for each employee which occurs in any twenty-four (24) hour period. There will normally be two (2) shifts in a twenty-four (24) hour period viz, day and night. The day shift shall start between the hours of 0730 and 0830. The night shift shall start between the hours of 1930 and 2030.

- (w) "**Standby**" means any period of time during which an employee is designated to be available to return to duty when called upon by their Employer.
- (x) "**Temporary Assignment**" means the explicit assignment of an employee by the Employer to fill temporarily a position on a higher classification and for which a higher rate of pay is applicable.
- (y) "**Temporary Employee**" means an employee who is employed for a specified period of time or for the purpose of performing specific work. The employee's rights when the temporary employment ends are specified in Articles 23.11. Temporary employees shall be hired for a period of more than three (3) months and at least fifty percent (50%) of full time hours with the exception of a temporary employee who is replacing a permanent employee working less than half time hours.

Effective date of signing, when filling temporary vacancies which exceed or are expected to exceed twenty (20) weeks, they shall be filled in accordance with Article 24 of the collective agreement. (This clause overrides any language to the contrary in the transition agreements attached as Schedule L).

- (z) "Union" means the Registered Nurses' Union Newfoundland and Labrador. Whenever it is required in the Agreement that the consent of the Union be obtained, or that notice be given to the Union, these references shall be taken to mean the head office of the Registered Nurses' Union Newfoundland and Labrador, 229 Majors Path, St. John's, Newfoundland and Labrador.
- (aa) "Unregistered Nurse" means a nurse who has graduated from an approved school of nursing, but whose registration with the College of Registered Nurses of Newfoundland and Labrador is pending.
- (bb) "Week" means the period from 0001 hours Monday to 0001 hours the following Monday inclusive.

Article 3 – Recognition

3.01 **Recognition of Union**

- (a) The Employer recognizes the Union as the sole bargaining agent for those employees employed by the Employer in the classifications specified in Schedule "D".
- (b) In the event that there is a merger, amalgamation, consolidation or change of Employer, the representation rights of the Union shall be protected based on the *Public Service Collective Bargaining Act*.

3.02 New Classifications

In the event of the creation of a new classification during the term of this Agreement, the Employer agrees to consult with the Union as to whether such classification should be included in the bargaining unit. Should the parties be unable to agree, the matter shall be referred to the Labour Relations Board for adjudication.

3.03 **No Private Agreement**

There shall be no written or verbal agreement between an employee and the Employer which may conflict with the terms of the Collective Agreement except where mutually agreed between the employee, the Union and the Employer.

3.04 Agreement Overrules Hospital Policy

The provisions of this Collective Agreement shall take precedence over any and all policies, rules and regulations made by the Employer concerning wages, benefits or working conditions affecting employees within the scope of this Collective Agreement.

3.05 Future Legislation

In the event that any law passed by the Legislature applying to the employees covered by this Agreement renders null and void any provisions of this Agreement, the remaining provisions of the Agreement shall remain in effect for the term of this Agreement.

3.06 Union Access

Employees shall have the right at any time to have the assistance of a paid representative(s) of the Union on all matters relating to Employer-employee relationships. Union representative(s) shall have access to the Employer's premises in order to provide the required assistance. Employees involved in such discussions or investigation of grievances shall not absent themselves from work except with permission from their immediate supervisor, and such permission will not be unreasonably withheld.

3.07 **Permission to Hold Meetings**

Permission to hold Union meetings on the Employer's premises shall be obtained from the Chief Executive Officer and such meetings shall not interfere with the operation of the Employer.

Article 4 - Management Rights

4.01 The Union recognizes and agrees that all the rights, powers and authority both to operate and manage the hospital under its control and to direct the working force is vested exclusively with the Employer except as specifically abridged or modified by the express provisions of this Agreement. Should a question arise as to the exercise of management's rights in conflict with the specific provisions of this Agreement, failing agreement by the parties, the matter shall be determined by the Grievance and Arbitration Procedure.

Article 5 - Employee Rights

- 5.01 Notwithstanding anything contained in this Agreement, any employee may present a personal complaint to the Chief Executive Officer or designate.
- 5.02 Notwithstanding anything in this agreement, any employee may present a workplace issue that is not properly the subject of a grievance in accordance with Article 49, Professional Practice Committee.

*5.03 Employer Shall Not Discriminate

The Employer agrees that in accordance with the provisions of the *Newfoundland and Labrador Human Rights Act*, there shall be no discrimination with respect to any matter of hiring, wage rates, training, upgrading, promotion, transfer, layoff, recall, discipline, classification, discharge, assignment of work or otherwise because of race, color, nationality, ethnic origin, social origin, religious creed, religion, age, disability, disfigurement, sex, sexual orientation, gender identity, gender expression, marital status, family status, source of income, and political opinion or activity in the Union.

5.04 **Posting of Rules and Regulations**

All rules, policies, regulations, directives and memos of the Employer which directly affect employees in the bargaining unit required for proper management of health services being provided shall be readily available in the workplace.

5.05 Work of the Bargaining Unit

An employee who is a nurse, whose job is not included in the bargaining unit may work on any job which is included in the bargaining unit for the purposes of instruction and emergencies or when regular employees are temporarily unavailable. Employees whose jobs are in the bargaining unit shall not work on any jobs which are not included in the bargaining unit except in cases of temporary assignment or as mutually agreed upon by the Employer and the employee.

5.06 **Signing Reports**

An employee shall be given an opportunity to sign all evaluations and assessments of their work performance. The employee's signature on these evaluations and assessments shall not be construed to mean agreement with the evaluations or assessments but merely receipt of same. If the employee refuses to sign any evaluations or assessments, it shall be indicated on the evaluation or assessment that it was presented to them and they refused to sign it.

5.07 **Personal Files**

There shall be one (1) recognized personal file and this file shall be maintained by the Personnel Department. A copy of any document which might, at any time, be the basis of disciplinary action or which may affect the employee's advancement or standing with the Employer or which may affect the Employer's recommendation concerning the employee to another place of employment or educational institution shall be placed in their personal file and a copy shall be supplied concurrently to the employee. If this procedure is not followed, such a document shall not become a part of their record for use against them at any time. The employee shall acknowledge having received such document by signing the file copy.

5.08 Consultation of File

An employee has the right, after making an appointment and during regular working hours, to consult their personal file. Any employee who has been terminated may consult their personal file within twenty one (21) days of the date of their termination. Any employee involved in grievance or arbitration proceedings shall have the right to copy all documents contained in their personal file.

Article 6 - Union Representation

6.01 (a) Names of Representatives

The Union shall provide the Employer with a list of the Officers of the Union. The Union shall also provide the Employer with a list of shop stewards and area(s) represented before the Employer shall be required to recognize the steward(s).

(b) **Shop Stewards**

- (i) The Union shall be permitted to appoint one (1) shop steward in each place of employment for each eighteen (18) nurses employed or part thereof.
- (ii) The Union shall be permitted to appoint one (1) shop steward for each eighteen (18) Community Health Nurses (including Public Health Nurses who work in a community setting) employed or part thereof.

6.02 Leave for Union Business

- (i) (a) For an employee elected to the Provincial Executive of the Union and who is required to attend Executive meetings of the Provincial Executive, leave with pay not exceeding twenty-two point five (22.5) working hours in any one (1) year.
 - (b) For an employee elected to positions on the Board of Directors of the Union who is required to attend Board meetings, leave with pay not exceeding twenty-two point five (22.5) working hours in any one (1) year.
 - (c) For an employee who is a member of the Provincial Executive, Board of Directors or an elected delegate and who is required to attend the annual Convention of the Union, leave with pay not exceeding twenty-two point five (22.5) working hours in any one (1) year. The number of delegates selected from each hospital covered by this agreement shall not in any event exceed the number of shop stewards appointed or elected under Article 6.01 (b).

- (d) Leave with pay shall be granted to an employee to attend educational seminars, provided that the total leave with pay granted under this Clause in any one (1) year shall not in any event exceed that number of working hours which is obtained by multiplying the number of shop stewards in the bargaining unit by seven point five (7.5) hours.
- (e) An employee who is a member of the Canadian Federation of Nurses' Unions or an elected delegate shall be allowed leave with pay not exceeding thirty-seven point five (37.5) hours in any one (1) year for the purpose of attending the Canadian Federation of Nurses' Unions Convention. No more than two (2) employees from each hospital at any one (1) time.
- (f) The minimum period of leave for Union business that can be taken is one (1) hour.
- (ii) Members of the Negotiating Team shall suffer no loss in pay while attending collective agreement negotiations with the Employer. Permission to attend such meetings shall not be unreasonably withheld.
- (iii) Additional leave without pay for the purpose of attending to Union business may be granted to the Provincial Vice President and the Secretary-Treasurer by the Chief Executive Officer on request. Such leave shall not be unreasonably denied.
- (iv) Once the schedule has been posted in accordance with Article 8.03, an employee's schedule shall not be changed to avoid the payment of benefits in accordance with Article 6.02 (i) (d).

(v) Notice and Approval of Union Leave

Approval of Union leave is subject to the requirement that reasonable notice of the request be provided from the Union.

6.03 Leave of Absence for Full-Time Union Representatives

- (a) An employee who is elected or selected for a full-time executive position with the Union shall be granted leave of absence without loss of seniority or accrued benefits for a period of three (3) years. Such leave shall be renewed every three (3) years, on request, during their term of office.
- (b) An employee who is selected for a full-time position with the Union shall be granted a leave of absence without loss of seniority or accrued benefits for a period of three (3) years. Such leave shall be renewed each three (3) years upon request.

Article 7 – Union Dues

7.01 **Union Dues**

The Employer shall deduct from every employee in the bargaining unit and remit monthly to the Union an amount equal to the dues of the Union. The Union shall notify the Employer of the amount of Union dues to be deducted.

7.02 **Deductions**

Deductions shall be forwarded to the Union not later than the 15th day of the month following the month in which the deductions are made. The Employer will forward to the Union, with the first dues deductions cheque following the signing of the agreement, a list which shows the employee's full name, classification number, and the employee's contribution. Each month thereafter, a similar list highlighting additions and deletions will be forwarded with the dues deduction cheque.

7.03 **T4 Slips**

The Employer shall indicate on the employee's T4 slip the total amount of Union dues paid during the previous taxation year.

Article 8 - Hours of Work

*8.01 Hours of Work

(a) **Eight (8) Hour Shifts**

The normal hours of work for employees working an eight (8) hour shift schedule shall be thirty-seven and one-half (37 1/2) hours per week or seventy-five (75) hours a fortnight divided into seven and one half (7 1/2) hour shifts excluding a one half (1/2) hour unpaid meal period but including a rest period of fifteen (15) consecutive minutes in the first half and second half of the shift. The meal period and rest period(s) may be combined by mutual agreement between the employee and their supervisor.

(b) Twelve (12) Hour Shifts

The hours of work for employees working a twelve (12) hour shift schedule shall be seventy-five (75) hours a fortnight divided into eleven and one quarter (11 1/4) hours or a combination of seven and one half (7 1/2) and eleven and one quarter (11 1/4) hours. The work schedule may be changed by mutual consent between the employees and the Employer. The seven and one half (7 1/2) hour shifts excludes a thirty (30) minute unpaid meal period but includes a rest period of fifteen (15) consecutive minutes in the first half and second half of the shift. The eleven and one quarter (11 1/4) hour shift excludes a forty five (45) minute unpaid meal period but includes a rest period of fifteen (15) consecutive minutes during each third of the shift. The meal periods and rest period(s) may be combined by mutual agreement between the employee and their supervisor.

*(c) Flexible Hours of Work Community Health

(i) Work Schedule

The work schedule (minimum of 4 weeks schedule) showing the shifts and days of rest shall be communicated to employees at least two weeks in advance.

(ii) **Hours of Work**

Recognizing the unique needs of clients requiring community based nursing and the unpredictable nature of the work, the parties agree that the work day for employees shall operate on a flexible basis in the interest of client care and/or efficiency of the overall operations.

- (a) It is intended that the base work schedule to which flexibility will be applied will be seven and one-half (7 ½) hour work days (or a seven (7) hour workday applicable to those nurses grandparented) excluding a thirty (30) minute unpaid meal period (or a one (1) hour unpaid meal period applicable to those nurses grandparented) but including a rest period of fifteen (15) consecutive minutes in the first half and second half of the day.
- (b) The hours of work for nurses shall be one hundred and fifty (150) hours (or one hundred and forty (140) applicable to those nurses grandparented) within a four (4) week period, or a shorter designated period as agreed to by the Employer and union. As of the date of signing of this agreement the parties will agree that an initial four week period will be designated for each employee which will correspond with a specific payroll period. Once start dates are agreed upon the designated four week periods will continue thereafter.
- (c) The seven and one-half (7 ½) hour workday (or the seven (7) hour workday applicable to those nurses grandparented) may be flexed to ten (10) hours per day in the interest of client care and/or efficiency and/or to complete work. The shift shall not begin before 0700 hours or extend beyond 2200 hours.
- (d) The Employer shall make every reasonable effort to notify an employee of any anticipated changes to the length of their workday.
- (e) It is agreed that no premium or overtime shall apply where an employee chooses to work:
 - 1) on their scheduled day of rest; or
 - 2) in excess of ten (10) hours per day; or
 - 3) outside the hours of 0700 to 2200 hours; or
 - 4) in excess of one hundred and fifty (150) hours in a four (4) week designated period(or one hundred and forty (140) hours applicable to those nurses grandparented).

*(iii) Flextime Hours

It is the expectation and responsibility of nurses to endeavour to balance flextime worked with flextime taken. If, at the end of a designated flextime period, a nurse has worked in excess of one hundred and fifty (150) hours (or of one hundred and forty (140) hours applicable to those nurses grandparented) and those excess hours are not approved by the manager as overtime, such hours must be scheduled off in the next flextime period at the rate of straight time. At the end of the second flex

time period, if the nurse was unable to flex their time off or it was unable to be scheduled off by the manager they shall receive payment at time and one half (1 ½). Where the nurse requests and the manager approves to maintain flex hours beyond the next flex time period, these shall be banked and paid at straight time.

(iv) **Differentials**

- (a) Shift differential shall be paid for all hours of work scheduled as the base schedule by the Employer between 1700 hours and 0700 hours.
- (b) Shift differential shall be paid for all flextime hours worked between 1800 hours and 0700hours.

(v) **Consecutive Shift**

No employee shall be compelled to work more than seven (7) consecutive days.

(vi) **Days of Rest**

Days of rest shall be allocated at the rate of the minimum of two (2) consecutive days of rest per week except where mutually agreed between the employee and the Employer.

(vii) Meal and Rest Periods

The meal period and the rest period(s) may be combined by the employee depending on operational requirements. Meal periods and rest periods cannot be scheduled to allow for delayed start of shift or early departure from work.

(viii) Weekends Off

All employees, including part-time employees, shall receive a minimum of two (2) weekends off out of every four (4) weekends unless otherwise agreed by mutual consent between the employee and the Employer. The Employer shall endeavor to grant every second weekend off unless otherwise agreed by mutual consent between the employee and the Employer. The weekend off shall be a Saturday and the Sunday immediately following, ensuring the employee receives a minimum of fifty six (56) consecutive hours off duty.

(ix) **Split Shift**

An employee shall not be required to work a split shift without their consent.

(x) **Overtime**

- (a) When a full-time or part-time nurse is required by the Employer to work:
 - 1) on their scheduled day of rest; or
 - 2) in excess of ten (10) hours per day; or

- 3) outside the hours of 0700 to 2200 hours; or
- 4) in excess of one hundred and fifty (150) hours in a four (4) week designated period (or one hundred and forty (140) hours applicable to those nurses grandparented),

they shall be granted at their option compensatory time off or overtime at the applicable rate.

- (b) When a casual employee is required by the Employer to work:
 - 1) in excess of ten (10) hours per day; or
 - 2) in excess of one hundred and fifty (150) hours in a four (4) week designated period,

they shall be paid at the applicable overtime rate.

(c) Employees shall be entitled to maintain a maximum of forty (40) hours of compensatory time. All compensatory time in excess of forty (40) hours will be paid.

8.02 **Meal Period**

- (a) The Employer will endeavour to post the meal period of employees prior to the commencement of the shift, but in any event it shall be posted within the first two (2) hours of the shift.
- (b) During the meal period employees are permitted to leave the premises.
- (c) Employees required to work during meal breaks shall be paid at time and one half (1 1/2) rate for the meal period.

8.03 Working Schedule

The working schedule (minimum of four weeks schedule) showing the shifts and days of rest shall be posted in an appropriate place at least two (2) weeks in advance.

8.04 **Rotation of Shifts**

(a) **Eight (8) Hour Shifts**

The rotation of shifts shall be carried out in an equitable manner. Each employee shall receive at least eight (8) days of day shifts in a month provided they may waive this right.

(b) Twelve (12) Hour Shifts

The rotation of shifts shall be carried out in an equitable manner. Each employee shall receive at least seven (7) days of day shifts per month provided they may waive this right. Where the current twelve (12) hour shift scheduling provides a greater benefit, this benefit will not be reduced during the term of this agreement.

8.05 Consecutive Shifts

(a) **Eight (8) Hour Shifts**

No employee shall be compelled to work more than seven (7) consecutive day shifts or more than six (6) consecutive evening or night shifts unless otherwise agreed by mutual consent between the employee and their supervisor.

(b) Twelve (12) Hour Shifts

No employee shall be compelled to work more than three (3) consecutive shifts. In any case, employees, by mutual consent with their supervisor may agree to work a greater number of consecutive shifts.

8.06 **Days of Rest**

- (a) Days of rest shall be allocated at the rate of a minimum of two (2) consecutive days of rest per week, except where mutually agreed between the employee and their supervisor.
- (b) For the purposes of Clause 9.06, in scheduling part-time employees, the Employer shall allocate two (2) consecutive days of rest per week unless otherwise agreed between the Employer and the employee.

8.07 **Preference for Days of Rest**

Each nurse may state their preference with regard to days of rest before the work schedule is drawn up and consideration shall be given to these preferences wherever they do not conflict with the need to maintain service and adequate levels of staffing.

8.08 Exchanging Days of Rest

Employees may exchange their days of rest with the approval of the manager or designated representative.

8.09 Weekends Off

(a) **Eight (8) Hour Shifts**

Employees, including part-time employees, shall receive a minimum of every third weekend off and the Employer shall endeavour to grant every second weekend off, unless otherwise agreed by mutual consent between the employee and their supervisor. The weekend off shall mean a Saturday and the Sunday immediately following, ensuring the employee a minimum of fifty-six (56) hours off duty.

(b) Twelve (12) Hour Shifts

Employees, including part-time employees, shall receive a minimum of two (2) weekends off out of every four (4) weekends and the Employer shall endeavour to grant every second (2nd) weekend off, unless otherwise agreed by mutual consent between the employee and their supervisor. Where the current twelve (12) hour shift schedule provides for every second (2nd) weekend off, this benefit shall be maintained during the term of the current agreement. A weekend off shall be a

period including Saturday and Sunday from at least Friday at 0800 hours to Monday at 0800 hours or from at least Friday at 2000 hours to Monday at 2000 hours.

8.10 Rest Between Scheduled Shifts

(a) **Eight (8) Hour Shifts**

There shall be at least sixteen (16) hours between scheduled shifts (excluding overtime) unless otherwise agreed between the employee and their supervisor. Where sixteen (16) hours of rest (excluding overtime) are not provided, the employee shall receive pay at the rate of time and one half (1 1/2) for each hour worked on the scheduled shift which infringes on the sixteen (16) hour rest period.

(b) Twelve (12) Hour Shifts

There shall be at least twelve (12) hours between scheduled shifts (excluding overtime) unless otherwise agreed between the employee and their supervisor. Where twelve (12) hours of rest (excluding overtime) are not provided, the employee shall receive pay at the rate of time and one half (1 1/2) for each hour worked on the scheduled shift which infringes on the twelve (12) hour rest period.

8.11 **Double Shift**

An employee shall not be required to work a double shift without their consent.

8.12 **Differentials**

(a) (i) Shift Differential (Eight (8) Hour Shifts)

Effective date of signing, a shift differential of two dollars and thirty cents (\$2.30) per hour shall be paid for each hour the employee works between the hours of 1600 hours on one day and 0800 hours on the following day, excluding those employees working on a recognized day shift.

(ii) Shift Differential (Twelve (12) Hour Shifts)

Effective date of signing, a shift differential of two dollars and thirty cents (\$2.30) per hour shall be paid for each hour the employee works between the hours of 1600 hours on one day and 0800 hours on the following day.

(b) Saturday and Sunday Differential

Effective date of signing, a Saturday and Sunday differential of two dollars and fifty-five cents (\$2.55) per hour shall be paid for each hour worked by an employee between the hours of 0001 hours Saturday and 2400 hours Sunday.

(c) If an employee qualifies for both differentials under (a) and (b) above, they shall receive both.

8.13 **No Split Shift**

There shall be no split shifts.

8.14 Hours of Work Not Applicable to Public Health Nurses

The provisions of Clause 8.01 will not apply to Public Health Nurses and in such cases the hours of work currently in effect will not be changed during the term of this agreement.

8.15 Two Shift Rotation (Eight (8) Hour Shifts)

Through mutual consent the Union and the Employer may agree to institute a two shift rotation in places of employment. The two shift rotation shall be either evenings and nights, days and evenings or days and nights.

8.16 Twelve (12) Hour Work Schedule

(a) The Employer and the Union may agree that employees in a particular work area may work a twelve (12) hour shift schedule. The request for a twelve (12) hour shift schedule may come from seventy-five (75) percent of the employees in the work area or the Employer, and if agreed upon by the parties, all employees in that work area shall work a twelve (12) hour shift schedule. This twelve (12) hour shift schedule shall remain in effect in the work area until either of the parties gives the other thirty (30) calendar days' notice of its intention to terminate this agreement. The request to terminate the twelve (12) hour shift schedule by the Union must have the support of seventy-five (75) percent of the employees in the work area.

Only employees holding permanent positions in the work area, including those permanent employees on leaves of absences, will be eligible to vote. Casual employees, employees holding temporary positions in the work area, and employees occupying permanent float positions, shall not be eligible to vote.

- (b) The first twelve (12) weeks of the twelve (12) hour shift schedule shall be considered a trial period. Prior to the completion of the trial period, a meeting of the employees in the particular work area will take place to review the twelve (12) hour shift schedule. Assuming the Employer and seventy-five (75) percent of the employees in the work area wish to continue the twelve (12) hour shift schedule, it shall be continued.
- (c) All Clauses designated <u>Twelve (12) Hour Shifts</u> shall apply to employees working the twelve (12) hour shift schedule. All other Clauses of the agreement, which have not been amended for <u>"Twelve (12) Hour Shifts"</u> shall apply to employees who work on a Twelve (12) Hour Shift schedule as they do to all other employees.

8.17 **Permanent Evenings and Nights**

The Employer shall endeavour to accommodate the request of Employees to work permanent evenings and/or nights provided the other employees in the work area have no objection.

When an employee works permanent evenings and/or nights, the employee shall be required to work up to one hundred and fifty (150) hours of days in each calendar year, scheduled by mutual agreement between the employee and their supervisor.

With the appropriate notice (not less than 30 days) by the employee or the Employer, either party may terminate the permanent evening and/or nights arrangement.

Employees working permanent evenings and/or nights as of April 1, 1999, shall continue to work in that arrangement and the one hundred and fifty (150) hour requirement shall not apply to those employees.

Article 9 – Overtime

9.01 **Overtime Rate**

When an employee is required to work in excess of their normal hours, they shall be granted, at their option, compensatory time off at the rate of one and one half (1 1/2) times the number of hours worked in excess of their normal hours, or overtime pay at the rate of one and one half (1 1/2) times their regular rate of pay for the time worked in excess of the normal hours of work. The employee's decision to receive compensatory time off for the overtime must be conveyed to the manager within seventy-two (72) hours of the conclusion of working the overtime. The maximum amount of overtime that can be accumulated as compensatory time off is one hundred and fifty (150) hours unless otherwise agreed to between the union local and Employer. For overtime in excess of one hundred and fifty (150) hours the employee shall receive pay at the applicable overtime rate.

9.02 **Regular Hourly Rate**

The regular hourly rate for employees other than Public Health Nurses employed by the Province of Newfoundland and Labrador shall be calculated by dividing their annual salary by 1950. The regular hourly rate for a Public Health Nurse shall be calculated by dividing their annual salary by 1820.

9.03 **Calculation of Overtime**

Subject to Clause 9.04, overtime shall be calculated in thirty (30) minute units.

9.04 **Approval of Overtime**

All overtime must be authorized by the Employer except in the case of emergency.

9.05 **Sharing of Overtime**

When overtime is required by the Employer, it shall be offered to those employees who have indicated their willingness to work the required overtime. The Employer will not compel the employee to work overtime if another qualified employee is willing to work that overtime, provided that there is no additional cost to the Employer and provided the employee is available. Subject to the foregoing overtime shall be divided as equally as possible amongst employees qualified to perform the available work.

9.06 Change of Days of Rest

When an employee's days of rest are changed without having been given at least forty-eight (48) hours prior notice of having to work on their day(s) of rest, they shall be paid double their regular hourly rate for each hour worked on the scheduled day(s) of rest. This Clause shall not apply if the day(s) of rest was changed at the request of the employee.

9.07 **Change of Shift**

When an employee's regularly scheduled shift is changed to another shift in that day, they shall be given prior notice as follows:

- (i) twenty-four (24) hours notice before the originally scheduled shift if the rescheduled shift occurs after the originally scheduled shift.
- (ii) twenty-four (24) hours notice before the rescheduled shift if the rescheduled shift occurs before the originally scheduled shift.

Should the required notice not be given in accordance with this article, the employee shall be paid at the rate of time and one half (1 1/2) their regular hourly rate for the shift worked.

This Clause shall not apply if the change of shift was made at the request of the employee.

9.08 Notification of Changes in Schedule

In cases where the work schedule is changed while an employee is on their day(s) of rest and such employee is affected by the change, the Employer is responsible for notifying those employees affected by the change before they report for work.

9.09 State of Emergency Due to Weather Conditions - Adverse Weather Conditions

- (a) The following provisions shall apply to employees during adverse weather conditions necessitating a state of emergency declared by either the Employer or the appropriate provincial or municipal authority.
 - (i) All employees are required to report for duty as scheduled.
 - (ii) When an employee through no fault of their own is unable to report for work because of a declared state of emergency, such employee shall suffer no loss of pay or other benefits, nor shall they be required to make up, in anyway, for time lost due to not reporting for work.
 - (iii) Notwithstanding 9.09 (a) (i) above, the Employer reserves the right to close down or reduce staffing levels in any department(s) in which event employees so affected will not be required to report for duty and shall be paid in accordance with the terms of 9.09 (a) (ii) above.
 - (iv) An employee who worked during the emergency will be paid at the rate of time and one half (1 1/2) for all hours worked.
- (b) If during adverse weather conditions, the Employer provides transportation for nurses to the hospital, then it shall be the Employer's responsibility to provide transportation for those nurses when they return to their homes.

9.10 No Layoff to Compensate For Overtime

An employee shall not be laid off during regular hours to equalize any overtime worked.

9.11 **Consecutive Work Premium**

(a) **Eight (8) Hour Shifts**

Subject to Clause 8.05, all work performed on the seventh (7th) consecutive evening or night shifts (or any combination of evening or night shifts) shall be paid for at the rate of time and one half (1 1/2) and work performed on the eighth (8th) and subsequent consecutive shifts shall be paid for at a double time rate. This Clause

shall not apply to those consecutive shifts worked subject to the written and signed request of the employee.

(b) Twelve (12) Hour Shifts

Subject to Clause 8.05, employees who work under a twelve (12) hour shift schedule will be paid time and one half (1 1/2) for all work performed on the fourth (4th) consecutive shift and double (2) time for the fifth (5th) and subsequent consecutive shifts. This Clause shall not apply to those consecutive shifts worked subject to the written and signed request of the employee.

9.12 **Daylight Saving Time**

The changing of Daylight Saving Time to Standard Time, or vice versa, shall not result in employees being paid less than normal scheduled daily hours and no overtime shall accrue.

*9.13 **Mandatory Exemption**

Where an employee has agreed to do overtime for a shift, the Employer shall endeavour to not mandate them at the end of their shift or the following day.

Article 10 - Standby

10.01 **Standby Duty Rates**

(a) **Eight (8) Hour Shifts**

Effective date of signing, employees performing standby duty shall be paid at the rate of twenty dollars and forty cents (\$20.40) per eight (8) hour shift or part thereof.

(b) Twelve (12) Hour Shifts

Effective date of signing, employees performing standby duty shall be paid at the rate of thirty dollars and sixty cents (\$30.60) per twelve (12) hour shift or part thereof.

10.02 **Standby Duty Rates on Designated Holiday**

(a) **Eight (8) Hour Shifts**

Effective date of signing, when standby duty is required on a designated holiday, the rate of pay shall be twenty-two dollars and sixty cents (\$22.60) for each eight (8) hour shift or part thereof.

(b) Twelve (12) Hour Shifts

Effective date of signing, when standby duty is required on a designated holiday, the rate of pay shall be thirty-three dollars and ninety cents (\$33.90) per twelve (12) hour shift, or part thereof.

10.03 Failure to Report for Standby Duty

No payment shall be granted for the total period of standby duty if the employee does not report for work when required.

10.04 **Authorization of Standby Duty**

All standby duty shall be authorized and scheduled by the Employer at least six (6) days in advance except where otherwise mutually agreed between the employee and supervisor.

Article 11 – Callback

11.01 Callback Pay Rate

When an employee is called back to work and reports for work, they shall be paid for a minimum of three (3) hours at the applicable overtime rate.

11.02 Transportation Expenses

An employee shall not receive any payment for transportation expenses where:

- (a) they live in subsidized hospital accommodation adjacent to the hospital; or
- (b) transportation is provided by the Employer.

11.03 Transportation Expenses

Subject to Clause 11.02, when an employee is recalled to work under the conditions described in Clause 11.01, they shall be paid the cost of transportation to and from their place of work. Transportation allowance for employees shall be the cost of taxi fare from their residence to their place of work and return to a maximum of nine (9) dollars or a kilometre allowance from their place of residence to their work and return to a maximum of forty (40) kilometres at the rate established by the Government Memorandum of Agreement regarding Automobile Reimbursement.

11.04 Additional Duties

If it becomes necessary to assign duties on a callback which are additional to those for which the callback was made, the employee shall be paid at the applicable overtime rate with a minimum of one (1) hour at the applicable overtime rate for the time required to perform such additional duties.

11.05 Return to Work Following Callback

In cases where a nurse is required to work on a callback beyond 0200 hours and who has not had a sufficient rest period, they will be entitled to up to an eight (8) hour rest period without loss of pay.

Article 12 - Escort Duty

12.01 Compensation for Escort Duty

When an employee on duty is required to attend a patient as an escort during an ambulance trip (air, road, etc.) and the time involved on a trip exceeds the employee's regular shift, they shall receive time off or pay, at their option, at the applicable overtime rates for all time in excess of their normal shift which the employee spends with the patient or attending to the needs of the patient.

12.02 (i) Eight (8) Hour Shifts-Returning Within Seven and One-Half Hours

If the employee on escort duty commences their return to their hospital with equipment, drugs or documents and if none, then to their residence, within seven and one-half (7 1/2) hours following relief of escort duty, they shall receive time off or

pay, at their option, at the applicable overtime rates for all time spent in return travel in excess of their normal seven and one-half (7 1/2) hour shift.

(ii) Twelve (12) Hour Shifts-Returning Within Eleven and One Quarter (11 1/4) Hours

If the employee on escort duty commences their return to their hospital with equipment, drugs or documents and if none, then to their residence, within eleven and one-quarter (11 1/4) hours following relief of escort duty, they shall receive time off or pay, at their option, at the applicable overtime rates for all time spent in return travel in excess of their normal eleven and one-quarter (11 1/4) hour shift.

12.03 No Loss of Days of Rest

(i) **Eight (8) Hour Shifts**

Subject to Clause 12.02, if an employee is detained following relief of escort duty due to weather or other transportation difficulties, they shall not lose their day(s) of rest. The time they are so detained and the time spent travelling back to their hospital shall be deemed to be time worked with a maximum of seven and one-half (7 1/2) hours at straight time in a twenty four (24) hour period.

(ii) Twelve (12) Hour Shifts

Subject to Clause 12.02, if an employee is detained following relief of escort duty due to weather or other transportation difficulties, they shall not lose their day(s) off. The time they are so detained and the time spent travelling back to their hospital shall be deemed to be time worked with a maximum of eleven and one-quarter (11 1/4) hours at straight time in a twenty-four (24) hour period.

12.04 Return to Work Following Escort Duty

(i) **Eight (8) Hour Shifts**

No employee returning from escort duty will be required to commence another regular seven and one-half (7 1/2) hour shift within twelve (12) hours of their return unless otherwise agreed between the Employer and the employee concerned. In cases where a nurse is required to work beyond 0200 hours, and who has not had a sufficient rest period, they will be entitled to up to an eight (8) hour rest period without loss of pay.

(ii) Twelve (12) Hour Shifts

No employee returning from escort duty shall be required to commence another regular eleven and one-quarter (11 1/4) hour shift within twelve (12) hours of their return unless otherwise agreed between the Employer and the employee concerned. In cases where a nurse is required to work beyond 0200 hours, and who has not had a sufficient rest period, they shall be entitled to up to a twelve (12) hour rest period. The first eight (8) hours of the rest period shall be without loss of pay and the remainder shall be granted at their option, as compensatory overtime or annual leave.

12.05 Escort Duty - Travel Allowance

An employee shall be given a travel advance for all anticipated travel expenses by the Employer before commencing escort duty. A subsequent travel claim will be submitted in accordance with normal procedure and regulations.

12.06 Escort Duty Pay

- (a) Due to the nature of the assignment, an employee shall receive an additional fifty one dollars (\$51) for each assignment to escort duty. When involved with air ambulance or when more than four (4) hours are spent attending the patient the employee shall receive sixty-one dollars (\$61) for each assignment to escort duty. A nurse shall not be assigned to more than one (1) critically ill patient.
- (b) An employee required to escort a patient on their return travel to the hospital without prior notice shall receive twice the applicable rate in 12.06 (a).

12.07 **Escort Duty Overtime**

When an employee is called back to do escort duty Article 9 and Clause 11.03 shall apply.

12.08 Volunteer Escort Duty

Should an employee volunteer in advance for escort duty, and elect to take their days of rest before returning to duty, they shall receive only straight time for the travelling time on the return journey. The method of return travel is to be mutually agreed between the employee and the supervisor before the employee leaves.

12.09 No Allowance Within 25 Mile Radius

With the exception of Bell Island, the provisions of this Article shall not apply where an employee is escorting a patient within the same city or town, or another city or town within a radius of twenty-five (25) miles of the hospital.

12.10 Travel Insurance

Employees required to do escort duty shall be covered by adequate life insurance both while on and returning from the escort assignment regardless of the ownership or mode of transportation. Particulars of such insurance and/or copies of relevant policies shall be provided to the Union upon request.

12.11 Meal Allowance on Escort

Employees on escort duty shall be entitled to meal allowance in accordance with the provisions of Clause 37.03.

12.12 Private Accommodations

Employees on escort duty shall be entitled to the benefits as outlined under Clause 37.04.

Article 13 - Strikes – Lockouts

13.01 No Strikes or Lockouts

The Union agrees that there shall be no strikes, suspension or slowdown of work, picketing or any other interference with the Employer's business during the life of this agreement, and

the Employer agrees there shall be no lockout of employees during the life of this agreement.

Article 14 - Dispute Resolution

14.01 **Prompt Procedure**

It is of mutual interest to both the Employer and the Union that any grievances arising over the application, interpretation, administration or alleged violation of any of the terms of this Agreement be settled as expeditiously as possible in accordance with the procedure set forth hereunder.

14.02 (a) **Assistance from Shop Stewards**

The Employer acknowledges the rights and duties of the Union's Shop Stewards to assist any employee in preparing and presenting their grievance in accordance with the grievance procedure. At no time will an employee be requested by management at any level to discuss the grievance unless the employee has been informed of the purpose of the meeting and that the employee may have a shop steward or other union representative in attendance.

(b) **Processing of Grievance**

Shop Stewards shall suffer no loss in pay for the time spent processing grievances or attending meetings with the Employer's representatives.

(c) Permission to Leave Work

It is agreed that Shop Stewards will not absent themselves from their departments for the purpose of handling grievances without first obtaining permission of the Shop Steward's supervisor and that permission will not be unreasonably withheld.

14.03 **Types of Grievances**

Grievances arising out of the interpretation, application, administration or alleged violation of this agreement shall be subject to the Grievance and Arbitration Procedure set out hereunder. The following types of grievances concerning the application of Article 14 are recognized:

- (i) Employee Grievance: which shall be defined as the complaint of an individual employee.
- (ii) Group Grievance: which shall be defined as the complaint of a group of employees.
- (iii) Policy Grievance: which shall be defined as the complaint of the Employer or of the Union.

All grievance forms may be signed by a representative of the Union as agent for the employee, group or union, as the case may be.

14.04 Employer Representative at Grievance Steps

The Employer shall designate a representative at Step III of the grievance procedure and shall inform the Union office of the name and title of the person so designated at each step

together with the name or title of the immediate supervisor or local officer in charge to whom a grievance is to be presented.

14.05 Effective Date of Grievance Settlement

Settlement of grievance without reference to arbitration shall be applied retroactively to the date of occurrence of the action or situation which gave rise to the grievance, unless the settlement states otherwise.

14.06 **Technical Objection**

No grievance shall be defeated or denied by a technical objection caused by a clerical, typographical or similar technical error, or by the inadvertent omission of a step in the grievance procedure.

14.07 **Settlement of Dispute**

All disputes shall be processed without stoppage of work according to the following procedure:

NOTE: All references to meetings shall include the option of some or all participants participating by teleconference or video conference.

Step I: Informal Discussion

An employee with a dispute shall within seven (7) calendar days of the occurrence or the discovery of the facts giving rise to the dispute, discuss the dispute with the appropriate management representative. Within seven (7) calendar days, the parties shall work to resolve the issue without the necessity of filing a grievance.

Step II: Early Intervention Process

Failing satisfactory resolution of the dispute at Step I, and within a further seven (7) calendar days, a meeting shall occur to discuss the dispute with the goal of resolving it without the necessity of having to file a grievance. The management representative shall be responsible for setting up the meeting. A union representative, normally a branch representative, and human resources representative may be present at this meeting if requested by either of the parties. When appropriate, other management representatives may be invited to this meeting. Resolutions agreed upon at this meeting shall be without prejudice and without precedent.

Step III: Formal Grievance

Failing satisfactory resolution of the dispute at Step II, the employee may bring the dispute to the attention of a union representative, normally a branch representative, who will determine if a grievance should be filed. If it is determined that the grievance is to be filed at Step III, the employee or union representative, normally a branch representative, shall, within fourteen (14) calendar days of the early intervention meeting, submit the grievance to the Employer. A written decision from the management representative, outlining the nature of the decision, shall follow within seven (7) calendar days.

Depending on the nature of the dispute, either party may request a second meeting prior to providing a response. The human resources representative shall be responsible for arranging the second meeting. If this meeting occurs, the written response shall be

provided no later than (14) calendar days from when the grievance is filed. Any settlements reached at this step or any decision in writing shall be without prejudice and without precedent.

Step IV: Referral to Alternate Dispute Resolution (ADR) Process

Failing satisfactory settlement of the issue in Step III, and within fourteen (14) calendar days of receipt of management's written response in Step III, the Employer and Union (this decision may be delegated to in house counsel for the Employer and the Union) shall attempt to mutually agree on an Alternate Dispute Resolution (ADR) process such as external mediation, arbitration, mediation/arbitration and expedited arbitration. If no decision can be reached on the mechanism to be used, the grievance shall be automatically referred to arbitration within ninety (90) calendar days following the date of submission of the grievance.

At all steps of the Grievance Procedure an employee may be accompanied by a Union representative.

14.08 Grievance Presented by Mail

(a) Registered Mail

When a grievance is processed through the mail, all correspondence shall be registered. The time while the mail is moving from one destination to another shall not be considered in the Grievance Procedure time limits.

(b) Referral to Arbitration via E-Mail

The Employer or the Union may refer a grievance to arbitration via email at Step 4 of the Grievance Procedure or Step 3 of the Grievance Pilot Project. The grievance shall be deemed to have been submitted on the date that the email copy of the grievance was sent by the Employer or the Union. The recipient shall acknowledge by return email that the submitted grievance has been received. The parties shall provide each other with a list of the email addresses that are to be used for referring grievances to arbitration via email.

14.09 Amending of Time Limits

- (i) The above-mentioned time limits may be extended, in individual cases, by the consent in writing of both parties to the grievance.
- (ii) At any point in the grievance resolution process the parties may agree to have additional dispute resolution meetings.

14.10 May Omit Grievance Steps

An employee considered to be wrongfully or unjustly disciplined shall be entitled to bypass Steps I, II and III and grieve at Step IV directly. The grievance must be lodged within twenty-one (21) calendar days of the date of the discipline or the date notice of discipline was received, whichever is later.

14.11 **Bypass Steps**

Where a dispute arises involving a question of general application or interpretation of this agreement, the Union may initiate a grievance and the parties may mutually agree to bypass Steps 1 and 2 of this Article.

Article 15 – Arbitration

15.01 Composition of Board of Arbitration

When a grievance is submitted to arbitration the case shall be heard by a three-person board of arbitration which shall be established as follows:

- (i) Each party shall name an appointee to the Board within fourteen (14) calendar days of notification of Arbitration under Step 4 of the Grievance Procedure, and shall notify the other party of the name and address of the appointee.
- (ii) The appointees of the parties shall then select a chair upon whom they both agree which selection shall be made within seven (7) calendar days of the date of the appointment of the last named appointee by either party.
- (iii) If the party receiving the notice of arbitration fails to appoint an appointee, or if the two (2) appointees fail to agree upon a chair within seven (7) calendar days of their appointment, the appointment shall be made by the Minister of Labour upon the request of either party.
- (iv) The parties may mutually agree to the substitution of a single arbitrator for the Arbitration Board, in which event the provisions of Clauses 15.02, 15.03, 15.04, 15.05 and 15.06 (ii) shall apply equally to a single arbitrator where reference is made to an arbitration board.

15.02 **Board Procedure**

The Arbitration Board shall hear the evidence and shall render its decision in an expeditious manner and in any case, the decision shall be rendered in writing within one (1) month after the hearing subject to mutual agreement to the contrary.

15.03 **Decision of the Board**

Arbitration awards shall be final and binding and communicated to the parties in the form of a signed copy.

15.04 Jurisdiction of the Board

An arbitration board may not alter, modify or amend any provisions of this agreement but shall have the power to set aside or modify a decision of the Employer. No arbitration board shall make an award which would amend or change a collective agreement, a judgement or an earlier award.

15.05 **Decision on Monetary Award**

If following an arbitration award involving the payment of a sum of money not determined by the award, agreement cannot be reached between the parties regarding the amount to be paid, the matter shall be referred to the arbitration board who heard the grievance for settlement of the matter.

15.06 Expenses of the Board

Each party shall pay:

- (i) the fees and expenses of the appointees it appoints;
- (ii) one-half (1/2) of the fees and expenses of the chair.

15.07 Amending of Time Limits

The above mentioned time limits may be extended in individual cases by the consent in writing of both parties to the grievance.

15.08 Time Off for Witnesses

At any stage of the grievance or arbitration procedure, the parties shall have the assistance of any employee(s) concerned as witnesses and any other witnesses. Witnesses shall suffer no loss of pay as a result of assisting the parties in the grievance or arbitration process.

15.09 Expedited Arbitration

Subject to the agreement of the Employer and the Union, a form of expedited arbitration One (1) or Two (2) may be used following Step III of the grievance procedure. Both parties retain access to the complete arbitration process as described in Article 15 of the Agreement where either party does not agree to expedited arbitration.

Expedited 1

- (i) In any dispute of interpretation or application of the agreement, the parties agree to submit written briefs to a sole arbitrator. The briefs must be submitted within fifteen (15) calendar days of the appointment of the arbitrator.
- (ii) The sole arbitrator must be agreed to by both parties within seven (7) calendar days of the date of referral to arbitration. The arbitrator shall render a written decision within thirty (30) calendar days following receipt of the written briefs for each party. There shall be no right to rebut the other party's brief.
- (iii) Decisions pursuant to Expedited 1 shall not be precedential and will be without prejudice to any subsequent grievances of a similar nature.

Expedited 2

- (i) In any dispute of interpretation or application of the agreement, the parties agree to submit a written brief and present oral argument to the sole arbitrator. The written briefs must be submitted within fifteen (15) calendar days of the appointment of the arbitrator. The arbitrator shall schedule a hearing for the parties to present oral argument within ten (10) calendar days of the submission of the written briefs.
- (ii) The sole arbitrator must be agreed to by both parties within seven (7) calendar days of the date of the referral to arbitration. The arbitrator shall render a written decision within thirty (30) calendar days following the arbitration hearing.
- (iii) The sole arbitrator may, for the purpose of clarification, request the appearance of witness(s) for questioning at the time of the hearing or during the decision period when an additional meeting for this purpose may be convened by the arbitrator.
- (iv) Decisions pursuant to Expedited 2 shall not be precedential and will be without prejudice to any subsequent grievance of a similar nature.

Article 16 – Vacation

16.01 **Length of Vacation**

(a) **Eight (8) Hour Shifts**

An employee, other than Public Health Nurses employed by the Province of Newfoundland and Labrador, shall receive an annual vacation with pay in accordance with their years of employment as follows:

- (i) less than one (1) year one and two-thirds (1 2/3) working days for each month of service;
- (ii) one (1) year or more but less than ten (10) years four (4) weeks;
- (iii) more than ten (10) years but less than twenty five years of service five (5) weeks:
- (iv) more than twenty-five (25) years of service six (6) weeks.

(b) Twelve (12) Hour Shifts

An employee, other than Public Health Nurses employed by the Province of Newfoundland and Labrador, shall receive an annual vacation with pay in accordance with their hours of employment as follows:

- (i) Less than one thousand nine hundred and fifty (1,950) hours twelve point five (12.5) working hours for each one hundred and sixty-two point five (162.5) hours of service;
- (ii) One thousand nine hundred and fifty (1,950) hours or more but less than nineteen thousand five hundred (19,500) hours, one hundred and fifty (150) working hours;
- (iii) Nineteen thousand five hundred (19,500) hours of service but less than fortyeight thousand seven hundred and fifty (48,750) hours - one hundred and eighty-seven point five (187.5) working hours;
- (iv) More than forty-eight thousand seven hundred and fifty (48,750) hours of service two hundred and twenty-five (225) working hours;
- (v) The minimum period of vacation that can be taken is one (1) hour.

16.02 Length of Vacation (Public Health Nurses)

Public Health Nurses employed by the Province of Newfoundland and Labrador shall be entitled to annual vacation in accordance with the following:

- (i) Less than one (1) year one and one quarter (1 1/4) working days for each month of service:
- (ii) One (1) year or more but less than ten (10) years three (3) weeks;

- (iii) More than ten (10) years but less than twenty-five (25) years of service -four (4) weeks;
- (iv) More than twenty five (25) years of service five (5) weeks.

16.03 Vacation Period

All employees shall receive their annual vacation between May 1 and October 31, both dates inclusive. An employee shall be entitled to request annual vacation other than between May 1 and October 31, which request shall not be unreasonably denied. The Employer shall post the vacation schedule by May 1 of each year.

16.04 **Selection of Vacation Dates**

Employees in each nursing unit, in consultation with their supervisor, shall determine the method of selecting vacation dates. In the event that majority agreement cannot be reached, preference in vacations shall be regulated according to a rotation plan. The initial placing of employees in the rotation plan will be in accordance with seniority, thereafter, the rotation will proceed without regard to seniority.

16.05 Choice of Vacations

Employees shall be granted a choice of vacation in accordance with 16.03 and 16.04 provided that once the schedules have been posted they shall not be changed except by mutual consent between the employee and the Employer. Vacations shall be given consecutively except by mutual consent between the employee and the Employer. Vacation for the purposes of this Article shall include all current, accumulated and accrued annual leave. Employees shall not be compelled to take more annual leave than they request.

16.06 **Annual Leave Pay**

An employee who has earned at least two (2) weeks annual leave, upon giving at least three (3) weeks notice prior to the pay day preceding the day on which they wish to receive their advance payment, shall receive, prior to the commencement of their annual leave, any regular pay cheque(s) which may fall due during their annual leave.

16.07 Annual Leave Commencement

Whenever possible, an employee shall be entitled to receive their regular scheduled days of rest immediately before the commencement of their annual leave.

16.08 **Refund of Overdrawn Vacation Pay**

Any employee who leaves the employ of the Employer before the end of a vacation year in which they have taken their vacation shall have the applicable proportion of their salary recovered from them in accordance with Schedule "E".

16.09 Anticipated Vacation

Subject to Clause 16.08, any employee who has successfully completed their probationary period may anticipate their vacation to the end of the current vacation year.

16.10 Maximum Annual Leave Accumulation

(a) **Eight (8) Hour Shifts**

An employee may carry forward to another year any proportion of annual leave not taken by them in previous years until, by so doing, they have accumulated a maximum of:

- (i) Twenty (20) days annual leave, if they are eligible for twenty (20) days in any year;
- (ii) Twenty-five (25) days annual leave, if they are eligible for twenty-five (25) days in any year;
- (iii) Thirty (30) days annual leave, if they are eligible for thirty (30) days in any year.

Employees who are prohibited from taking annual leave because of workers' compensation benefits or extended sick leave shall be allowed to carry forward additional days.

(b) Twelve (12) Hour Shifts

An employee may carry forward to another year, any proportion of annual leave not taken by them in previous years, until by so doing they have accumulated a maximum of:

- (i) One hundred and fifty (150) hours annual vacation, if they are eligible for one hundred and fifty (150) hours in any year;
- (ii) One hundred and eighty-seven point five (187.5) hours annual vacation, if they are eligible for one hundred and eighty-seven point five (187.5) hours in any year;
- (iii) Two hundred and twenty-five (225) hours annual leave if they are eligible for two hundred and twenty-five hours in any year.

Employees who are prohibited from taking annual leave because of workers' compensation benefits or extended sick leave shall be allowed to carry forward additional hours.

16.11 Calculation of Length of Vacation

For the purpose of calculation of length of annual vacation with pay, it is agreed that an employee's service will be that service performed in the twelve (12) month period currently used by hospitals for such calculation. It is understood that this period may vary among hospitals (e.g. January 1 - December 31, April 1 - March 31, or July 1 - June 30), and that no hospital will change its currently accepted accumulation period without prior discussion with the Union.

16.12 Substitution of Vacation for Sick Leave

An employee who qualifies for sick leave under Article 18 while on vacation may change the status of their leave to sick leave effective the date of notification to the Employer. The employee shall submit on their return to duty a medical certificate stating the total period during which they qualified for sick leave.

16.13 Substitution of Vacation for Compassionate Leave

An employee who, while on vacation qualified for compassionate leave, shall be credited the appropriate number of days to their vacation leave provided they notify the Employer, in writing, within five (5) calendar days of their return to duty.

16.14 Change of Vacation Entitlement

When an employee becomes eligible for a greater amount of vacation leave, they may be allowed in the year in which the change occurs, a portion of the additional leave for which they have become eligible based on the ratio of the unexpired portion of the year to twelve (12) months, computed in full working days.

16.15 Vacation Leave on Resignation or Transfer

(a) On Resignation

When an employee resigns or dies before the end of the year, they or their estate shall receive payment for the proportionate part of their unused vacation credits.

(b) Transferring Vacation Leave

Subject to Clause 39.01, employees transferring or accepting employment under the transfer and portability Article, may upon termination, receive pay for all or part of their accumulated vacation leave.

16.16 Vacation Credits for the First and Last Month of Employment

For the purpose of this Article, an employee who is paid full salary or wages in respect of fifty (50) percent or more of the days in the first or last calendar month of their service shall in each case, be deemed to have had a month of service.

16.17 Compensation for Work Performed During Vacation

Should an employee agree to work during their approved annual leave, when requested by the Employer, they shall be paid in addition to their annual leave pay, time and one half (1 1/2) the regular rate of pay for each day in which they performed any work.

16.18 Leave of Less Than One Week – Annual Leave, Compensatory Time Off or Rescheduled Statutory Holidays

If an employee requests annual leave, compensatory time off or rescheduled statutory leave before the work schedule is posted, such requests shall not be unreasonably denied by the manager or designate and the manager or designate shall notify the employee before the work schedule is posted. While employees are encouraged to request leave before the schedule is posted, if an employee requests annual leave, compensatory time off or rescheduled statutory leave after the work schedule is posted, the Employer shall attempt to accommodate the employee's request and the response will either approve the leave, deny the leave, or indicate the factors on which the approval of leave may depend.

Article 17 - Holidays

17.01 **Designation of Holidays**

- (a) Employees, other than Public Health Nurses, employed by the Province of Newfoundland and Labrador shall receive seven point five (7.5) working hours paid leave for each of the nine (9) holidays as follows:
 - (a) New Year's Day
 - (b) Good Friday
 - (c) Commonwealth Day
 - (d) Memorial Day
 - (e) Labour Day
 - (f) Christmas Day
 - (g) Boxing Day
 - (h) Plus two (2) additional holidays in each year as mutually agreed between the majority of employees as defined in this agreement and the Employer by December 31st for the next calendar year.

In any calendar year in which mutual agreement cannot be reached, the two (2) additional holidays shall be a Thanksgiving Day and Armistice Day.

(b) Public Health Nurses employed by the Province of Newfoundland and Labrador shall receive one (1) working day paid leave for each of the designated fourteen (14) statutory holidays as follows:

(a)	New Year's Day	(g)	Memorial Day
(b)	St. Patrick's Day	(h)	Orangeman's Day
(c)	Good Friday	(i)	Labour Day
(d)	St. George's Day	(j)	Thanksgiving Day
(e)	Commonwealth Day	(k)	Armistice Day
(f)	Discovery Day	(1)	Christmas Day
		(m)	Boxing Day

One (1) additional holiday as mutually agreed in each hospital by the parties.

17.02 **New Holidays**

Should any new holiday not routinely scheduled be specifically proclaimed by Provincial or Federal authorities, it shall be granted to employees within the scope of this agreement.

17.03 Work on Designated Holiday

When an employee is required to work on a designated holiday, they shall be entitled to, at their option, time off on the basis of one and one-half (1 1/2) hours for each hour worked or pay at the rate of time and one-half (1 1/2) for each hour worked. Should the employee elect to take the time off they shall be granted such time off within two (2) months of the designated holiday. If the time off is not granted within the two month period they shall receive pay in lieu at the rate of one and one-half (1 1/2) times their regular rate of pay for all hours worked on the holiday.

17.04 Statutory Holiday During Vacation Period

If a paid holiday falls or is observed during an employee's vacation period, the employee shall be allowed those additional vacation hours with pay at a time to be mutually agreed between the employee and their supervisor.

17.05 Statutory Holiday on Day of Rest

When a statutory holiday falls on an employee's day of rest and they are required to work on such holiday they shall receive two and one-half (2 1/2) hours off for each hour worked or at their option two and one-half (2 1/2) hours pay for each hour worked.

17.06 Compensation for Holidays Falling on Saturday

For the purpose of this agreement when any of the aforementioned holidays falls on a Saturday and is not proclaimed as being observed on some other day, shift workers shall observe the day proclaimed as a holiday for calculation of benefits under Article 17 - Holidays. All other employees shall observe the following Monday as a holiday.

17.07 Compensation for Holidays Falling on a Sunday

For the purpose of this agreement when any of the above noted holidays falls on a Sunday and is not proclaimed as being observed on some other day, shift workers shall observe the day proclaimed as a holiday for calculation of benefits under Article 17 - Holidays. All other employees shall observe the following Tuesday, (where the preceding Clause already applied to Monday) as the holiday.

17.08 Christmas and New Year's Leave

An employee scheduled to work on Christmas Day shall not be scheduled to work on New Year's Day and shall receive New Year's Eve as a scheduled day of rest. An employee scheduled to work on New Year's Day shall not be scheduled to work on Christmas Day and shall receive Christmas Eve as a scheduled day of rest unless otherwise mutually agreed.

A casual employee scheduled to work on Christmas Day shall not be scheduled to work on New Year's Eve and New Year's Day. A casual employee scheduled to work on New Year's Day shall not be scheduled to work on Christmas Eve and Christmas Day unless otherwise mutually agreed.

17.09 **Statutory Holiday During Sick Leave**

If an employee is sick on a day that has been designated by the Employer as a statutory holiday in accordance with Clause 17.01, the employee shall be paid for the statutory holiday for that day and there shall be no deduction from the employee's sick leave for that day.

17.10 Statutory Holiday While on Workers' Compensation

If an employee is on Workers' Compensation benefits on a day that has been designated by the Employer as a statutory holiday in accordance with Clause 17.01, the employee shall be paid for the statutory holiday for that day and not be considered as being on workers' compensation benefits for that day.

Article 18 - Sick Leave

18.01 (a) **Annual Paid Sick Leave**

An employee shall be entitled to fifteen (15) working hours of sick leave per one hundred and sixty-two point five (162.5) hours of service accumulating to a total of one thousand eight hundred (1,800) working hours. Sick leave previously accumulated shall be retained. The minimum period of sick leave that can be taken is one (1) hour.

- (b) Notwithstanding Clause 18.01(a), an employee hired after December 1, 2006 is eligible to accumulate sick leave at the rate of seven and one half (7.5) hours per one hundred and sixty-two and one half (162.5) hours of service accumulating to a total of one thousand eight hundred (1,800) hours. The minimum period of sick leave that can be taken is one (1) hour.
- (c) For employees hired after December 1, 2006 the maximum amount of sick leave which may be awarded to an employee in any consecutive twenty (20) year period shall not exceed one thousand eight hundred (1,800) hours.

*18.02 **Proof of Illness**

- (a) Sick leave with full pay in excess of three (3) consecutive days shall not be awarded to an employee unless they have submitted in respect thereof, a medical certificate. In cases of an established pattern of sickness, the Employer reserves the right to request a medical certificate for any period of sickness. An employee shall have the option of being attended to by a doctor of their choice and under no circumstances will an employee be penalized in any way by the Employer for exercising their option of being attended to by their personal physician.
- *(b) Employees with an ongoing medical condition(s) may submit a medical certificate at the beginning of each calendar year. If the Employer is satisfied with the medical certificate, the Employee shall be exempt from providing medical certificates in accordance with Article 18.02 (a) for each day of illness regarding the ongoing medical condition(s) for the remainder of the calendar year.

18.03 Expiration of Paid Sick Leave

When an employee has reached the maximum of the sick leave which may be awarded them, they shall, if they are still unfit to return to duty, proceed at their option on annual leave (including current, accumulated and accrued leave) if they are eligible to receive such leave, or if not, on special leave without pay to a maximum of one (1) year.

Employees shall continue to accumulate seniority while on special leave without pay to a maximum of one (1) year under this Clause.

18.04 Sick Leave for Preventative Medical Care

Employees may be allowed to take sick leave in order to engage in personal preventative medical and dental care. The employee shall be required to show proof of having received such care. The employee shall also provide as much notice to the Employer as is reasonably possible and endeavour to a reasonable extent to schedule appointments during off duty hours.

18.05 Sick Leave Record

In January, the Employer shall post the amount of sick leave accumulated to each employee's credit and the amount of sick leave used in the previous year by each employee. These lists shall include up to the end of December 31st of the previous year.

18.06 **Group Life and Extended Health Plan**

The Employer shall pay fifty (50) percent of the premium of an extended Health and Group Life Insurance Plan and the employee will pay fifty (50) percent. The Employer shall provide a copy of the Insurance Plan to the Union within thirty (30) days of the signing of this agreement.

18.07 Fringe Benefits

An employee shall continue to accumulate sick leave benefits while on a leave of absence which is required by the Employer. Employees who are off on sick leave shall continue to accrue all fringe benefits.

18.08 Sick Leave Credits for the First and Last Month of Employment

For the purpose of this Article, an employee who is paid full salary or wages in respect of fifty (50) percent or more of the days in the first or last calendar month of their service shall, in each case, be deemed to have had a month of service.

18.09 **Injury on Duty**

An employee who is injured during working hours and is required to leave for treatment or sent home for such injury, shall receive payment for the remainder of the shift or work day at their regular rate of pay without deduction from sick leave, provided that a medical practitioner, the staff health officer, or the appropriate Management representative, stated that the employee is unfit for further work.

18.10 Extension of Sick Leave

(a) **Eight (8) Hour Shifts**

An employee with more than five (5) years of service who has exhausted their sick leave credits may be allowed in the event of illness, an extension of their sick leave to a maximum of fifteen (15) working days. This sick leave extension shall be repaid by the employee upon their return to duty from their normal monthly accumulation.

(b) Twelve (12) Hour Shifts

An employee with more than five (5) years of service who has exhausted their sick leave credits may be allowed in the event of illness, an extension of their sick leave to a maximum of one hundred and twelve point five (112.5) hours. This sick leave extension shall be repaid by the employee upon their return to duty from their normal monthly accumulation.

Article 19 - Maternity/Adoption/Parental Leave

19.01 (a) **Request for Maternity Leave**

An employee shall be eligible for and shall be permitted to commence maternity leave at the beginning of the 6th month of pregnancy. Permission to commence maternity leave shall not be unreasonably denied.

(b) Request for Adoption Leave

An employee shall notify their Employer when an adoption is imminent. The employee shall be eligible for and shall be permitted to commence adoption leave immediately following the adoption.

(c) Request for Parental Leave

An employee shall be eligible for and shall be permitted to take parental leave in combination with maternity or adoption leave or at some other time (by either parent).

(d) Subject to 19.06, 19.07, the maximum leave allowed under this Clause shall be seventy-eight (78) weeks in total. However, the Employer may grant leave without pay when an employee is unable to return to duty after the expiration of leave under this Clause, during which the employee shall earn service for seniority purposes.

19.02 **Dismissal**

Pregnancy will not constitute cause for dismissal. The Employer retains the right to require an employee to commence maternity leave if the state of their health becomes incompatible with the requirements of the job.

19.03 **Return to Work**

An employee may return to duty provided they have given the Employer two (2) weeks' notice of their intention to do so.

19.04 Vacation Before or After Leave

At their request, an employee may be awarded vacation immediately before or immediately following maternity/adoption/parental leave.

19.05 **Period of Protection**

An employee's position as defined in 20.01, will be protected for the total period of their maternity/adoption/parental leave.

19.06 Service While on Maternity/Adoption/Parental Leave

While on maternity/adoption/parental leave, an employee shall continue to accumulate service for seniority, step progression and length of vacation in accordance with Article 16.01 to a maximum of seventy-eight (78) weeks. While on maternity/adoption/parental leave, an employee shall continue to accumulate service for annual leave, and sick leave accrual to a maximum of fifty-two (52) weeks. Should an employee change employment status at the employee's request (i.e. full time to part time or part time to full time) prior to or during the period of leave under this Article, the benefits outlined above will be credited on the basis of the new hours of work effective the date of the change in status.

19.07 Group Insurance While on Maternity/Adoption/Parental Leave

Employees on maternity/adoption/parental leave will have the option of continuing to pay their portion of the group insurance plan premiums. Where the employee continues to pay their premiums, the Employer shall continue to pay its portion of the group insurance plan premiums while the employee is on maternity/adoption/parental leave to a maximum of seventy-eight (78) weeks.

19.08 Standby, Callback and Overtime - Pregnant Employees

Pregnant employees may obtain upon request an exemption from the performance of standby, callback, or overtime provided a sufficient number of employees in the work area agree to share the necessary standby, callback and overtime duty.

19.09 Standby, Callback and Overtime - Breastfeeding

An employee returning from maternity leave or parental leave and still breastfeeding their child may be exempt from standby, callback, and overtime until the child is one (1) year old provided that a sufficient number of employees in the work area agree to share the necessary standby, callback and overtime.

19.10 **New Positions**

An employee who applies for a position in accordance with Article 24 while on maternity/adoption/parental leave shall be considered for that position in accordance with the provisions of Article 24. If the employee on maternity/adoption/parental leave is successful, their trial period shall start upon their return to work.

*19.11 **Job Postings**

An employee who is on maternity/adoption/parental leave shall have electronic access to internal job postings prepared in accordance with Article 24.

19.12 Illness Associated with Pregnancy

An employee may be awarded sick leave for illness that is a result of, or may be associated with, pregnancy provided that they are not on maternity leave.

Article 20 - Position Protection

20.01 Position

For the purpose of this Article "position" means:

- (a) the particular nursing unit or hospital department (e.g. OR, Medical, OPD, Nursery, etc.) where an employee was regularly employed before their leave of absence; or,
- (b) where an employee was not employed in a "Nursing Unit" situation their "position" would be considered the function or group of functions which they regularly carried out before their leave of absence (e.g. clinic nurse, team leader, day care nurse, staff health nurse, liaison nurse, etc.).

20.02 **Period of Protection**

Employees who have been granted leave in accordance with Articles 16, 18, 21 and 22 will have their positions protected for the period of the approved leave up to a maximum of twenty-four (24) consecutive months unless the position is abolished. Where the

position has been abolished the employee's classification and salary will be protected for the period of the approved leave.

20.03 **Protection for Education Leave**

Notwithstanding Clause 20.02, employees who have been granted special leave for educational purposes in accordance with Article 22 will have their positions protected for the total period they are on special leave.

20.04 Protection for Employees on Workers' Compensation

Subject to Article 23, employees who are on Workers' Compensation shall have their positions protected for the duration of time they are on Workers' Compensation.

*20.05 Protection for Permanent Employees Awarded Temporary Positions

The Employer shall endeavour to provide position protection when requested by a permanent employee who has been awarded a temporary position. Should the Employer be unable to approve position protection, upon request, the employee will be informed, by the respective Manager, of the reason(s) why they were unsuccessful.

Position protection shall be granted to permanent employees who have been awarded temporary positions when:

- (i) the employee has a minimum of two (2) years of service in their current position; and
- *(ii) the temporary position has a higher rate of pay, or an increased number of biweekly hours, or no scheduled night shifts (where the employee's current position includes night shifts); and
- (iii) the temporary position is of a nine (9) months duration or greater.

Permanent employees who are granted position protection shall be released when the position is backfilled or at an earlier date at the discretion of the Employer. In cases where there are less than twenty (20) weeks remaining in the temporary position when the replacement is secured, position protection may be rescinded at the discretion of the Employer or the employee may elect to remain in their current position.

Article 21 - Compassionate Leave

*21.01 Compassionate Leave

Compassionate leave with pay shall be awarded to an employee as follows:

- *(a) In the case of the death of an employee's parents, siblings, children, spouse, common-law-spouse, children of common-law spouse, legal guardian, grandparents, grandchild, children-in-law, parents-in-law, step relatives, foster parents, or near relative living in the same household, three (3) working days; and
- *(b) In the case of their siblings-in-law, one (1) working day.
- (c) If the death of a relative referred to in Clause 21.01 (a) occurs outside the Province, the employee may be granted leave with pay not exceeding four (4) working days for the purpose of attending the funeral.

- (d) In cases where extraordinary circumstances prevail, the Employer at their discretion may grant special leave with pay for bereavement up to a maximum of two (2) working days in addition to that provided in Clauses 21.01 (a) and (c).
- (e) For the purpose of Clause 21.01, a working day is defined as the day that the employee would have been working on the date for which the leave is requested.
- *(f) Upon request of the employee, the Employer shall endeavor to provide leave not covered in (a) and (b) for compassionate reasons and employees may avail of paid leave banks or unpaid leave for this purpose.

21.02 Special Leave Without Pay

Special compassionate leave without pay may be granted with the approval of the Employer.

Article 22 - Special Leave

22.01 Special Leave Without Pay

An employee may be granted special leave without pay in exceptional circumstances provided they have no current or accumulated annual leave available to them, provided they return to the employ of the Employer, they shall retain accumulated sick leave, seniority, vacation entitlement and salary. While on such leave, employees shall continue to accumulate seniority.

22.02 Paid Jury or Court Witness

- (a) The Employer shall grant a leave of absence without loss of pay, seniority, or accumulated benefits to an employee who is summoned for jury service, or serves as a juror or who is subpoenaed to attend upon a court as a witness in any court.
- (b) If the employee is required to be in court as a witness in any matter arising out of their employment with the Employer on their scheduled day of rest or after their regularly scheduled shift, they shall be compensated at their regular hourly rate or time off in lieu of pay for each hour spent in court on their day of rest or after their regularly scheduled shift to a maximum of seven and one half (7 1/2) hours per day.
- (c) When requested, the employee shall present proof that they attended as a juror or a witness pursuant to sub-paragraph (a) or sub-paragraph (b).
- (d) Where an employee appears before a court for reasons other than those stated in (a) and (b) above, they shall be granted a leave of absence without pay or allowed to take annual leave or time off in lieu of overtime they may have to their credit.

22.03 Educational Leave

An employee may be granted educational leave without pay for varying periods, which leave shall not be unreasonably withheld, provided a suitable replacement is available to work. Except in the case of extenuating circumstances, any employee desiring educational leave shall apply to the appropriate manager or their designate in writing a minimum of two (2) months prior to the anticipated date of commencement of the leave. Any employee

desiring to terminate their educational leave early shall give a minimum of one (1) month's notice in writing indicating the date the employee anticipates returning to work. While on education leave, an employee:

- (i) shall retain and continue to accumulate seniority; and
- (ii) shall retain all other benefits of the collective agreement accumulated prior to the date of commencement of their leave.

22.04 Extended Unpaid Leave

Upon written request, a permanent employee who has completed two (2) years of service shall be granted leave up to an initial maximum of twelve (12) months without pay and without loss of accumulated seniority and benefits provided that such leave shall not cause an unreasonable interference with the Employer's operation and that a suitable replacement is available. An employee shall be entitled up to a maximum of twelve (12) months unpaid leave for each two (2) years of service with the understanding that an employee with an additional two (2) years of service may request an additional twelve (12) months of leave (maximum of twenty-four (24) consecutive months leave) without the necessity of having to return to work. Such an extension must be by mutual agreement between the employee and the Employer. Employees shall not be subject to any benefits of this agreement during this period, except that while on such leave, employees shall continue to accumulate service for seniority purposes only. The minimum amount of unpaid leave an employee may request under this Clause is eight (8) weeks.

22.05 **Deferred Salary Leave Plan**

The deferred salary leave plan as outlined in Schedule I shall be effective January 1, 1995.

Article 23 - Seniority, Layoff and Recall

23.01 **Seniority Defined**

Subject to Clause 33.09 and 33.11 seniority is defined as the length of continuous service (excluding overtime) with the Employer and shall date from the last entry into employment with the Employer.

23.02 **Seniority Roster**

Subject to Clause 23.01, a seniority roster of all employees covered by this agreement who have seniority shall be posted by the Employer before January 31 of each year in a place accessible to all employees so affected, showing the employee's name, classification, last date of hiring, the actual seniority earned by the employees expressed in hours or seniority numbers if applicable, and the employee's employment status (full time, part time, temporary or casual).

23.03 Protest Period

The seniority roster shall be turned over to the Union or a person designated by the Union after a thirty (30) day protest period following each posting.

23.04 Layoff and Recall

Layoff and recall shall be in accordance with Schedule "N".

23.05 Advance Notice of Layoff

Employees shall only be laid off because of lack of work. The Union shall be given forty-eight (48) hours notice of the layoff of any employee within the scope of this agreement. Unless legislation is more favourable to the employee, the Employer shall notify permanent employees who are to be laid off thirty (30) calendar days prior to the date of layoff and temporary employees fourteen (14) calendar days prior to the date of layoff.

If the employee has not had an opportunity to work the days as provided in this Article, they shall be paid for the days for which work was not made available. An employee who is bumped by another employee shall be deemed to have been given the notice on the same date the notice was given to the senior employee whose position is affected.

23.06 Loss of Seniority on Suspension

Seniority shall be forfeited by an employee if they are suspended for just cause, in which event, the loss of seniority shall be for the period of suspension.

*23.07 Loss of Seniority

Seniority shall be forfeited by an employee and they shall be deemed to be terminated for any of the following reasons:

- (a) they resign;
- (b) they are discharged for just cause;
- (c) they have been laid off for a period of two (2) years or more and/or recall rights have expired;
- (d) they fail to return to work upon recall in circumstances where they were obligated to accept recall; or
- *(e) they are a casual employee and one (1) year has elapsed from the last date of work, except where mutually agreed between the Parties.

23.08 No New Employee(s)

No new employee(s) shall be employed to fill a vacant nursing position in the bargaining unit until all those who have been laid off have been given the opportunity of reemployment.

23.09 Transfer and Seniority Outside the Bargaining Unit

(a) No employee shall be transferred to a position outside the bargaining unit without their consent. If an employee is transferred to a position outside the bargaining unit, they shall retain their seniority accumulated up to the date of leaving the unit, but will not accumulate any further seniority while outside the unit.

- (b) An employee who transfers outside the bargaining unit and who obtains a casual or temporary position within the bargaining unit will accumulate seniority in accordance with Clause 23.01 from the date of appointment to a casual or temporary position but will not be credited with their former bargaining unit seniority until such time as they obtain a permanent position.
- (c) An employee who accepts a temporary management position(s) will be granted position protection from their bargaining unit position for a maximum of two cumulative years, provided the Employer posts the resulting bargaining unit position on a temporary basis. The Employer shall provide prior written notice of both the commencement and termination dates of the temporary position to the Union and branch.
- (d) An employee of any Employer included in Schedule C who obtains a permanent position with another Employer included in Schedule C, and has successfully completed the probationary period, will be immediately credited with their bargaining unit seniority accumulated with the former Employer, and will continue to accumulate seniority in accordance with Clause 23.01 from the date of appointment with the new Employer.
- (e) An employee of any Employer included in Schedule C who obtains a casual or temporary position with another Employer included in Schedule C, will accumulate seniority in accordance with Clause 23.01 from the date of appointment with the new Employer but will not be credited with their former bargaining unit seniority until such time as a permanent position is obtained with the new Employer and they have successfully completed the probationary period.

Subject to Clause 23.09 (d) and (e) the parties agree that no employee can be awarded more than equivalent full time seniority and that no former employee of any Employer included in Schedule C can use their former bargaining unit seniority to obtain a position with another Employer.

23.10 **Probation For Newly Hired Employees**

After completion of the probationary period, seniority shall be effective from the last date of hire.

23.11 End of Temporary Employment

Unless granted position protection in Article 20.02, the following shall apply when:

- (a) the temporary position ends prior to the date specified in their letter of appointment, the employee shall have recall rights until the date specified in their letter of appointment and be assigned casual status; or
- (b) the temporary position ends, as specified in the letter of appointment, the temporary employee shall be assigned to casual status.
- (c) the Employee shall retain all seniority hours and accumulated service while occupying the temporary position.

Subject to Articles 9.01 and 16.10, when an employee is assigned a casual position, they have the option of requesting payment of accumulated benefits including annual leave, statutory holidays, overtime or compensatory time off. Such payment shall be received by the employee within thirty (30) days of their request. When the employee does not elect this option, all accumulated benefits including annual leave, statutory holidays, overtime or compensatory time off shall be maintained in the employees leave banks and shall be accessed upon reversion to a temporary or permanent position, or upon termination of employment. Sick leave accumulated during a temporary position shall be maintained in the employee's bank and shall be accessed upon appointment to a temporary or permanent position.

23.12 **Temporary Employee Commitment**

Employees accepting temporary positions will not be able to apply for other temporary positions until two (2) months prior to the end date specified in their initial letter of appointment to the temporary position unless the employee is applying for a position:

- (i) with a higher rate of pay; or
- (ii) with an increased number of bi-weekly hours; or
- (iii) without a scheduled night shift.

Article 24 - Promotions and Staff Changes

24.01 **Effective Date**

Changes in pay rates as a result of a promotion shall be effective from the date of the promotion stated in the letter of appointment.

24.02 **Posting of Vacancies**

All vacancies and all newly created positions in the classifications covered by this agreement, shall be posted electronically (where possible) and/or on all bulletin boards provided for a minimum of ten (10) calendar days.

24.03 Contents of Postings

- (a) The notice posted within the hospital shall contain the following:
 - (i) classification and salary scale;
 - (ii) qualifications, including required knowledge, education and any skills necessary to perform the job; and
 - (iii) whether shift work could be involved.

Such qualifications may not be established in an arbitrary or discriminatory manner.

(b) Copies of all postings shall be forwarded to the branch president or their designate at the time of posting.

24.04 Awarding of Positions

- (a) All positions within the Level 1 Registered Nurse 1 series and Psychiatric Registered Nurse 1 positions shall be considered on the basis of seniority, fitness and qualifications before appointments are made.
- (b) In making all other staff changes, primary consideration shall be given to qualifications, ability and fitness to perform the required duties. Where qualifications, ability and fitness are equal, seniority as defined in Article 23 shall prevail.

24.05 Notification of Successful Applicant

Within five (5) days of the date of an appointment to a position, applicants and the Branch President (or designate) will be notified of the successful applicant.

24.06 **Trial Period**

(a) (i) **Eight (8) Hour Shifts**

The employee who accepts a promotion to a higher position is entitled to a trial period of up to sixty (60) working days. During this period, the employee may return to their former position and salary without loss of seniority.

(ii) Twelve (12) Hour Shifts

The employee who accepts a promotion to a higher position is entitled to a trial period of up to four hundred and fifty (450) working hours. During this period, the employee may return to their former position and salary without loss of seniority.

(b) Casual employees who obtain a permanent position or a temporary position for more than sixty (60) working days (450 working hours), may be required to work a trial period. The length of time of the trial period will be as outlined in (a) above. No casual or temporary employee shall be required to complete a second trial period in the same unit or work area within a twelve (12) month period.

24.07 Return to Former Position

- (a) If, in the opinion of the Employer, the successful applicant proves unsatisfactory in the position during the aforementioned trial period, they shall be returned to their former position and salary without loss of seniority and any other employee promoted or transferred because of the rearrangement of the position shall be returned to their former position and salary without loss of seniority.
- (b) A casual employee who is unsuccessful in their trial period shall return to their casual status.

24.08 **Temporary Vacancies**

No position of a permanent nature will be filled by keeping temporary employees or part time employees on to fill the position without having the position posted.

*24.09 **Displacement**

Displacement designates any temporary change of position of an employee affected at the request of the Employer. If the displacement requires the employee to move to another place of employment, it shall not exceed one month unless mutually agreed otherwise.

- *(a) The employee must accept displacement in the case of:
 - (i) increased or decreased work load:
 - *(ii) unexpected absence resulting in an urgent or imperative need for personnel;
 - (iii) emergencies.

In all other cases of displacement there must be mutual agreement between the employee and the Employer.

- *(b) In all cases of displacement, the Employer should not displace any employee if the utilization of other means is expedient. The Employer shall seek volunteers before displacement.
- *(c) Displacement should be carried out in an equitable manner among the employees.
- *(d) The Employer will endeavor not to displace an employee greater than one hundred (100) kms.
- *(e) Where an employee is displaced, per diem meal allowance, accommodation and all travel related expenses shall be paid in accordance with Article 37.

24.10 Reduction of Hours of Work

- (a) Where a permanent full time employee wishes to reduce their hours of work to part-time, they may approach their Employer with their request indicating the number of shifts each week they are prepared to work. Should the Employer agree to consider this request, the Employer will fill the remaining shifts in accordance with Article 24. If an applicant is selected in accordance with Article 24 for the available shifts on a part-time basis, the request of the full time employee shall be granted.
- (b) With the consent of the Employer, a permanent part-time employee may reduce their hours of work. Any remaining shifts may be posted at the discretion of the Employer.
- (c) With the consent of the Employer, a permanent full-time employee or a permanent part-time employee who wishes to reduce their hours of work on a temporary basis of two (2) years or less may do so, provided the Union and Branch are given prior written notice of both the commencement and termination dates by the Employer. The Employer shall endeavor to fill the remaining

bargaining unit shifts. Furthermore, the parties recognize that any employee hired to replace a nurse, who temporarily reduces their hours of work, shall be considered to meet the definition of temporary employee.

Article 25 – Salaries

25.01 Receive Salaries as Per Schedule A

Subject to Clause 25.02, all employees covered by this agreement shall receive salaries as specified in Schedule "A".

25.02 Unregistered Nurses

- (a) Subject to the provisions of Schedule "A", unregistered nurses shall receive a salary of nine hundred and fifty dollars (\$950.00) below the minimum salary of a Nurse I and such nurses shall not advance up the salary scale nor hold a position above a Nurse I level.
- (b) If a registered nurse was registered in another Canadian Province or another country, they shall be entitled to the benefits of experience credits contained in Schedule "B" upon confirmation of a registration in this Province.

25.03 Isolation Allowance

An isolation allowance of six hundred and sixty dollars (\$660.00) per annum shall be paid on a prorated basis to all employees working in the following areas: Burgeo, Fogo and Harbour Breton and other areas as may be mutually agreed upon by the Minister of Health and Community Services and the Union. Employees currently receiving more than a six hundred and sixty dollars (\$660.00) allowance will continue to receive that amount.

25.04 Labrador Benefits

The Labrador benefits shall be paid to employees covered by this agreement who are eligible to receive such benefits as outlined in the Labrador Agreement as outlined in Schedule "H". If the rates are increased for any of the signatories to the Labrador Benefits Agreement, they will also be increased for the RNUNL.

25.05 **Reclassification**

No employees shall suffer loss of pay as a result of reclassification.

Article 26 - Pension Plan

26.01 **Pension Plan**

- (a) The Employer shall make available a pension plan for employees covered by this Agreement who are eligible to participate in such a plan.
- (b) The Employer shall make available a Money Purchase Pension Plan for employees covered by this agreement who are eligible to participate in such a plan.
- (c) The union shall be entitled to copies of the plans upon request.

Article 27 - Staff Development

27.01 List of Duties and Responsibilities

The Employer shall provide each new employee with a listing of the duties and responsibilities for their work area within two (2) weeks of their date of hire.

27.02 **Orientation Program**

The Employer shall provide an orientation program of not less than two (2) weeks to include essential information such as agency policies, procedures, routines, location of supplies and equipment, fire and disaster plans, etc.

27.03 **In-Service Program**

- (a) The Employer shall provide an in-service program on a continuing basis focused on the needs of the staff for the improvement of patient care. When employees are required to attend compulsory in-service programs outside of the employees' normal hours of work, all such time in attendance shall be paid at the applicable overtime rate.
- (b) When selecting topics for in-service education, the Employer will give consideration to the following programs:
 - (i) CPR
 - (ii) Fire Safety
 - (iii) Evacuations
- (c) The Employer shall make available an in-service program in the Prevention of Back Injuries.

27.04 In-Service Program - Specialty Units

The Employer shall provide an adequate in-service training program for new employees entering specialty units.

27.05 Interviewing New Employees

A representative of the Union shall be given an opportunity to interview each new employee within regular working hours without loss of pay during the first month of employment for the purpose of acquainting the new employee with the benefits and responsibilities of union membership. Where possible such interviewing will take place on a group basis during the orientation program for new employees. Interviews shall be held at a time mutually agreeable between the Employer and the union representative in the hospital.

27.06 **Bulletin Boards**

A bulletin board shall be made available for use by the Union for posting information on current events, meeting notices, etc. in a place accessible to all employees.

27.07 Occupational Health and Safety Committee

- (a) The Employer and the Union recognize the importance of health and safety in the workplace. It is the intention of both parties to achieve and maintain through mutual cooperation, a healthy and safe working environment.
- (b) There shall be an Occupational Health and Safety Committee in the workplace, established and governed in accordance with the *Occupational Health and Safety Act* and regulations with at least one bargaining unit nurse on this committee.
- (c) The nurse on the Occupational Health and Safety Committee shall be selected by the employees and shall not lose regular salary or benefits for the time spent in committee meetings.
- (d) The Occupational Health and Safety Committee shall establish its own guidelines for determining which incidents and injuries are investigated. The Committee will endeavour to ensure that all incidents resulting in lost time Workers' Compensation claims are investigated.

Article 28 - Health Program

28.01 Staff Health Services

A staff health service shall be available in the case of accident, illness or injury while a nurse is on duty.

28.02 Medical Examinations

Medical examinations, x-ray, etc., required by the Employer shall be provided free of charge to the employee through the staff health program.

28.03 Employee's Choice of Physician

An employee shall have the option of being attended to by a doctor of their choice in the situations described in Clauses 28.01 and 28.02 provided there is no expense to the Employer.

28.04 Immunizations

Immunizations which are required as a preventative health measure shall be provided free of charge to all employees.

28.05 Radiological Hazards

The Employer will ensure that radiation levels are monitored in all areas using x-ray equipment and any other device that could expose employees to radiation levels.

Article 29 - Termination of Employment

29.01 Vacation Pay on Termination of Service

Upon termination of service for any cause, an employee will receive vacation pay for all their current annual vacation on a proportionate basis taking into account the actual number of months worked since the beginning of the vacation year per Schedule "E" less any previously taken in the period plus pay for their accumulated vacation carried forward from a previous year.

29.02 Letter of Intent

An employee shall not be required to submit a letter of intent regarding their continued employment with the Employer. The Employer may survey members of the staff to obtain the necessary information to assist in future staff planning.

29.03 **Period of Notice**

Three (3) weeks written notice shall be given regarding termination of employment by the employee unless otherwise reduced or eliminated by mutual agreement between the Employer and the employee.

29.04 **Pay on Termination**

The Employer will endeavour to pay employees all earned benefits on the payday following termination providing the employee gives three (3) weeks written notice of termination.

Article 30 – Discipline

30.01 Employer Actions

An employee who has completed their probationary period may be suspended, discharged, or disciplined but only for just cause.

30.02 **Procedure**

When an employee is disciplined, such employee shall be advised in writing by the Employer of the reason for such discipline. Such disciplinary action will be taken in accordance with Clause 30.03 below. This notification shall be by mail or hand-delivered letter.

30.03 Adverse Report

The Employer shall notify the employee in writing of any dissatisfaction concerning their work within fourteen (14) calendar days of the event of the complaint. This notification shall include particulars of the work performance which led to such dissatisfaction. If this procedure is not followed, the expression of dissatisfaction shall not become a part of their record for use against them at any time. This Article shall apply in respect to any expression of dissatisfaction relating to their work or otherwise which may be detrimental to an employee's advancement or standing with the Employer. The record of the employee shall not be used against them after eighteen (18) months have elapsed, providing another warning or reprimand relating to the same or similar offence has not been given within that period. The employee's written reply to such notification of dissatisfaction shall become

part of their record. A letter of discharge, or a letter of suspension, shall also be considered an adverse report for the purposes of Article 30.

30.04 **Reinstatement**

Where it is determined that an employee has been suspended without pay or discharged in violation of Clause 30.01, that employee shall be immediately reinstated in their former position without loss of seniority or any other benefit which would have accrued to them if they had not been suspended or discharged. One of the benefits which they shall not lose is their regular pay during the period of suspension or discharge which shall be paid to them at the end of the next complete pay period following their reinstatement.

30.05 Personal Record

All reference to a suspension or discharge placed on an employee, who was reinstated under the provisions of Clause 30.04, shall be removed from their personal file and future references for employment requested by the employee or another Employer shall contain no indication of the suspension or discharge.

30.06 Redress

Subject to Clause 30.04, in cases where an employee is found to be unjustly disciplined, redress shall be awarded the employee in such a manner as is acceptable to the Union and the Employer or as directed by an Arbitration Award.

30.07 Work Days

Saturdays, Sundays and Holidays (recognized herein) shall not be included when determining the time within which action is to be taken under Article 30.

30.08 **Employee File**

An employee shall be entitled to their rights under Clauses 5.07 and 5.08.

*30.09 Justice and Dignity Provision

If upon investigation, the Employer feels that disciplinary action is necessary, such action shall be taken in accordance with the provisions of the Collective Agreement. In situations where the Employer is unable to investigate the matter to its satisfaction but feels that the employee should be removed from their place of employment, it shall be with pay.

Article 31 - Dressing Rooms and Parking Facilities

31.01 **Dressing Rooms**

In places of employment where not already provided, dressing rooms, individual lockers and washroom facilities, shall be provided within six (6) months of the signing of this agreement. A lounge, for the use of employees covered by this Agreement, shall be provided in all hospitals and institutions. In hospitals and institutions where the problem of space exists, the Union and the Employer may mutually agree to vary the time limits imposed by this Article.

31.02 **Parking Facilities**

The Employer shall provide, whenever possible, adequate facilities for employees to park their cars during the working hours.

Article 32 - Part-Time Employees

32.01 **Hourly Rate**

The hourly rate for all part time, casual and temporary employees except where they are Public Health Nurses shall be calculated by the following formula:

Annual Salary (Full Time) 1950

The hourly rate for all part time, casual and temporary Public Health Nurses shall be calculated by the following formula:

Annual Salary (Full Time) 1820

32.02 Awarding of Benefits to Part-Time Employees

The benefits awarded to part-time employees shall be prorated to those of full time employees.

32.03 Calculation of Seniority

Seniority of the part time employees shall be calculated on a prorated basis from the date of hiring.

32.04 Letter of Appointment

An employee under this Article shall receive a letter of appointment clearly stating their employment status and the expected duration of employment.

32.05 Part-Time Employees - Additional Shifts

All time worked by part-time employees in excess of equivalent full-time hours on a daily basis or weekly basis shall be considered overtime. After their regular scheduled hours up to equivalent full-time hours, part time employees shall be paid the sum of thirty (30) cents per hour in addition to their regular hourly rate.

Article 33 – Casual Employees

*33.01 **Obligation**

A casual employee, hired prior to date of signing, has no obligation to the Employer to come when they are called and the Employer has no obligation to call any one particular employee.

*33.02 Casual Employee Availability

Effective date of signing, all newly hired casual employees, including existing employees who become casual after date of signing, with the exception of retired nurses who are in

receipt of PSPP or GMPP, are required to accept a minimum of nine hundred (900) hours each fiscal year, unless mutually agreed otherwise.

Of the nine hundred (900) hours:

- (a) (i) A casual employee working twelve (12) hour shifts shall be available to be scheduled to work a minimum of three (3) shifts during the Christmas and New Year's period (December 23 to January 2 day inclusive);
 - (ii) A casual employee working eight (8) hour shifts shall be available to be scheduled to work a minimum of three (3) shifts during the period of December 23 to January 2 inclusive;
 - (iii) The Employer reserves the right to schedule the casual employee's commitments under (a) (i) or (ii) above. The Employer recognizes the employee's available preference over the Christmas period or the New Year's period. The Employer endeavors to recognize the shift preference in meeting the obligation of the casual employee under (a) (i) or (ii) above when the Christmas schedule is developed.

AND:

- (b) At least thirty percent (30%) of the nine hundred (900) hours shall be within the months of June, July, and August.
- (c) At least fifty percent (50%) of the nine hundred hours shall be on weekends, nights and/or evenings (where applicable).
- (d) Should a casual employee who has accepted a temporary position revert to casual status the hours worked in the temporary position shall be credited towards their nine hundred (900) hour commitment.
- (e) Should the need for a casual employee change on the unit(s) for which a shift(s) was originally booked before the commencement of or during the shift(s) such that the casual employee is no longer required on that unit(s), the Employer shall notify the casual employee and assign them to the new work assignment. The casual employee is obligated to work the new assignment as offered. The Registered Nurse in charge/team lead or designate in the new work area will provide the casual employee with the necessary information to safely perform duties upon arrival to unit.

33.03 **Benefits**

Casual employees shall be entitled to all the benefits of the Collective Agreement except for the following articles:

9.05, 9.06, 11, 16, 17.01, 17.02, 17.03, 17.04, 17.05, 17.06, 17.07, 17.09, 17.10, 18, 19, 20, 21, 22, 23.01, 23.04, 23.05, 23.07(c), 23.07(d), 23.08, 23.11, 24.06(a), 24.07(a), 24.08, 24.09, 24.10, 26.01(a), 29, 32.02, 32.03, 32.05, 35.01, 39.01, 40, 44.01 and 46.

In lieu of the benefits outlined in these articles, the employees shall receive twenty (20) percent of their basic salary as in Schedule "A". Casual employees shall participate in the Government Money Purchase Pension Plan.

33.04 List of Casual Employees

A list of the casual employees used by the Employer in each work area shall be posted in that area.

33.05 Letters of Appointment and Availability

A casual employee shall receive a letter of appointment within thirty (30) days of their appointment to casual status or of accepting a position in accordance with Clause 33.02. This letter shall include a mutually agreeable statement of the casual employee's days and shifts of availability for work of a casual nature. The commitment to availability specified in the letter of appointment shall be subject to mutually acceptable revision. Such revision shall occur once per year or on a more frequent basis if requested by either of the parties.

33.06 Contact Information

It is the responsibility of the casual employee to keep the Employer informed of their current address and telephone number.

*33.07 Notification of Unavailability

Subject to Clause 33.05, two weeks prior to April 1 and October 1, a casual employee shall indicate in writing to their manager or designate periods of unavailability for that six (6) month period and the Employer will not normally call the casual employee during designated period of unavailability.

33.08 Obligation Regarding Pre-Booked Shifts

(i) Obligation of the Casual Employee

The casual employee shall not cancel a pre-booked shift(s) except in cases of illness or just cause.

(ii) Obligation of the Employer

Should the need for a casual employee change on the unit(s) for which a shift(s) was originally booked before the commencement of the shift(s) such that the casual employee is no longer required on that unit(s), the Employer shall notify the casual employee and offer them another work assignment. If the casual employee does not wish to accept the work assignment, the Employer has no further obligation to the casual employee for that shift(s) and the shift(s) will be cancelled.

33.09 **Seniority for Casual Employees**

Seniority for casual employees shall be defined as time worked after January 13, 1995 (excluding overtime) and shall apply only to Clause 24.04.

(i) Seniority for casual employees who have worked 487.5 hours as of January 13, 1995, shall be counted from the date of signing (no retroactivity).

- (ii) Seniority for casual employees who have not worked 487.5 hours as of January 13, 1995, shall be counted only after the casual employee has worked 487.5 hours. After the casual employee has worked 487.5 hours they shall be credited with seniority for time worked after the date of signing.
- (iii) Seniority for casual employees hired after January 13, 1995, shall be counted only after the casual employee has worked 487.5 hours. After the casual employee works 487.5 hours, they shall be credited with 487.5 hours of seniority.
- (iv) Effective date of signing, a casual employee who takes time off work due to maternity, adoption or parental leave or because of a workplace injury for which the employee is in receipt of Workers' Compensation benefits shall continue to accumulate seniority.

Seniority will be calculated by averaging the biweekly seniority earned by the casual employee in the twenty-six (26) pay periods immediately prior to the date of the injury or the start of maternity, adoption or parental leave. If the casual employee has had less than twenty-six (26) pay periods since the last date of hire, then the weekly or daily averages that have actually been worked by the casual employee will be assessed to determine a reasonable seniority calculation. Casual employees will be responsible for notifying the Employer of the start date of their eligibility.

The maximum amount of maternity, adoption or parental leave for which seniority will continue to accumulate will be seventy-eight (78) continuous weeks. The maximum leave restrictions do not apply to casual employees in receipt of Workers' Compensation benefits.

33.10 Work on Statutory Holiday

Casual employees who work on a statutory holiday shall be paid at the rate of time and one-half (1 ½) for each hour worked on the statutory holiday. No payment shall be made for the statutory holiday as outlined in Clause 17.01, as this is covered by the twenty (20) percent in Clause 33.03.

33.11 Seniority for Casual Employees While in Temporary Positions

Subject to Clause 23.11, casual employees who return to casual status following a period in a temporary bargaining unit position, or temporary employees who obtain casual employment following a period in a temporary bargaining unit position, shall retain seniority earned in the temporary position. Such seniority shall be added to seniority earned by the employee while in the casual status (if any) and shall be used only for the purposes of Clause 24.04.

33.12 Casual Employee Unavailable Because of Maternity/Adoption/Parental Reasons

A casual employee who applies for a position in accordance with Clause 24.04 while unavailable because of maternity/adoption/parental reasons for a maximum of seventy eight (78) weeks, shall be considered for the position in accordance with Clause 24.04. If the casual employee is successful, they shall assume the position upon their return to work.

*Article 34 - Nurse Practitioners

*34.01 Nurse Practitioners who act as a first assessor in the Medical Assistance in Dying program (MAID) shall be paid double their regular hourly rate for each hour worked up to a maximum of 5 hours;

Nurse Practitioners who act as a second assessor in the Medical Assistance in Dying program (MAID) shall be paid double their regular hourly rate for each hour worked up to a maximum of 4 hours; and

Nurse Practitioners who complete MAID shall be paid double their regular hourly rate for each hour worked up to a maximum of 5 hours.

The above relates to work performed outside of regular work hours.

Article 35 - Special Allowance

35.01 Uniform Allowance

An employee who is required to wear a uniform not supplied by the Employer shall receive a uniform or a clothing allowance in the amount of one hundred and fifty five dollars (\$155) which shall be paid on the first pay period in December or upon termination on a pro-rated basis.

35.02 **Provision of Protective Clothing**

Where protective clothing is necessary and is routinely required to be worn, the Employer shall provide such clothing free of charge. In cases where laundering of such clothing is required it shall be provided free of charge. Protective clothing shall not be deemed to be normal uniform.

35.03 Areas for Wearing of Protective Clothing

Protective clothing shall be provided in the following areas:

- (a) Pediatric Units
- (c) Case Rooms
- (b) Operating Rooms
- (d) Intensive Care Units

35.04 **Damaged Attire**

Any attire or article worn by an employee which is damaged by a patient in the course of their duties without their carelessness shall be repaired or replaced at the Employer's cost.

Article 36 - Temporary Assignment

36.01 Assignment to Higher Classification

Subject to Clause 36.02, when an employee is temporarily assigned to a position of higher classification they shall be paid an additional sixty-five (65) cents per hour per shift for each shift or part thereof when so assigned, provided they work four (4) hours.

36.02 Assignment Exceeding Ten (10) Days

(a) **Eight (8) Hour Shifts**

When a period of temporary assignment exceeds ten (10) consecutive working days exclusive of days of rest, the employee shall be paid the rate of pay for the higher classification in accordance with the promotion procedure for all time worked in the higher classification in excess of ten (10) working days.

(b) Assignment Exceeding Fourteen (14) Calendar Days Twelve (12) Hour Shifts

When a period of temporary assignment exceeds fourteen (14) calendar days, the employee shall be paid the rate of pay for the higher classification in accordance with the promotion procedure for all time worked in the higher classification in excess of fourteen (14) calendar days.

36.03 No Additional Payment

When an employee qualifies for the rate of pay established in accordance with Clause 36.02, they will no longer receive the additional payment set forth in Clause 36.01.

*36.04 Registered Nurse in Charge of Unit

*(a) Eight (8) and Twelve (12) Hour Shifts (including all weekend shifts)

When a Level I registered nurse, in any classification, is designated in charge of a unit, they shall receive an additional one dollar (\$1.00) per hour provided they work five (5) hours of the shift. A registered nurse shall be considered to be in charge if they are responsible for the patient care in their work area, as well as the assignment and direction of other employees assigned to that area, whether or not those employees are members of the RNUNL bargaining unit.

(b) In Charge 0800-1600 hours (Monday through Friday)

The in-charge premium shall be paid for each hour worked in charge between 0800-1600 hours (Monday through Friday) when the following criteria have been met and provided no other bargaining unit member has been assigned in-charge duties:

- (i) The Manager designates a Level 1 registered nurse to be in charge (either verbally or in writing) between 0800 and 1600 hours, or
- (ii) The Manager anticipates being absent from the work area/unit for two or more consecutive hours (exclusive of meal breaks and rest periods) between 0800 and 1600 hours, or
- (iii) The Manager anticipates being absent from the work area/unit for more than four (4) cumulative hours (exclusive of meal breaks and rest periods) between 0800 and 1600 hours.

In either of the criteria outlined above, the work areas/units will be determined based on current structure in place for the assignment of in charge between the hours of 1600 and 0800. However, in the event of restructuring or reorganization of services, Employers may change the structure of work areas/units. Provisions will be made

for discussion between the union and Employer where these changes impact on the number of registered nurses assigned in charge.

(c) A Level 1 registered nurse, in any classification, shall not be deemed to have had incharge responsibilities on the shift unless they have been designated in charge under this Clause.

36.05 Temporary Assignment to Lower Classification

When an employee is required to perform temporarily any work in a classification for which a lower rate than their own is paid, they shall continue to be paid their regular rate of pay.

*36.06 Team Leader

When a Level I registered nurse in any classification is a team leader on the eight (8) to four (4) shift Monday to Friday they shall receive an additional one dollar (\$1.00) per hour provided they work five (5) hours of the shift. A registered nurse shall be considered as a team leader if they are responsible for the patient care of the patients assigned to their team as well as the assignment and direction of the employees on their team.

Article 37 - Use of Automobile

37.01 Reimbursement for Use of Automobile

- (a) An employee who is not required as a condition of employment to have an automobile at their disposal, but who is required by the Employer to use their own car on the Employer's business shall be reimbursed at the rate of thirty-one point five (31.5) cents per kilometre effective April 1, 2000. If Government increases the rates paid to non-bargaining unit and management employees during the life of this agreement, those rates will also apply to employees covered by this agreement. (See Kilometre Rate Adjustment Formula on page 240).
- (b) When a nurse is required as a condition of employment to have an automobile at their disposal, effective date of signing of the collective agreement, they shall be paid for their travel on Employer business at the rate of forty-five point four (45.4) cents per kilometre for the first nine thousand kilometres per annum and thirty-one point five (31.5) cents per kilometre for kilometres in excess of nine thousand kilometres per annum. If Government increases the rates paid to non-bargaining unit and management employees during the life of this agreement, those rates will also apply to employees covered by this agreement. (See Kilometre Rate Adjustment Formula on page 240.)

*37.02 Nurses' Automobile Expenses

- (a) On receipt of invoice, employees will be reimbursed for the difference between private and business insurance.
- (b) Employees will be reimbursed for parking meter expenses incurred while on the business of the Employer, up to a maximum of ten (10) dollars per week effective April 1, 1999.

*(c) The Employer commits to reviewing job postings for nurses and nurse practitioners who are required to work at multiple sites to assess if a vehicle is required as a condition of employment.

*37.03 Meal Rates

Effective the date of signing, for each day or part thereof, on travel status, the maximum rate allowable for meals, inclusive of taxes and gratuities, shall be as follows:

	Breakfast	Lunch	Dinner	Total
NL	\$9.60	\$16.80	\$26.04	\$52.44
Other Provinces	\$12.18	\$19.68	\$28.38	\$60.24
USA (USD\$)	\$12.18	\$19.68	\$28.38	\$60.24
Other	\$13.50	\$21.54	\$31.20	\$66.24

In areas where the cost of meals is likely to exceed these rates, in the opinion of the Chief Executive Officer, vouchered expenses may be submitted and when the rates are increased by Treasury Board for meals, these rates will apply to nurses.

37.04 **Private Accommodations**

When an employee provides their own accommodations while travelling on the Employer's business, they will be compensated at the rate of twenty-five dollars (\$25) per night, effective April 1, 1999. In the event that during the life of this collective agreement, Treasury Board increases the amounts paid to employees who utilize private accommodations, these increases shall be paid to all eligible employees from the date specified by the Treasury Board.

The Employer agrees to maintain existing conditions that are more beneficial than the terms of this Clause for the life of this agreement.

Article 38 - Workers' Compensation

38.01 Scope of Workplace Health, Safety and Compensation Act

All employees shall be covered by the Workplace Health, Safety and Compensation Act.

38.02 **Pending Settlement**

Pending a settlement of an insurable claim, the employee shall be paid by the Employer the amount they would be entitled to under the *Workplace Health*, *Safety and Compensation Act*. The existing benefit accumulation practices shall be retained for the life of this Agreement, unless agreed otherwise.

38.03 **Retention of Benefits**

An employee who is unable to work due to injury received on duty shall not lose vacation or sick leave credits.

38.04 **Notice of Injury**

Subject to Section 53 of the *Workplace Health, Safety and Compensation Act*, 1990, no compensation is payable to an employee unless:

- (i) They give notice of the injury to their Employer as soon as practicable after the occurrence thereof and before they have voluntarily left the employment in which they were injured; and
- (ii) The claim for compensation is made within three (3) months from the occurrence of the injury or in case of death, within six months after the date of death.

38.05 **Rehabilitation**

- (a) A local representative of the employees and a representative of the Employer shall make every reasonable effort to have an injured employee returned to work with the Employer. These representatives and a representative of the Workplace Health, Safety and Compensation Commission, in consultation with the employee, shall review each long term claim (excess of 60 days) to discuss options, including any rehabilitation programs to assist the employee to return to work.
- (b) The Employer and the Union support the objectives of returning injured employees to employment through modified and ease back programs.
- (c) The representative of the employees and the representative of the Employer shall agree to develop a system of reviewing Workplace Health, Safety and Compensation claims and suggest methods of preventing injuries.
- (d) The representative of the employees and the Employer shall serve as coordinators in structuring ease back or modified work programs sponsored by the Workplace Health, Safety and Compensation Commission and shall act as support and contact persons for the injured employee while they are involved in ease back or modified work programs.

38.06 Extended Earnings Loss

It is understood and agreed by the parties to this collective agreement that an employee who is approved for full extended earnings loss (EEL) benefits from the Workplace Health, Safety and Compensation Commission, after the date of signing of this agreement, shall no longer accumulate benefits under this agreement but shall have their position with the Employer protected for two (2) calendar years following the date of such approval, immediately following which their employment shall be terminated, subject to the *Human Rights Act*.

Article 39 - Transfer and Portability

*39.01 **Benefits**

- (a) An employee transferring in good standing between hospitals covered in Schedule "C" or an employee who accepts employment in a hospital within one hundred and twenty (120) days of the resignation date from another hospital, shall retain the following benefits:
 - (i) accumulated sick leave credits
 - (ii) accumulated annual leave entitlement
 - (iii) health and insurance plan

- (iv) pension plan in accordance with the provisions of the pension plan for hospital employees
- (v) service for step progression
- (b) Employees who receive portability of benefits under Clause 39.01 (a) shall be placed on the appropriate salary scale at their new place of employment in accordance with the following:
 - (i) if the new position carries a pay range higher than that of the position just vacated, the employee shall be placed on the appropriate step of the new pay range in accordance with existing promotion procedures
 - (ii) if the new position carries a pay range equivalent to that of the position just vacated, the employee shall be placed on the same step of the equivalent pay range in accordance with existing transfer procedures
 - (iii) if the new position carries a pay range lower than that of the position just vacated, the employee shall be placed on the step of the new pay range in accordance with existing voluntary demotion procedures

39.02 **Re-employment After Termination**

Employees, including casual employees, re-employed by any Employer covered by this collective agreement after termination shall be placed on their respective salary scales on a step not lower than the step they were on at the date of termination provided that they have not been out of the employment of an Employer covered by the *Public Service Collective Bargaining Act* for a period of more than two (2) years.

39.03 Re-employment After Termination - Casual Employees

A permanent employee who is accepted for a casual position with another Employer covered by Schedule "C" within forty-five (45) calendar days and subsequently obtains permanent status without a break in service shall receive as a permanent employee with their new Employer the same step on the salary scale as they received from their previous Employer, provided they are employed in a classification at the same level.

Article 40 - Technological Change or New Method of Operation

40.01 Advance Notice

Before the introduction of any technological change or new method of operation which affects the right of employees, conditions of their employment, wage rates, or workloads, the Employer shall notify the Union of the proposed changes. Discussions on the proposed changes shall take place at least one (1) month following notification by the Employer.

40.02 No Layoff

No permanent employee will be laid off because of technological change or new method of operation unless such an employee refuses without good reason, to avail of additional training provided to equip the employee with the new or greater skills required by the technological change or new method of operation.

40.03 **Income Protection**

A permanent employee who is displaced from their job by virtue of technological change or new method of operation will suffer no reduction in their regular salary unless such an employee refused, without good reason, to avail of additional training provided to equip the employee with the new or greater skills required to prevent displacement.

Article 41 - Complete Agreement

41.01 Complete Agreement

This Agreement is the entire Agreement of the parties hereto terminating all prior agreements and practices with respect to those matters specifically provided for herein and concluding all collective bargaining during the term of this Agreement.

41.02 Amending or Altering Agreement

It is agreed by the parties to this Agreement that any provision in the Agreement, other than the term of the Agreement, may be amended or altered by mutual consent of the Employer and the Union.

*Article 42 - Leave for Attendance - CRNNL Annual Meeting

*42.01 Leave to Attend Meeting

Upon written request by the employee to the Employer, and with the approval of the Employer, leave with pay may be awarded as follows:

*(i) **Eight (8) Hour Shifts**

Paid leave of absence not exceeding three (3) days may be granted to an employee to attend the College of Registered Nurses of Newfoundland and Labrador annual meeting. The Employer will make every endeavour to grant such leave of absence.

*(ii) Twelve (12) Hour Shifts

Paid leave of absence not exceeding twenty-two point five (22.5) working hours shall be granted to an employee to attend the College of Registered Nurses of Newfoundland and Labrador annual meeting.

Article 43 – Schools of Nursing

43.01 Hours of Work

The regular weekly hours shall be thirty-seven and one half (37 1/2) hours a week or seventy-five (75) hours a fortnight divided into seven and one half (7 1/2) hour shifts excluding a one-half (1/2) hour meal period per shift but including a rest period of fifteen (15) consecutive minutes in the first half of a shift and in the second half of a shift. Where the instructor and Director of School of Nursing mutually agree, the instructor may take a one (1) hour meal break (exclusive of clinical days) and work the extra half (1/2) hour at the beginning or end of the shift.

The hours may be varied from day to day by mutual agreement with the instructor being able to report for duty not later than 1100 hours and leave work not earlier than 1500 hours provided the full shift is worked. No instructor shall be required to punch a time clock.

43.02 Work Preparation

The hours of work shall include the period which they met with other instructors to plan their student learning experience in the classroom, clinical area or community and the period during which they are responsible for students planned learning experiences in the classroom, clinical area or community. The Employer shall make provision for time to be available during the regular work period for instructors to do preparatory and other types of work required of them in connection with their teaching duties. Subject to the operational requirements of the School of Nursing, instructors shall be permitted to use learning resources outside the school during working hours to obtain up to date and pertinent literature for student instruction.

43.03 Attendance at Meetings of Faculty

An instructor's hours of work shall include attendance at meetings of the faculty or faculty committee. Meetings should be contained within the regular hours of work.

43.04 Leave of Absence

The instructor may take annual leave or time off in lieu of overtime worked when the students for whom they are responsible are also on annual leave. Consideration will be given to special requests for annual leave.

43.05 Text Books

Necessary required course text books will be supplied on loan to the instructor, and other necessary supplies shall be provided for lectures and clinical preparation to the instructor such as pens, pencils, paper, overhead transparencies and markers.

43.06 Attendance at Workshops

Faculty members attending workshops or related educational programs on their own time which are related to their faculty position, and when attendance is approved by the Employer, will have their tuition fees paid up to an amount approved by the Employer.

43.07 Condition of Employment

Where the Employer stipulates that as a condition of continued employment nurse instructors must be attending university courses toward completion of a baccalaureate degree the tuition fee for such courses shall be paid by the Employer.

43.08 Clothing

Notwithstanding Clause 35.01, lab coats shall be provided by the Employer to those nursing instructors who are required by the Employer to wear lab coats during working hours.

43.09 Instructional Assignments

The Director of the School of Nursing or their designate, shall meet at least once annually with the members of the Faculty of the School for the purpose of providing an opportunity to discuss the assignment of areas of responsibility or changes in the areas of responsibility.

43.10 **Graduation**

The instructor shall be allowed time off to attend the formal graduation ceremonies of the School of Nursing.

43.11 Merger and Amalgamation

The Employer shall meet and discuss proposals for the merger and amalgamation of any of the Schools covered by this Collective Agreement with the Union representatives at the request of the Union.

Article 44 - Educational Differential

44.01 Educational Allowance

(a) An allowance of eighty-two dollars (\$82) per month shall be paid to a nurse who is awarded a baccalaureate degree in nursing or who holds this degree and who has been on the top step of the Agreement for at least one (1) year of service.

An allowance of one hundred and ten dollars (\$110) per month shall be paid to a nurse who is awarded a Master's degree in Nursing or who holds this degree and who has been on the top step of the Agreement for at least one (1) year of service.

(b) Other Course Credits

- (i) CNA/CHA Nursing Unit Administration Course four hundred dollars (\$400) per year.
- (ii) Approved or recognized post graduate clinical preparation of three (3) months but less than six (6) months duration three hundred dollars (\$300) per year.
- (iii) Approved or recognized post-clinical preparation of six (6) months or more five hundred dollars (\$500) per year.
- (iv) "Approved or recognized" means approved or recognized by the Government of Newfoundland and Labrador.
- (v) These allowances will be paid for any period during which a nurse performs duties in the area to which their post graduate educational qualifications apply.

44.02 Psychiatric Classification BN

A nurse who has obtained a baccalaureate degree in Nursing and who is appointed to a psychiatric nursing position shall be considered a psychiatric nurse.

Article 45 - Contracting Out

45.01 **Job Protection**

Should the Employer contract out work, the Employer agrees to provide other positions for any employee that would normally be laid off by the decision to contract out work and the employee's regular salary at the time of contracting out shall be maintained during the duration of the contract.

45.02 Period of Notice

No contract services will be finalized without the union being given sixty (60) calendar days' notice of the Employer's intention to contract out the service.

Article 46 - Family Leave

*46.01 Family Leave

*(i) Eight (8) Hour Shifts

- (a) Subject to Clause 46.01 (b) and (c) an employee who is required to:
 - (i) attend to the temporary care of a sick family member living in the same household or the employee's mother, father or dependent child not necessarily living in the same household;
 - (ii) attend to the needs related to the birth of employee's child;
 - (iii) accompany a dependent family member living in the same household on a dental or a medical appointment;
 - (iv) attend meetings with school authorities;
 - (v) attend to needs related to the adoption of a child; and
 - (vi) attend to needs related to home or family emergencies;

shall be awarded up to three (3) days paid family leave in any calendar year.

- (b) In order to qualify for family leave, the employee shall:
 - (i) provide as much notice to the Employer as is reasonably possible;
 - (ii) provide to the Employer valid reasons why such leave is required;
 - (iii) where appropriate, and in particular with respect to (iii), (iv), and (v) of 45.01 (a) have endeavoured to a reasonable extent to schedule such events during off-duty hours.
- (c) Employees shall not be permitted to change any other leave to family leave but shall be entitled to change family leave to be reavement leave or sick leave.

*(ii) Twelve (12) Hour Shifts

- (a) Subject to Clause 46.01(b) and (c) an employee who is required to:
 - (i) attend to the temporary care of a sick family member living in the same household or the employee's mother, father or dependent child not necessarily living in the same household;
 - (ii) attend to the needs related to the birth of the employee's child;

- (iii) accompany a dependent family member living in the same household on a dental or medical appointment;
- (iv) attend meetings with school authorities;
- (v) attend to needs related to the adoption of a child; or
- (vi) attend to needs related to home or family emergencies;

shall be awarded up to twenty-two point five (22.5) hours paid family leave in any calendar year.

- (b) In order to qualify for family leave, the employee shall:
 - (i) provide as much notice to the Employer as is reasonably possible;
 - (ii) provide to the Employer valid reasons why such leave is required; and
 - (iii) where appropriate, and in particular with respect to (iii), (iv) and (v) of 46.01(a), have endeavoured to a reasonable extent to schedule such events during off duty hours.
- (c) Employees shall not be permitted to change any other leave to family leave but shall be entitled to change family leave to be reavement leave or sick leave.

Article 47 - Workplace Violence and Sexual and Personal Harassment

*47.01 Workplace Violence

Workplace violence is physical or non-physical violence which includes the threatened, attempted, or actual work-related incident of physical force or psychological abuse which can result in physical, emotional, and sexual injury, harm, or trauma. Workplace violence includes sexual harassment, harassment, bullying, and lateral violence which may also include cyber-bullying and domestic abuse.

The Employer and the Union recognize the right of employees to work in an environment free from workplace violence, and the Employer shall develop policies in support of this principle which shall be reviewed annually by the Occupational Health and Safety Committee. The parties shall undertake to expediently investigate alleged occurrences. If workplace violence against a bargaining unit member has taken place, the Employer shall take appropriate action to ensure the workplace violence ceases.

47.02 Sexual and Personal Harassment

- (a) Both the Employer and the Union consider sexual and personal harassment to be reprehensible and are committed to maintaining an environment in which sexual and personal harassment do not exist.
- (b) The Employer and the Union recognize the right of employees to work in an environment free from sexual and personal harassment and the parties shall undertake to investigate alleged occurrences with all possible dispatch. If sexual harassment or personal harassment of a bargaining unit member has taken place, the Employer shall take appropriate action to ensure that the sexual or personal harassment ceases.

The victim shall be protected from repercussions which may result from their complaint.

(c) Sexual harassment is comprised of sexual comments, gestures or physical contact that the individual knows, or ought reasonably to know, to be unwelcome, objectionable or offensive. The behaviour may be on a one (1) time basis or a series of incidents, however minor. It is unsolicited, one-sided and/or coercive. Both males and females may be the victim of sexual harassment.

Sexual harassment may involve favours, or promises of favours, or advantages in return for submission to sexual advances, or, alternatively, the threat of reprisal for refusal.

Sexual harassment can be expressed in a number of ways which may include:

- Unnecessary touching or patting;
- Suggestive remarks or other sexually aggressive remarks;
- Leering (suggestive staring) at a person's body;
- Demand for sexual favours;
- Compromising invitations;
- Physical assaults.
- (d) Personal harassment is any behavior by any person in the workplace that is directed at, or is offensive to, an employee, endangers an employee's job, undermines the performance of that job, or threatens the economic livelihood of the employee.

Personal harassment also occurs when an individual uses their authority or position, with its implicit power, to undermine, sabotage or otherwise interfere with the career of another employee.

Personal harassment may be defined as repeated, intentional, offensive comments or actions deliberately designed to demean an individual, or to cause personal humiliation.

The definition includes such blatant acts of misuse of power as intimidation, blackmail, and/or coercion.

*Article 48 - Family Violence Leave

- *48.01 An employee shall be granted a period of family violence leave of three (3) days paid leave and seven (7) days unpaid leave in a year where the employee or a person to whom the employee is a parent or caregiver has been directly or indirectly subjected to, a victim of, impacted or seriously affected by family violence or has witnessed family violence by:
 - (i) a person who is or has been a family member;
 - (ii) a person who is or has been in an intimate relationship or who is living or has lived with the employee;
 - (iii) a person who is the parent of a child with the employee;
 - (iv) a person who is or has been a caregiver to the employee; or
 - (v) any other person who is a member of a class of persons prescribed in the regulations pursuant to the *Labor Standards Act*.

Confidentiality

All personal information concerning family violence will be kept confidential in compliance with relevant Legislation.

An employee who wishes to take leave under this Clause may be required to provide the Employer with reasonable verification of the necessity of the leave.

Article 49 – Professional Practice Committee

49.01 **Professional Practices Committee**

(a) **Formation**

Within sixty (60) calendar days of receipt of a written request from the Union or the Employer, the Employer and the branch shall form a Professional Practices Committee (PPC) in all places of employment employing six (6) or more employees, unless such a committee already exists and is functioning.

(b) Composition of the Committee

This Committee shall be composed of three (3) nurses designated by the branch and three (3) persons designated by the Employer. An officer(s) or a representative(s) of the Union may attend meetings as a consultant. The Employer may also appoint a person(s) as a consultant. Each party shall endeavour to notify the other, seven (7) calendar days in advance of the attendance of a consultant at the meeting.

(c) Meetings of the Committee

The Committee shall meet at the request of either one of the parties but in any case at least once a month, unless mutually agreed otherwise. Employees who sit on the Committee shall be released from duty without loss of remuneration to attend meetings of the Committee or to perform any work required by the Committee. No employee serving on the PPC or in meetings with their manager shall lose any salary or benefits. Employees required to appear before the Committee may do so without loss of remuneration. Employees attending a PPC meeting outside normal working hours shall be compensated with straight time pay or time off in lieu. Employees

who are required to travel to attend these meetings will be compensated in accordance with the Employers travel policy.

(d) Chair of the Meeting

The meetings of the Committee shall be chaired alternately by the Employer's representative and the representative of the Branch.

(e) Minutes of Meeting

The Agenda for each meeting shall be circulated a minimum of seven (7) calendar days prior to each meeting. Minutes of each meeting shall be prepared, approved and signed by a representative of the Employer and the branch prior to circulation. The branch will be responsible for ensuring that a copy is forwarded to the Union.

(f) **Jurisdiction of the Committee**

The PPC shall meet in order to study any question that either of the parties may wish to discuss which is not properly the subject matter of a grievance. In workplaces where the parties have agreed to the establishment of joint committees to deal with professional practice/labour management issues, professional practice concerns shall be dealt with first where there are both professional practice and labour management issues on the agenda.

The committee shall review written concerns relative to patient/resident/client care including but not limited to:

- (i) Safety of patient/clients/residents and nurses;
- (ii) Quality practice environments;
- (iii) Professional Standards of Practice and Code of Ethics; and
- (iv) Workload where the nurse feels the patient load in the work area has exceeded what is believed to be safe patient care levels.

The PPC shall develop a procedure for investigating written concerns. The Committee may invite individuals to the meeting who will be able to assist in the investigation. The Committee shall make recommendations to the parties with respect to its discussion and conclusions. Records of written concerns shall be maintained separately in the PPC's files.

(g) **Reporting Form**

The Employer and Union shall agree upon a provincial reporting form for registering a written concern with the PPC.

(h) **Process to Address Concerns**

A nurse who has a concern related to Clause 49 (f) shall bring the matter to the attention of their immediate manager, or where appropriate, the manager's designate. If the matter is not satisfactorily resolved, the nurse may file a written Professional Practice Form (PPF). The PPF shall be submitted to the manager within seven (7) calendar days of the nurse identifying the concern.

The manager will provide a written response to the nurse within ten (10) calendar days of the form being received. A copy of the PPF containing the manager's response shall then be immediately forwarded to the PPC by the manager.

The PPC shall meet as soon as possible to resolve the concern. If resolved, the PPC shall provide a written response to the nurse within fourteen (14) calendar days of the meeting. If necessary, the PPC may engage the appropriate Director prior to providing a response.

Where the concern is not satisfactorily resolved at the PPC level, a report with a recommendation shall be forwarded by the PPC to the appropriate director and the Chief Nursing Officer. The Chief Nursing Officer/ designate shall provide a written response to the nurse and the PPC within thirty (30) calendar days.

(i) The PPC may invite the Chief Executive Officer and one member of the Board of Directors to attend one meeting of the Professional Practices Committee on an annual basis to discuss concerns unresolved in the process in Clause 49(h). At least one (1) months' notice shall be given and a mutually agreeable date and time shall be scheduled to accommodate the Chief Executive Officer and the Board representative.

Article 50 - Retroactivity

50.01 **Retroactivity**

Salaries are retroactive to July 1, 2022 as per Schedule A.

Article 51 - Terms of Agreement

*51.01 **Duration**

Except as otherwise provided in Clause 50.01, this agreement shall be effective from the date of signing and remain in full force and effect until June 30, 2026, and thereafter, from year to year unless either party gives notice in writing of its desire to change or amend this agreement not more than one hundred and twenty (120) calendar days and not less than thirty (30) calendar days prior to the date of expiration.

51.02 **Notice of Change**

Either party desiring to propose changes to this agreement shall within thirty (30) calendar days following receipt of notice under Clause 51.01, give notice in writing to the other party that changes are being proposed. Within a further thirty (30) days, the parties are required to enter into negotiations or agree upon a mutually acceptable time to commence negotiations for a new collective agreement.

51.03 Cost of Printing

The Employer shall print, at a cost to be equally shared between the Union and the Employer, sufficient copies of the Agreement as soon as possible after the Agreement is signed.

The parties have agreed to alphabetical listing for printed collective agreement.

IN WITNESS WHEREOF the parties hereto have executed this Agreement the day and year first before written.

<u>SIGNED</u> on behalf of the Treasury Board representing His Majesty the King in Right of Newfoundland and Labrador by the Honourable Siobhan Coady, President of Treasury Board and the Honourable Tom Osborne, Minister of Health and Community Services, in the presence of the witness hereto subscribing

WITNESS President of Treasury Board

Minister of Health and Community Services

SIGNED on behalf of the Registered Nurses' Union Newfoundland and Labrador by its proper officers in the presence of the witness hereto subscribing

President of Registered Auraes Union NL

*SCHEDULE A

STEP PROGRESSION

- (a) Employees shall advance one step on their respective salary scales on the date when twelve (12) months of service is accumulated.
- (b) The phrase "effective and date" shall mean as follows:
 - (i) if the anniversary date falls on any of the calendar days one (1) through seven (7) inclusive of the pay period, the effective date shall be the commencement of the pay period in which the anniversary date falls.
 - (ii) if the anniversary date falls on any of the calendar days eight (8) through fourteen (14) inclusive of the pay period, the effective date shall be the commencement of the pay period immediately following the pay period in which the anniversary date falls.
- (c) It is understood by the parties that, in the event that a nurse obtains employment in a higher paying classification within the Level I nursing classifications, including any new classifications which may be added, they shall be placed on the higher paying salary scale in accordance with their salary in their current position, and in accordance with the JES Implementation Formula as outlined in the January 17, 2014 letter on page 188 of the current Collective Agreement and as appended to this MOU in Appendix B.

CONTACT ALLOWANCE

Contact allowance shall be paid to nurses employed by the following hospitals: Waterford Hospital Harbour Lodge

The rate of the contact allowance shall be equivalent to that paid the support staff employees of these hospitals.

This allowance shall be pensionable.

This allowance shall not be paid to employees while they are attending courses at other hospitals. Employees on sick leave or special leave with pay in excess of twenty (20) consecutive days shall not be entitled to the allowance for the period of leave in excess of twenty (20) days.

This allowance shall be paid while employees are on sick leave, vacation and Workers' Compensation.

*SALARIES

Expiration date of contract to be June 30, 2026

Registered Nurses

```
July 1, 2022 – 2%
July 1, 2023 – 2%
July 1, 2024 – 2%
July 1, 2025 – 2%
```

*Effective date of signing, an employee will receive a one-time recognition bonus payment of \$2,000 pro-rated based on regular full-time hours for the hours worked during the previous twelve (12) months.

*Effective date of signing:

- The first two steps shall be dropped from the July 1, 2023 salary scale (NS-28 to NS-35) and an additional two steps shall be added. The new step calculation shall be consistent with the percentage differential between current steps (4 to 5 and 5 to 6). The salary scale shall be renumbered Step 1 to Step 6.
- An employee shall move to the same step on the new salary scale (NS-28 to NS-35).

Nurse Practitioners

```
July 1, 2022 – 2%
July 1, 2023 – 2%
July 1, 2024 – 2%
July 1, 2025 – 2%
```

*Effective date of signing, an employee will receive a one-time recognition bonus payment of \$2,000 pro-rated based on regular full-time hours for the hours worked during the previous twelve (12) months.

*Effective date of signing:

- A new salary scale will be created for Nurse Practitioners. NS-35 to NS-37 will be removed from the NS pay scale and an independent pay scale will be implemented.
- The first two steps shall be dropped from the July 1, 2023 salary scale (NS-35 to NS-37) and an additional two steps shall be added. Each new step calculation shall be consistent with the percentage differential between current steps (4 to 5 and 5 to 6). The salary scale shall be renumbered Step 1 to Step 6.
- Add 2% to each step on the salary scale.
- Nurse Practitioners shall move to the same step on the new salary scale (NP-35 NP-37).

Long Service Pay Premium

```
15 year – 1%
20 year – 15 year plus 2%
25 year – 20 year plus 2%
```

- Shall be effective date of signing and paid on a bi-weekly basis.
- Shall be applied to permanent employees, including permanent full-time and part-time.
- RNs and NPs are eligible for the 15 year, the 20 year and the 25 year wage increase after 15, 20 or 25 years as an RN and/or NP, respectively.
- Long Service Pay Premium is non-pensionable.

NURSES
EFFECTIVE July 1, 2022

PAY			, _, _,			
RANGE NUMBER	STEP 1	STEP 2	STEP 3	STEP 4	STEP 5	STEP 6
NS-28						
ANNUAL	\$66,909.96	\$69,535.44	\$72,379.71	\$75,840.57	\$79,281.54	\$82,841.85
BI-WEEKLY	\$2,573.46	\$2,674.44	\$2,783.84	\$2,916.95	\$3,049.29	\$3,186.23
HOURLY	\$34.31	\$35.66	\$37.12	\$38.89	\$40.66	\$42.48
NS-29						
ANNUAL	\$68,779.62	\$71,564.22	\$74,428.38	\$77,391.99	\$80,872.74	\$84,552.39
BI-WEEKLY	\$2,645.37	\$2,752.47	\$2,862.63	\$2,976.62	\$3,110.49	\$3,252.02
HOURLY	\$35.27	\$36.70	\$38.17	\$39.69	\$41.47	\$43.36
NS-30						
ANNUAL	\$71,086.86	\$74,070.36	\$77,153.31	\$80,832.96	\$84,492.72	\$88,271.82
BI-WEEKLY	\$2,734.11	\$2,848.86	\$2,967.44	\$3,108.96	\$3,249.72	\$3,395.07
HOURLY	\$36.45	\$37.98	\$39.57	\$41.45	\$43.33	\$45.27
NS-31						
ANNUAL	\$73,891.35	\$76,835.07	\$79,898.13	\$83,100.42	\$86,839.74	\$90,758.07
BI-WEEKLY	\$2,841.98	\$2,955.20	\$3,073.01	\$3,196.17	\$3,339.99	\$3,490.70
HOURLY	\$37.89	\$39.40	\$40.97	\$42.62	\$44.53	\$46.54
NS-32						
ANNUAL	\$76,914.63	\$80,176.59	\$83,577.78	\$87,157.98	\$91,076.31	\$95,173.65
BI-WEEKLY	\$2,958.26	\$3,083.72	\$3,214.53	\$3,352.23	\$3,502.94	\$3,660.53
HOURLY	\$39.44	\$41.12	\$42.86	\$44.70	\$46.71	\$48.81
NS-33						
ANNUAL	\$79,520.22	\$83,319.21	\$86,839.74	\$90,638.73	\$94,736.07	\$98,972.64
BI-WEEKLY	\$3,058.47	\$3,204.59	\$3,339.99	\$3,486.11	\$3,643.70	\$3,806.64
HOURLY	\$40.78	\$42.73	\$44.53	\$46.48	\$48.58	\$50.76
NS-34						
ANNUAL	\$83,080.53	\$86,660.73	\$90,360.27	\$94,298.49	\$98,535.06	\$102,990.42
BI-WEEKLY	\$3,195.41	\$3,333.11	\$3,475.40	\$3,626.87	\$3,789.81	\$3,961.17
HOURLY	\$42.61	\$44.44	\$46.34	\$48.36	\$50.53	\$52.82

ANNUAL	\$86,660.73	\$90,360.27	\$94,238.82	\$98,336.16	\$102,811.41	\$107 <i>,</i> 445.78
BI-WEEKLY	\$3,333.11	\$3,475.40	\$3,624.57	\$3,782.16	\$3,954.29	\$4,132.53
HOURLY	\$44.44	\$46.34	\$48.33	\$50.43	\$52.72	\$55.10

UNREGISTERED

NURSE

ANNUAL \$65,855.79
BI-WEEKLY \$2,351.99
HOURLY \$33.77

NURSES
EFFECTIVE July 1, 2023

PAY			• •			
RANGE NUMBER	STEP 1	STEP 2	STEP 3	STEP 4	STEP 5	STEP 6
NS-28						
ANNUAL	\$68,248.16	\$70,926.15	\$73,827.30	\$77,357.38	\$80,867.17	\$84,498.69
BI-WEEKLY	\$2,624.93	\$2,727.93	\$2,839.51	\$2,975.28	\$3,110.28	\$3,249.95
HOURLY	\$35.00	\$36.37	\$37.86	\$39.67	\$41.47	\$43.33
NS-29						
ANNUAL	\$70,155.21	\$72,995.50	\$75,916.95	\$78,939.83	\$82,490.19	\$86,243.44
BI-WEEKLY	\$2,698.28	\$2,807.52	\$2,919.88	\$3,036.15	\$3,172.70	\$3,317.06
HOURLY	\$35.98	\$37.43	\$38.93	\$40.48	\$42.30	\$44.23
NS-30						
ANNUAL	\$72,508.60	\$75,551.77	\$78,696.38	\$82,449.62	\$86,182.57	\$90,037.26
BI-WEEKLY	\$2,788.79	\$2,905.84	\$3,026.78	\$3,171.14	\$3,314.71	\$3,462.97
HOURLY	\$37.18	\$38.74	\$40.36	\$42.28	\$44.20	\$46.17
NS-31						
ANNUAL	\$75,369.18	\$78,371.77	\$81,496.09	\$84,762.43	\$88,576.53	\$92,573.23
BI-WEEKLY	\$2,898.81	\$3,014.30	\$3,134.47	\$3,260.09	\$3,406.79	\$3,560.51
HOURLY	\$38.65	\$40.19	\$41.79	\$43.47	\$45.42	\$47.47
NS-32						
ANNUAL	\$78,452.92	\$81,780.12	\$85,249.34	\$88,901.14	\$92,897.84	\$97,077.12
BI-WEEKLY	\$3,017.42	\$3,145.39	\$3,278.82	\$3,419.27	\$3,572.99	\$3,733.74
HOURLY	\$40.23	\$41.94	\$43.72	\$45.59	\$47.64	\$49.78
NS-33						
ANNUAL	\$81,110.62	\$84,985.59	\$88,576.53	\$92,451.50	\$96,630.79	\$100,952.09
BI-WEEKLY	\$3,119.64	\$3,268.68	\$3,406.79	\$3,555.83	\$3,716.57	\$3,882.77
HOURLY	\$41.60	\$43.58	\$45.42	\$47.41	\$49.55	\$51.77
NS-34						
ANNUAL	\$84,742.14	\$88,393.94	\$92,167.48	\$96,184.46	\$100,505.76	\$105,050.23
BI-WEEKLY	\$3,259.31	\$3,399.77	\$3,544.90	\$3,699.40	\$3,865.61	\$4,040.39
HOURLY	\$43.46	\$45.33	\$47.27	\$49.33	\$51.54	\$53.87

ANNUAL	\$88,393.94	\$92,167.48	\$96,123.60	\$100,302.88	\$104,867.64	\$109,594.70
BI-WEEKLY	\$3,399.77	\$3,544.90	\$3,697.06	\$3,857.80	\$4,033.37	\$4,215.18
HOURLY	\$45.33	\$47.27	\$49.29	\$51.44	\$53.78	\$56.20

UNREGISTERED

NURSE

ANNUAL \$67,172.91 BI-WEEKLY \$2,583.57 HOURLY \$34.45

NURSES EFFECTIVE August 4, 2023

PAY		220				
RANGE NUMBER	STEP 1	STEP 2	STEP 3	STEP 4	STEP 5	STEP 6
NS-28						
ANNUAL	\$73,827.30	\$77,357.38	\$80,867.17	\$84,498.69	\$88,166.09	\$91,955.22
BI-WEEKLY	\$2,839.51	\$2,975.28	\$3,110.28	\$3,249.95	\$3,391.00	\$3,536.74
HOURLY	\$37.86	\$39.67	\$41.47	\$43.33	\$45.21	\$47.16
NS-29						
ANNUAL	\$75,916.95	\$78,939.83	\$82,490.19	\$86,243.44	\$89,955.34	\$93,870.13
BI-WEEKLY	\$2,919.88	\$3,036.15	\$3,172.70	\$3,317.06	\$3,459.82	\$3,610.39
HOURLY	\$38.93	\$40.48	\$42.30	\$44.23	\$46.13	\$48.14
NS-30						
ANNUAL	\$78,696.38	\$82,449.62	\$86,182.57	\$90,037.26	\$93,937.17	\$97,958.82
BI-WEEKLY	\$3,026.78	\$3,171.14	\$3,314.71	\$3,462.97	\$3,612.97	\$3,767.65
HOURLY	\$40.36	\$42.28	\$44.20	\$46.17	\$48.17	\$50.24
NS-31						
ANNUAL	\$81,496.09	\$84,762.43	\$88,576.53	\$92,573.23	\$96,559.43	\$100,728.23
BI-WEEKLY	\$3,134.47	\$3,260.09	\$3,406.79	\$3,560.51	\$3,713.82	\$3,874.16
HOURLY	\$41.79	\$43.47	\$45.42	\$47.47	\$49.52	\$51.66
NS-32						
ANNUAL	\$85,249.34	\$88,901.14	\$92,897.84	\$97,077.12	\$101,253.62	\$105,612.70
BI-WEEKLY	\$3,278.82	\$3,419.27	\$3,572.99	\$3,733.74	\$3,894.37	\$4,062.03
HOURLY	\$43.72	\$45.59	\$47.64	\$49.78	\$51.92	\$54.16
NS-33						
ANNUAL	\$88,576.53	\$92,451.50	\$96,630.79	\$100,952.09	\$105,318.28	\$109,826.48
BI-WEEKLY	\$3,406.79	\$3,555.83	\$3,716.57	\$3,882.77	\$4,050.70	\$4,224.10
HOURLY	\$45.42	\$47.41	\$49.55	\$51.77	\$54.01	\$56.32
NS-34						
ANNUAL	\$92,167.48	\$96,184.46	\$100,505.76	\$105,050.23	\$109,566.92	\$114,306.78
BI-WEEKLY	\$3,544.90	\$3,699.40	\$3,865.61	\$4,040.39	\$4,214.11	\$4,396.41
HOURLY	\$47.27	\$49.33	\$51.54	\$53.87	\$56.19	\$58.62

ANNUAL	\$96,123.60	\$100,302.88	\$104,867.64	\$109,594.70	\$114,365.22	\$119,298.04
BI-WEEKLY	\$3,697.06	\$3,857.80	\$4,033.37	\$4,215.18	\$4,398.66	\$4,588.39
HOURLY	\$49.29	\$51.44	\$53.78	\$56.20	\$58.65	\$61.18

UNREGISTERED

NURSE

ANNUAL \$72,969.46 BI-WEEKLY \$2,806.52 HOURLY \$37.42

NURSES
EFFECTIVE July 1, 2024

PAY			• •			
RANGE	STEP 1	STEP 2	STEP 3	STEP 4	STEP 5	STEP 6
NUMBER					0.12.0	
NS-28						
ANNUAL	\$75,303.85	\$78,904.53	\$82,484.51	\$86,188.66	\$89,929.41	\$93,794.32
BI-WEEKLY	\$2,896.30	\$3,034.79	\$3,172.48	\$3,314.95	\$3,458.82	\$3,607.47
HOURLY	\$38.62	\$40.46	\$42.30	\$44.20	\$46.12	\$48.10
NS-29						
ANNUAL	\$77,435.29	\$80,518.63	\$84,139.99	\$87,968.31	\$91,754.45	\$95,747.53
BI-WEEKLY	\$2,978.28	\$3,096.87	\$3,236.15	\$3,383.40	\$3,529.02	\$3,682.60
HOURLY	\$39.71	\$41.29	\$43.15	\$45.11	\$47.05	\$49.10
NS-30						
ANNUAL	\$80,270.31	\$84,098.61	\$87,906.22	\$91,838.01	\$95,815.91	\$99,918.00
BI-WEEKLY	\$3,087.32	\$3,234.56	\$3,381.01	\$3,532.23	\$3,685.23	\$3,843.00
HOURLY	\$41.16	\$43.13	\$45.08	\$47.10	\$49.14	\$51.24
NS-31	400.406.04	406 457 60	400 040 06	404 404 60	400 400 60	6400 740 70
ANNUAL	\$83,126.01	\$86,457.68	\$90,348.06	\$94,424.69	\$98,490.62	\$102,742.79
BI-WEEKLY	\$3,197.15	\$3,325.30	\$3,474.93	\$3,631.72	\$3,788.10	\$3,951.65
HOURLY	\$42.63	\$44.34	\$46.33	\$48.42	\$50.51	\$52.69
NS-32						
ANNUAL	\$86,954.33	\$90,679.16	\$94,755.80	\$99,018.66	\$103,278.69	\$107,724.95
BI-WEEKLY	\$3,344.40	\$3,487.66	\$3,644.45	\$3,808.41	\$3,972.26	\$4,143.27
HOURLY	\$44.59	\$46.50	\$48.59	\$50.78	\$52.96	\$55.24
	*	7	7	7000	7	****
NS-33						
ANNUAL	\$90,348.06	\$94,300.53	\$98,563.41	\$102,971.13	\$107,424.65	\$112,023.01
BI-WEEKLY	\$3,474.93	\$3,626.94	\$3,790.90	\$3,960.43	\$4,131.72	\$4,308.58
HOURLY	\$46.33	\$48.36	\$50.55	\$52.81	\$55.09	\$57.45
NS-34						
ANNUAL	\$94,010.83	\$98,108.15	\$102,515.88	\$107,151.23	\$111,758.26	\$116,592.92
BI-WEEKLY	\$3,615.80	\$3,773.39	\$3,942.92	\$4,121.20	\$4,298.39	\$4,484.34
HOURLY	\$48.21	\$50.31	\$52.57	\$54.95	\$57.31	\$59.79

ANNUAL	\$98,046.07	\$102,308.94	\$106,964.99	\$111,786.59	\$116,652.52	\$121,684.00
BI-WEEKLY	\$3,771.00	\$3,934.96	\$4,114.04	\$4,299.48	\$4,486.64	\$4,680.15
HOURLY	\$50.28	\$52.47	\$54.85	\$57.33	\$59.82	\$62.40

UNREGISTERED

NURSE

ANNUAL \$74,428.85 BI-WEEKLY \$2,862.65 HOURLY \$38.17

NURSES
EFFECTIVE July 1, 2025

PAY		220.	, _, _,			
RANGE NUMBER	STEP 1	STEP 2	STEP 3	STEP 4	STEP 5	STEP 6
NS-28						
ANNUAL	\$76,809.93	\$80,482.62	\$84,134.20	\$87,912.43	\$91,728.00	\$95,670.21
BI-WEEKLY	\$2,954.23	\$3,095.49	\$3,235.93	\$3,381.25	\$3,528.00	\$3,679.62
HOURLY	\$39.39	\$41.27	\$43.15	\$45.08	\$47.04	\$49.06
NS-29						
ANNUAL	\$78,984.00	\$82,129.00	\$85,822.79	\$89,727.68	\$93,589.54	\$97,662.48
BI-WEEKLY	\$3,037.85	\$3,158.81	\$3,300.88	\$3,451.06	\$3,599.60	\$3,756.25
HOURLY	\$40.50	\$42.12	\$44.01	\$46.01	\$47.99	\$50.08
NS-30						
ANNUAL	\$81,875.72	\$85,780.58	\$89,664.34	\$93,674.77	\$97,732.23	\$101,916.36
BI-WEEKLY	\$3,149.07	\$3,299.25	\$3,448.63	\$3,602.88	\$3,758.93	\$3,919.86
HOURLY	\$41.99	\$43.99	\$45.98	\$48.04	\$50.12	\$52.26
NS-31						
ANNUAL	\$84,788.53	\$88,186.83	\$92,155.02	\$96,313.18	\$100,460.43	\$104,797.65
BI-WEEKLY	\$3,261.10	\$3,391.80	\$3,544.42	\$3,704.35	\$3,863.86	\$4,030.68
HOURLY	\$43.48	\$45.22	\$47.26	\$49.39	\$51.52	\$53.74
NS-32						
ANNUAL	\$88,693.42	\$92,492.74	\$96,650.92	\$100,999.03	\$105,344.26	\$109,879.45
BI-WEEKLY	\$3,411.29	\$3,557.41	\$3,717.34	\$3,884.58	\$4,051.70	\$4,226.13
HOURLY	\$45.48	\$47.43	\$49.56	\$51.79	\$54.02	\$56.35
NS-33						
ANNUAL	\$92,155.02	\$96,186.54	\$100,534.68	\$105,030.55	\$109,573.14	\$114,263.47
BI-WEEKLY	\$3,544.42	\$3,699.48	\$3,866.72	\$4,039.64	\$4,214.35	\$4,394.75
HOURLY	\$47.26	\$49.33	\$51.56	\$53.86	\$56.19	\$58.60
NS-34						
ANNUAL	\$95,891.05	\$100,070.31	\$104,566.20	\$109,294.25	\$113,993.43	\$118,924.78
BI-WEEKLY	\$3,688.12	\$3,848.86	\$4,021.78	\$4,203.63	\$4,384.36	\$4,574.03
HOURLY	\$49.17	\$51.32	\$53.62	\$56.05	\$58.46	\$60.99

ANNUAL	\$100,006.99	\$104,355.12	\$109,104.29	\$114,022.32	\$118,985.57	\$124,117.68
BI-WEEKLY	\$3,846.42	\$4,013.66	\$4,196.32	\$4,385.47	\$4,576.37	\$4,773.76
HOURLY	\$51.29	\$53.52	\$55.95	\$58.47	\$61.02	\$63.65

UNREGISTERED

NURSE

ANNUAL \$75,917.43 BI-WEEKLY \$2,919.90 HOURLY \$38.93

NURSE PRACTITIONERS EFFECTIVE July 1, 2022

PAY RANGE NUMBER	STEP 1	STEP 2	STEP 3	STEP 4	STEP 5	STEP 6
NP-35						
ANNUAL	\$86,660.73	\$90,360.27	\$94,238.82	\$98,336.16	\$102,811.41	\$107,445.78
BI-WEEKLY	\$3,333.11	\$3,475.40	\$3,624.57	\$3,782.16	\$3,954.29	\$4,132.53
HOURLY	\$44.44	\$46.34	\$48.33	\$50.43	\$52.72	\$55.10
NP-36						
ANNUAL	\$90,141.48	\$93,980.25	\$98,057.70	\$102,254.49	\$106,868.97	\$111,682.35
BI-WEEKLY	\$3,466.98	\$3,614.63	\$3,771.45	\$3,932.87	\$4,110.35	\$4,295.48
HOURLY	\$46.23	\$48.20	\$50.29	\$52.44	\$54.80	\$57.27
NP-37						
ANNUAL	\$94,099.59	\$98,276.49	\$102,592.62	\$107,008.20	\$111,881.25	\$116,933.31
BI-WEEKLY	\$3,619.22	\$3,779.87	\$3 <i>,</i> 945.87	\$4,115.70	\$4,303.13	\$4,497.44
HOURLY	\$48.26	\$50.40	\$52.61	\$54.88	\$57.38	\$59.97

NURSE PRACTITIONERS EFFECTIVE July 1, 2023

PAY RANGE NUMBER	STEP 1	STEP 2	STEP 3	STEP 4	STEP 5	STEP 6
NP-35						
ANNUAL	\$88,393.94	\$92,167.48	\$96,123.60	\$100,302.88	\$104,867.64	\$109,594.70
BI-WEEKLY	\$3,399.77	\$3,544.90	\$3,697.06	\$3,857.80	\$4,033.37	\$4,215.18
HOURLY	\$45.33	\$47.27	\$49.29	\$51.44	\$53.78	\$56.20
NP-36						
ANNUAL	\$91,944.31	\$95,859.86	\$100,018.85	\$104,299.58	\$109,006.35	\$113,916.00
BI-WEEKLY	\$3,536.32	\$3,686.92	\$3,846.88	\$4,011.52	\$4,192.55	\$4,381.38
HOURLY	\$47.15	\$49.16	\$51.29	\$53.49	\$55.90	\$58.42
NP-37						
ANNUAL	\$95,981.58	\$100,242.02	\$104,644.47	\$109,148.36	\$114,118.88	\$119,271.98
BI-WEEKLY	\$3,691.60	\$3,855.46	\$4,024.79	\$4,198.01	\$4,389.19	\$4,587.38
HOURLY	\$49.22	\$51.41	\$53.66	\$55.97	\$58.52	\$61.17

NURSE PRACTITIONERS EFFECTIVE August 4, 2023

PAY RANGE NUMBER	STEP 1	STEP 2	STEP 3	STEP 4	STEP 5	STEP 6
NP-35						
ANNUAL	\$98,046.07	\$102,308.94	\$106,964.99	\$111,786.59	\$116,652.52	\$121,684.00
BI-WEEKLY	\$3,771.00	\$3,934.96	\$4,114.04	\$4,299.48	\$4,486.64	\$4,680.15
HOURLY	\$50.28	\$52.47	\$54.85	\$57.33	\$59.82	\$62.40
NP-36						
ANNUAL	\$102,019.23	\$106,385.57	\$111,186.48	\$116,194.32	\$121,211.46	\$126,435.53
BI-WEEKLY	\$3,923.82	\$4,091.75	\$4,276.40	\$4,469.01	\$4,661.98	\$4,862.91
HOURLY	\$52.32	\$54.56	\$57.02	\$59.59	\$62.16	\$64.84
NP-37						
ANNUAL	\$106,737.36	\$111,331.33	\$116,401.26	\$121,657.42	\$126,956.29	\$132,441.39
BI-WEEKLY	\$4,105.28	\$4,281.97	\$4,476.97	\$4,679.13	\$4,882.93	\$5,093.90
HOURLY	\$54.74	\$57.09	\$59.69	\$62.39	\$65.11	\$67.92

NURSE PRACTITIONERS EFFECTIVE July 1, 2024

PAY RANGE NUMBER	STEP 1	STEP 2	STEP 3	STEP 4	STEP 5	STEP 6
NP-35						
ANNUAL	\$100,006.99	\$104,355.12	\$109,104.29	\$114,022.32	\$118,985.57	\$124,117.68
BI-WEEKLY	\$3,846.42	\$4,013.66	\$4,196.32	\$4,385.47	\$4,576.37	\$4,773.76
HOURLY	\$51.29	\$53.52	\$55.95	\$58.47	\$61.02	\$63.65
NP-36						
ANNUAL	\$104,059.61	\$108,513.28	\$113,410.21	\$118,518.21	\$123,635.69	\$128,964.24
BI-WEEKLY	\$4,002.29	\$4,173.59	\$4,361.93	\$4,558.39	\$4,755.22	\$4,960.16
HOURLY	\$53.36	\$55.65	\$58.16	\$60.78	\$63.40	\$66.14
NP-37						
ANNUAL	\$108,872.11	\$113,557.96	\$118,729.29	\$124,090.57	\$129,495.42	\$135,090.22
BI-WEEKLY	\$4,187.39	\$4,367.61	\$4,566.51	\$4,772.71	\$4,980.59	\$5,195.78
HOURLY	\$55.83	\$58.23	\$60.89	\$63.64	\$66.41	\$69.28

NURSE PRACTITIONERS EFFECTIVE July 1, 2025

PAY RANGE NUMBER	STEP 1	STEP 2	STEP 3	STEP 4	STEP 5	STEP 6
NP-35						
ANNUAL	\$102,007.13	\$106,442.22	\$111,286.38	\$116,302.77	\$121,365.28	\$126,600.03
BI-WEEKLY	\$3,923.35	\$4,093.93	\$4,280.25	\$4,473.18	\$4,667.90	\$4,869.23
HOURLY	\$52.31	\$54.59	\$57.07	\$59.64	\$62.24	\$64.92
NP-36						
ANNUAL	\$106,140.80	\$110,683.55	\$115,678.41	\$120,888.57	\$126,108.40	\$131,543.52
BI-WEEKLY	\$4,082.34	\$4,257.06	\$4,449.17	\$4,649.56	\$4,850.32	\$5,059.37
HOURLY	\$54.43	\$56.76	\$59.32	\$61.99	\$64.67	\$67.46
NP-37						
ANNUAL	\$111,049.55	\$115,829.12	\$121,103.88	\$126,572.38	\$132,085.33	\$137,792.02
BI-WEEKLY	\$4,271.14	\$4,454.97	\$4,657.84	\$4,868.17	\$5,080.21	\$5,299.69
HOURLY	\$56.95	\$59.40	\$62.10	\$64.91	\$67.74	\$70.66

SCHEDULE B

EXPERIENCE CREDITS

Effective the date of signing of this collective agreement:

- (a) Less than two (2) years experience, Step 1 of the appropriate scale.
- (b) Two (2) years experience, but less than three (3), Step 2 of the appropriate salary scale.
- (c) Three (3) years experience, but less than four (4), Step 3 of the appropriate salary scale.
- (d) Four (4) years experience, but less than five (5), Step 4 of the appropriate scale.
- (e) Five (5) years experience, but less than six (6), Step 5 of the appropriate scale.
- (f) Six (6) years experience or more, Step 6 of the appropriate scale.
- (g) The provisions of paragraphs (a), (b), (c), (d), (e) and (f) will not apply to a nurse who is required to do a refresher course by the College of Registered Nurses of Newfoundland and Labrador until the nurse has completed twelve (12) months of continuous service with the Employer.
- (h) A nurse who moves into the Province and is employed at the unregistered nurses rate and who is completely eligible for registration pending confirmation of their full registration, shall upon receipt of full registration, be placed on the appropriate registered nurses scale retroactive to their date of employment.
- (i) Any registered nurse who has not attained the salary level outlined in paragraph (a), (b), (c), (d), (e), or (f), but who during the term of this agreement qualifies in accordance with those experience requirements shall receive the benefit of the step movement effective the date they qualify.

*SCHEDULE C

*(a) EMPLOYERS COVERED BY THIS AGREEMENT

Provincial Health Authority (known as "Newfoundland and Labrador Health Services")

The Provincial Health Authority was created pursuant to the *Provincial Health Authority Act* RSNL 2022 Chapter P-30.1, which came into force on April 1, 2023. The Provincial Health Authority, which is also known as Newfoundland and Labrador Health Services, is considered to have assumed and is to be vested with the assets, liabilities, rights and obligations, whether contractual, statutory or otherwise, of all services, groups or boards replaced by the Authority.

(b) Government of Newfoundland and Labrador Operated Hospitals and Institutions (represented herein by Treasury Board).

NOTES:

- (a) If a hospital presently operated by Government of Newfoundland and Labrador (Treasury Board) changes its management or operators, it shall be bound by this Agreement the same as if it were specifically listed in Schedule "C".
- (b) Treasury Board is also deemed to be the Employer of nurses in bargaining unit classes who do not work in hospitals or institutions per se but are employed directly by Government in the Division of Nursing Services in the Department of Health and Community Services.

*SCHEDULE D

CLASSIFICATIONS OF NURSES COVERED BY THIS AGREEMENT

	Classification	Pay Range
Class Title	Specification No.	No.
Community Health Registered Nurse I	H080	NS-30
Community Health Registered Nurse II	H081	NS-31
Community Health Registered Nurse IIA	H315	NS-32
Disease Control Registered Nursing Specialist	H104	NS-34
Infection Control Co-Ordinator	H124	NS-34
Infection Control Registered Nurse	H161	NS-31
Nurse Educator	H162	NS-33
Nurse Practitioner (Acute Care)	H172	NP-35
Nurse Practitioner (Adult)	H312	NP-35
Nurse Practitioner (Family/All Ages)	H295	NP-35
Nurse Practitioner (Primary Health Care)	H173	NP-35
Nursing Education Co-Ordinator I	H174	NS-28
Nursing Education Co-Ordinator II	H176	NS-30
Nursing Instructor I	H177	NS-30
Nursing Instructor II	H179	NS-32
Occupational Health Coordinator	5521	NS-31
Occupational Health Registered Nurse	H316	NS-31
Occupational Health Registered Nurse Coordinator	H336	NS-34
Psychiatric Registered Nurse I	H221	NS-28
Psychiatric Registered Nurse IA	H335	NS-29
Psychiatric Registered Nurse II	H223	NS-30
Psychiatric Registered Nurse IIA	H224	NS-31
Psychiatric Registered Nurse IIB	H225	NS-32
Psychiatric Registered Nurse IID	H227	NS-34
Public Health Registered Nurse I	H230	NS-30
Public Health Registered Nurse II	H231	NS-31
Regional Registered Nurse I	H251	NS-33
Regional Registered Nurse II	H252	NS-34
Registered Nurse I	H163	NS-28
Registered Nurse IA	H164	NS-29
Registered Nurse IB	H165	NS-30
Registered Nurse IC	H166	NS-31
Registered Nurse II	H167	NS-29
Registered Nurse IIA	H304	NS-30
Registered Nurse IIB	H169	NS-31
Registered Nurse IIC	H170	NS-32
Registered Nurse IID	H171	NS-33
Registered Nurse IIE	H314	NS-34
Research Registered Nurse Co-Ordinator	H255	NS-31
Transfusion Registered Nurse Specialist	H310	NS-34

SCHEDULE E

VACATION LEAVE

Leave entitlement based on the following number of days per annum:

	<u>15</u>	<u>20</u>	<u>25</u>	<u>30</u>
1 month	1	2	2	2.5
2 months	2	3	4	5
3 months	4	5	6	7.5
4 months	5	7	8	10
5 months	6	8	10	12.5
6 months	7	10	12	15
7 months	9	12	15	17.5
8 months	10	13	17	20
9 months	11	15	19	22.5
10 months	12	17	21	25
11 months	14	18	23	27.5
12 months	15	20	25	30

SCHEDULE F

THE CLASSIFICATION REVIEW AND APPEAL PROCESS

A. **Definitions**

- 1. "Appeal" means a request by an employee to the Classification Appeal Adjudicator to review specific factor allocations determined by the Classification & Organizational Design Division that they consider being incorrect.
- 2. "Adjudicator" refers to an individual who is appointed to the position of Classification Appeal Adjudicator of the Public Service Commission.
- 3. "Classification" means the identification of a position by reference to a classification title and pay range number.
- 4. "Classification Appeal Adjudicator" means the individual appointed to function in accordance with these procedures.
- 5. "Day" means a working day.
- 6. "Factor" means a compensable job element that applies to all jobs.
- 7. "Factor Rating" means the numerical value assigned to a factor.
- 8. "Permanent Head" means permanent head as defined below, or any official authorized by them to act on their behalf:
 - in respect of persons employed by government departments, the Deputy Minister of the department concerned;
 - in respect of employees of agencies not specifically covered by the definitions in this section, the highest management official in these agencies;
 - in respect of employees of Board operated hospitals and homes the CEO and/or Executive Director.
- 9. "Review" means a re-appraisal or re-assessment of an employee's position classification by the Classification & Organizational Design Division of the Treasury Board Secretariat upon request of an employee or the permanent head on behalf of an organization.
- 10. "Treasury Board" means Treasury Board as constituted pursuant to *The Financial Administration Act* as now or hereafter amended.
- 11. "Organization" means the Government of Newfoundland and Labrador, commission, agency, hospital or other entity mentioned in Section A.8.

B. Classification Review Process

1. The process of a classification review shall be available to an organization if the organization considers that a position has been improperly classified by the Classification & Organizational Design Division of the Treasury Board Secretariat.

The process of review and/or appeal pursuant to these procedures shall be available to any employee who considers that their position has been improperly classified by the Classification & Organizational Design Division of the Treasury Board Secretariat.

- 2. A review shall not be entertained on the grounds:
 - of inadequacy of the pay scale assigned to the pay range number; or
 - that the scope of duties and responsibilities has been improperly assigned to the position by management.
- 3. Organizations or employees who wish to have a position reviewed are able to do so by submitting a Position Description Questionnaire (PDQ) to the Classification & Organizational Design Division, Treasury Board Secretariat, Confederation Building, St. John's, A1B 4J6. Employees may use the Microsoft Word version of the PDQ or complete and submit the PDQ online.

Information on access to the necessary documents can be found in the Treasury Board Secretariat's website http://www.exec.gov.nl.ca/exec/hrs/newjobevaluation.html

- 4. A request for review shall be regarded as closed:
 - when a decision is rendered thereon by the Classification & Organizational Design Division;
 - if the employee(s) requests in writing the withdrawal of the request for review;
 - in the event an employee is dismissed with cause. If the employee separates from the organization for a reason other than dismissal for cause, the employee may request the difference in pay as a result of an outstanding classification review but would not be entitled to a further review or appeal; or
 - if the permanent head, in the case of an organization request for review, requests in writing the withdrawal of the request for review.
- 5. All documents relating to a classification review shall be maintained by the Classification & Organizational Design Division. Copies of such classification review materials shall be provided to the Classification Appeal Adjudicator upon its request.

C. <u>Classification Appeal Processs</u>

- 1. If an employee is dissatisfied with the decision of the Classification & Organizational Design Division, an appeal of the decision may be submitted to the Classification Appeal Adjudicator of the Public Service Commission. The request for an appeal must identify which factor(s) is/are being challenged and the associated rationale for each factor(s). The appeal process is restricted to those factors identified as being challenged and sufficient reasoning provided.
- 2. All such appeals shall be submitted to the Classification Appeal Adjudicator in writing within a period of not more than fourteen (14) days after the receipt by an employee of notification of the Classification & Organizational Design Division's decision as above mentioned.

- 3. A classification appeal of specific factor(s) shall not be accepted by the Classification Appeal Adjudicator based on job content which differs from that reviewed by the Classification and Organizational Division. In such a case, the employee or group of employees shall first approach the Classification and Organizational Division seeking a further review on the basis of the new circumstances involved.
- 4. The Classification Appeal Adjudicator shall be an independent position created within the Public Service Commission.
- 5. The Classification Appeal Adjudicator is hereby empowered to receive, hear and decide upon any appeal consistent with these procedures.
- 6. Changes in these procedures shall be recommended for approval only after co-ordination with the Public Service Commission, and the Treasury Board Secretariat.
- 7. The Classification Appeal Adjudicator may render decisions based on the information provided or may hold hearings if deemed necessary. The appellant may be required to appear at any time and in any place in the province deemed desirable.
- 8. The Classification Appeal Adjudicator shall only consider and rule <u>upon the factors challenged</u> by an individual employee, or group of employees having identical classifications, provided that such employee or group shall first have submitted their request in accordance with Section 3 of Part B and shall have been notified in writing of the Classification & Organizational Design Division's decision on the request.
- 9. The Classification Appeal Adjudicator has the right to refuse to receive or hear an appeal if it considers that the grounds on which the appeal was submitted are irrelevant or not in accordance with Sections 1 and 2 of Part B.
- 10. The employing organization concerned shall allow time off from regular duties to any employee who is required by the Classification Appeal Adjudicator to appear before him/her and, in respect of such absence; the employee shall be regarded as being O.H.M.S. It is the responsibility of the employee to obtain the prior approval of the permanent head before absenting themselves from their duties for this purpose.
- 11. On receipt of an appeal from an employee or a group of employees, the Classification Appeal Adjudicator may request the Classification & Organizational Design Division to assemble all pertinent information prepared as a result of the classification review, a copy of which will be provided to the appellant and the immediate supervisor by the Classification Appeal Adjudicator.
- 12. Where the appellant requires clarification on any point contained in the classification file or wishes to comment on any aspect of the classification file, they must file with the Classification Appeal Adjudicator within fourteen (14) days of receiving the file, a written statement including any supporting documentation which details their questions or comments.

- 13. A copy of the appellant's written statement and supporting documentation will be sent by the Classification Appeal Adjudicator, within three (3) days of receipt, to the Classification & Organizational Design Division who may respond or be requested to respond in writing within fourteen (14) days to the questions or observations raised by the appellant. Such response shall be forwarded by the Classification Appeal Adjudicator to the appellant within three (3) days of receipt. This cumulative documentation shall then constitute the entire appeal file to be considered by the Classification Appeal Adjudicator.
- 14. Where the Classification Appeal Adjudicator is satisfied that all relevant documentation is on file, it shall determine whether a hearing is warranted or if a decision can be rendered on the basis of the written documentation provided.
- 15. Where in the opinion of the Classification Appeal Adjudicator a group of appellants' position description questionnaires are sufficiently similar, have identical ratings and the appellants are employed in the same classification by the same organization, the Adjudicator may propose the consolidation of individual appeals to those appellants such that the appeals of individuals may be decided upon in a group appeal.
- 16. Where the Classification Appeal Adjudicator proposes a group review, the individual appellants must indicate their agreement with the group review in writing.
- 17. Where not all appellants agree to consolidation, the Classification Appeal Adjudicator will first decide on a consolidated basis the appeals of those appellants who are in agreement with consolidation. Those appellants not in agreement shall be provided an opportunity for individual review, as soon as practical following the determination of a consolidated appeal.
- 18. When the Classification Appeal Adjudicator renders a decision on those factors challenged on the basis of the written documentation, notification of such decision on those factors challenged shall be forwarded to the Classification and Organizational Design Division. The Classification and Organizational Design Division will notify and provide the necessary authority to the employing agency as well as provide a copy of the Classification Appeal Adjudicators decision and the impact, if any, on the position to the appellant and their designate.
- 19. If a hearing is warranted, the appellant, a permanent head or management designate and a representative of Classification and Organizational Design Division may be requested to appear before the Classification Appeal Adjudicator.
- 20. Appellants are to be given two opportunities to postpone appeal hearings after which appeals will then be withdrawn by the Classification Appeal Adjudicator.
- 21. The hearing will be presided over by the Classification Appeal Adjudicator who will retain control over the conduct of the hearing and who will rule on the relevancy of any questions raised by any of the parties.

- 22. The Classification Appeal Adjudicator may adjourn the hearing and order the appearance of any person or party who, at the Classification Appeal Adjudicator's discretion, it deems necessary to appear to give information or to clarify any issue raised during the hearing.
- 23. Following the conclusion of the hearing, the Classification Appeal Adjudicator will deliberate on and consider all relevant evidence and supporting information. Within fifteen (15) working days of reaching a decision, the Classification Appeal Adjudicator shall inform the Classification & Organizational Design Division in writing over the signature of the Classification Appeal Adjudicator of their decision on those factors challenged and a detailed explanation of the rationale of any change from the Classification and Organizational Design Division's original determination. The Classification and Organizational Design Division will notify and provide the necessary authority to the employing agency as well as provide a copy of the Classification Appeal Adjudicator's decision and the impact, if any, on the position to the appellant and their designate.
- 24. The Classification Appeal Adjudicator is required to submit written reasons to the Classification and Organizational Design Division for those decisions that result in changes in the factors challenged.
- 25. The impacts of changes in ratings arising from decisions of the Classification Appeal Adjudicator shall be processed by the Classification and Organizational Design Division in accordance with the Treasury Board Secretariat's compensation policies.
- 26. The decision of the Classification Appeal Adjudicator on an appeal is final and binding on the parties to the appeal.
- 27. An appeal shall be regarded as closed:
 - when a decision is rendered thereon by the Classification Appeal Adjudicator;
 - if the appellant requests in writing the withdrawal of the appeal;
 - in the event an employee is dismissed with cause; or
 - if the appellant postpones a hearing in accordance with Section 20 of Part C.

*SCHEDULE G

SUMMARY OF GROUP INSURANCE BENEFITS FOR MEMBERS OF THE GOVERNMENT OF NEWFOUNDLAND AND LABRADOR PLAN

The online "GNL Summary of Benefits" booklet contains a more detailed description of the benefits under the Plan. The following summary has been prepared to outline the basic content of the Plan only, as contractual provisions specified within the group insurance policies prevail. You may refer to the Government website at http://www.exec.gov.nl.ca/exec/hrs/working with us/employee benefits.html for further information.

BENEFITS

BASIC GROUP LIFE INSURANCE

You are insured for a life insurance benefit equal to two times your current annual salary rounded to the next higher \$1,000, if not already a multiple thereof, subject to a minimum of \$10,000 and a maximum of \$1,000,000.

If your insurance ceases on or prior to age 65, you may be entitled to convert the cancelled amount of basic group life insurance to an individual policy of the type then being offered by the insurer to conversion applicants within 31 days of the termination and no medical evidence of insurability would be required. The premium rate would be based on your age and class of risk at that time.

DEPENDENT LIFE INSURANCE

In the event of the death of your spouse or dependent child from any cause whatsoever, while you and your dependents are insured under the plan, the insurance company will pay you \$10,000 in respect of your spouse and \$5,000 in respect of each insured dependent child. This applies to those employees with family health coverage only.

BASIC ACCIDENTAL DEATH AND DISMEMBERMENT INSURANCE

The plan provides accidental death and dismemberment insurance coverage in an amount equal to your basic group life insurance (two times your current annual salary). Coverage is provided 24 hours per day, anywhere in the world, for any accident resulting in death, dismemberment, paralysis, loss of use, or loss of speech or hearing.

If you sustain an injury caused by an accident occurring while the policy is in force which results in one of the following losses, within 365 days of the accident, the benefit shown will be paid to you. In the case of accidental death, the benefit will be paid to the beneficiary you have named to receive your group life insurance benefits. Benefits are payable in accordance with the following schedule:

Schedule of Benefits

- Loss of life 100%
- Loss of both hands or both feet 100%
- Loss of one hand and one foot 100%
- Loss of the entire sight of both eyes 100%
- Loss of one hand and the entire sight of one eye -100%
- Loss of one foot and the entire sight of one eye -100%
- Loss of use of both arms or both legs or both hands 100%
- Loss of speech and loss of hearing in both ears − 100%
- Quadriplegia 200%
- Paraplegia 200%
- Hemiplegia 200%
- Loss of or loss of use of one arm or one leg 100%
- Loss of or loss of use of one hand or one foot 100%
- Loss of the entire sight of one eye -100%
- Loss of speech or loss of hearing in both ears 100%
- Loss of thumb and index finger on one hand -662/3%
- Loss of four fingers on one hand 66 2/3%
- Loss of hearing in one ear -662/3%
- Loss of all the toes on one foot $-33 \frac{1}{3}$ %

"Loss" means complete loss by severance except that in the case of loss of sight, speech or hearing, it means loss beyond remedy by surgical or other means.

"Loss of use" means total loss of ability to perform every action and service the arm, hand or leg was able to perform before the accident.

No more that the principal sum will be paid for all losses resulting from one accident.

General Limitations:

No benefits will be paid for loss resulting from or associated with the following:

- 1. suicide or intentionally self-inflicted injury, regardless of the person's state of mind and whether or not they were able to understand the nature and consequences of their actions;
- 2. viral or bacterial infections, except pyogenic infections occurring through the injury for which loss is being claimed;
- 3. disease or infirmity;
- 4. medical or surgical treatment, except surgical reattachment;
- 5. service, including part-time or temporary service, in the armed forces of any country;
- 6. war, insurrection, or voluntary participation in a riot.

The following additional benefits are also included; please see your employee booklet for details:

Repatriation Benefit	Occupational Training Benefit	Identification Benefit
Wheelchair Benefit	Seat Belt Benefit	
Education Benefit	Hospital Confinement Benefit	Workplace Modification Benefit
	Daycare Benefit	Family Transportation Benefit

WAIVER OF PREMIUM PROVISION

If an insured member becomes totally disabled before age 65, the group life, dependent life, accidental death and dismemberment, voluntary accidental death and dismemberment, voluntary term life insurance, and critical illness may be continued to age 65 without payment of premiums. To have premiums waived, claims forms must be submitted with 10 months from date of disability and the employee must be totally disabled for at least 119 calendar days.

HEALTH INSURANCE BENEFITS FOR MEMBERS AND DEPENDENTS

Hospital Benefit

If you or any of your insured dependents are confined in a hospital coverage is provided at 100% to a covered maximum of \$85 per day.

Prescription Drug Benefit

The program will pay the ingredient cost of eligible drugs (including oral contraceptives and insulin), you are responsible to pay the co-pay, which will be the equivalent of the pharmacists dispensing fee plus any applicable surcharge over the ingredient cost. The drug plan provides coverage for most drugs which require a prescription by law, however, but does not provide coverage for over-the-counter drugs, cough or cold preparations or nicotine products. The Government of Newfoundland and Labrador, through a consultation process with the insurer and drug experts, determines the drugs that are covered under the plan, and typically follows the recommendations of The Canadian Expert Drug Advisory Committee. There is no guarantee or obligation expressed or implied that all drugs recommended by physicians will be covered by the plan. Some drugs may require special authorization, details of the special authorization process are outlined in the online "GNL Summary of Benefits" booklet. Please see change to retirement benefits for employees hired on or after April 1, 2020.

Vision Care Benefit

You and your insured dependents are covered for the following vision care expenses:

- a. Eye examinations, including refractions, when they are performed by a licensed ophthalmologist or optometrist, and coverage is not available under your public health plan in your province of residence. The maximum covered expense is \$70 in a calendar year for a dependent child under age 18 and \$70 every 3 calendar years for any other person. Expenses for eye examinations are reimbursed at 80%.
- b. Adults (18+) one pair of glasses/contacts every 3 calendar years, reimbursed at 100% to a maximum of \$250 regardless of lens type.
- c. Dependent Children (17 and under) if there is a change in prescription, one pair of glasses/contacts every calendar year, reimbursed at 100% to a maximum of \$250 regardless of lens type.

- d. Coverage for Laser Eye Surgery when performed by a licensed ophthalmologist. Maximum covered expense is \$450 in a person's lifetime. No benefits will be paid for glasses or contact lenses purchased within 6 calendar years of the date of a claim for laser eye surgery.
- e. Up to 100% of eligible expenses to a maximum of \$250 every 2 calendar years for the purchase of contact lenses prescribed by a licensed ophthalmologist for severe corneal astigmatism, severe corneal scarring, keratoconus or aphakia, provided vision in the better eye cannot be corrected to the 20/40 level by glasses. Dependent children will be eligible for this benefit once in any calendar year, provided that a change in the strength of the prescription is required.
- f. One pair of eyeglasses when prescribed by an ophthalmologist, optometrist, or optician following non-refractive eye surgery, limited to a maximum covered expense of \$200 in a person's lifetime; and
- g. 50% of the cost of visual training or remedial therapy.

Extended Health Benefit

Reimbursement is provided for many types of services, such as registered nurse, physiotherapist, wheelchairs, braces, crutches, ambulance service, chiropractors, to name a few. Preauthorization is required for the rental and/or purchase of all durable equipment and all Nursing Care/Home Care benefits. Certain dollar amounts or time period maximums apply. It is important to note that reimbursement under the extended health care benefit is made at 80% of covered eligible expenses up to \$5,000; expenses over \$5,000 and less than \$10,000 are reimbursed at 90%, and expenses over \$10,000 are reimbursed at 100% in any calendar year. Where no maximum eligible expense is noted, reasonable and customary rates will apply. Please consult your online employee benefit booklet for details. Please see change to retirement benefits for employees hired on or after April 1, 2020.

Services not Covered Under the Supplementary Health Insurance Program

You and/or your dependents are not covered for medical expenses incurred as a result of any of the following:

- Expenses private insurers are not permitted to cover by law
- Services or supplies for which a charge is made only because you have insurance coverage
- The portion of the expense for services or supplies that is payable by the government public health plan in your home province, whether or not you are actually covered under the government public health plan
- Any portion of services or supplies which you are entitled to receive, or for which you are entitled to a benefit or reimbursement, by law or under a plan that is legislated, funded or administered in whole or in part by a provincial / federal government plan, without regard to whether coverage would have otherwise been available under this plan
- Services or supplies that do not represent reasonable treatment
- Services or supplies associated with:
- treatment performed only for cosmetic purposes
- recreation or sports rather than with other daily living activities

- the diagnosis or treatment of infertility
- contraception, other than contraceptive drugs and products containing a contraceptive drug
- Services or supplies associated with a covered service or supply, unless specifically listed
 as a covered service or supply or determined by Canada Life to be a covered service or
 supply
- Extra medical supplies that are spares or alternates
- Services or supplies received out-of-province in Canada unless you are covered by the government health plan in your home province and Canada Life would have paid benefits for the same services or supplies if they had been received in your home province
- Expenses arising from war, insurrection, or voluntary participation in a riot
- Chronic care
- Podiatric treatments for which a portion of the cost is payable under the Ontario Health Insurance Plan (OHIP). Benefits for these services are payable only after the maximum annual OHIP benefit has been paid
- Vision care services and supplies required by an Employer as a condition of employment
- Prescription sunglasses and safety glasses

Group Travel Insurance

The group travel plan covers a wide range of benefits which may be required as a result of an accident or unexpected illness incurred outside the province while travelling on business or vacation. The insurer will pay 100% of the reasonable and customary charges (subject to any benefit maximums) for expenses, such as hospital, physician, return home and other expenses as outlined in the employee booklet. Coverage under Group Travel Insurance is limited to a maximum of ninety (90) days per trip for travel within Canada. Coverage commences from the actual date of departure from your province of residence. Coverage under Group Travel Insurance is limited to thirty (30) days per trip for travel outside Canada. Coverage commences from the actual date of departure from Canada.

A person with an existing medical condition must be stable for 3 months prior to travelling. Stable means there has been no period of hospitalization, no increase or modification in treatment or prescribed medication, or no symptom for which a reasonably prudent person would consult a physician. Stable dosage does not apply to diabetics.

Additional coverage is available from Canada Life on an optional pay all basis.

OPTIONAL BENEFITS

Optional Group Life Insurance

This plan is available on an optional, employee-pay-all basis and you may apply to purchase additional group life insurance coverage for you and/or your spouse. Coverage is available from a minimum of \$10,000 to a maximum of \$300,000 in increments of \$10,000.

Optional Accidental Death and Dismemberment Insurance

This plan is available on an optional, employee-pay-all basis and enables you to purchase additional amounts of accidental death and dismemberment insurance on an employee and/or family plan basis. Coverage is available from a minimum of \$10,000 to a maximum of \$300,000 in \$10,000 increments.

Optional Long Term Disability Insurance

This plan is available to you on an optional and employee-pay-all basis. Long term disability insurance may provide disability benefits for periods of total disability which exceed 119 days. To be eligible for this benefit, you must be under the age of 65 and be a member of either the Public Service Pension Plan or the Uniformed Services Pension Plan. To be eligible for LTD benefits, claim forms must be submitted with 10 months from date of disability and the employee must be totally disabled for at least 119 calendar days.

Optional Dental Care Insurance

This plan is available to you and your insured dependents on an optional and employee-pay-all basis. Coverage is available for basic and major restorative dental procedures.

Optional Critical Illness Insurance

This plan is available to you and your dependents on an optional and employee-pay-all basis. Critical Illness Insurance will provide a lump sum payment to insured employees in the event he/she and/or dependents are afflicted, while coverage is in force, with a critical condition as defined in the policy.

GENERAL INFORMATION

For the purpose of the group insurance program, the following definition of dependent is applicable:

Spouse

- (a) an individual to whom you are legally married; or
- (b) an individual of the same or opposite sex who has been publicly represented as your spouse for at least one year.

Dependent Children

- your or your spouse's unmarried, natural, adopted, foster or step-children, including a child of an unmarried minor dependent, who are:
- (a) under 21 years of age and dependent upon you for support and maintenance; or
- (b) under 25 years of age and in full-time attendance at a university or similar institution and dependent upon you for support and maintenance; or

(c) age 21 or over who is incapacitated for a continuous period beginning before age 21 or while a full-time student and before age 25. A child is considered functionally impaired if they are incapable of supporting themselves due to a physical or psychiatric disorder.

Children of your spouse are considered dependents only if:

- they are also your children; or
- your spouse is living with you and has custody of the children

This plan does not cover a spouse or dependent child who is not a resident in Canada nor does it cover any child who is working more than 30 hour per week, unless the child is a full-time student.

Eligibility

- all full-time permanent employees working a regular work week are required to participate on the first day of employment
- all part-time permanent employees working a minimum of 50% of the regular work week are required to participate from the first day of employment
- all full-time temporary employees, if hired for a period of more than three months, are
 required to participate from the first day of employment. Full-time temporary employees
 who are hired for a period of less than three months, who have their contract extended for
 an additional period of at least three months, are required to participate on the date of
 notification that their employment was extended.
- contractual employees, coverage subject to the insurance policy eligibility rules and the specific provisions of the employment contract.

Retiree Benefits:

- eligibility and premium share for retired employees is outlined in the Letter of Understanding "Other Post-Employment Benefits ("OPEB") Eligibility.
- employees who elect early retirement will continue to be insured under the program as if active employees. Group life and accidental death and dismemberment insurance benefits will be calculated on the annual superannuation benefits. Optional Long Term Disability and Critical Illness will terminate. Basic Life and Basic AD&D coverage will be reduced on the first of the month following the date of retirement or age 65, whichever occurs first.
- for retirees, upon attainment of age 65, all basic life and optional life insurance policies terminate. At age 65, if you have been insured for a period of five years immediately prior to your 65th birthday, you may be eligible for a reduced paid-up life insurance policy on the first of the month following attainment of age 65, which will remain in force throughout your lifetime. You may also be eligible to continue your supplementary health and group travel insurance plans on a 50/50 cost-shared basis.
- Please note: All employees hired on or after **April 1, 2020**, upon retirement from government, will have different supplementary health coverage. Drug costs will be

reimbursed at 70 per cent on ingredient cost only, with dispensing fees paid by members, up to an out of pocket maximum of \$4,000 and reimbursed at 100 per cent thereafter. Other non-drug benefits will have a \$1,500 maximum coverage.

- Employees hired on or before March 31, 2020 will continue to have drugs covered at 100% of ingredient costs for approved drugs and will not be restricted to the \$1,500 cap on non-drug benefits in retirement.
- Notwithstanding the above, government maintains its right to administer and/or modify the post-employment group insurance programs.
- In the event of your death, your surviving spouse and any insurable dependent children, who on the date of your death was insured under the plan, may have the option of continuing in the group insurance program.

EMPLOYEE AND RETIREE RESPONSIBILITY

You should note that you have responsibilities to fulfill. Your responsibilities include, but are not limited to, the following:

- For ensuring that you have applied for the coverage you wish to have for yourself on your enrolment forms and your dependents within the appropriate time frames.
- To change your coverage from single to family within the appropriate time frame. If the coverage is not changed within 12 months of acquiring your first eligible dependent an Evidence of Insurability on Dependents is required for approval.
- To add a spouse to this plan in the event that he or she loses coverage under another plan within a 31 day period following the loss of coverage to avoid having to provide medical evidence.
- For examining payroll deductions for each pay period for all group insurance benefits. Examples would include family versus single coverage and optional benefit premiums particularly when you have requested changes in coverage and at the annual renewal date when the premiums are adjusted. This will ensure accuracy and allow for corrections on a timely basis. Coverage details can be confirmed through pay stubs, your plan administrator, Employers online benefit statements (where available), and the insurance carrier (by visiting www.canadalife.com and signing into Groupnet for plan members or by calling 1-844-349-5656).
- For amending your coverage to delete any coverage you no longer require. Contributions which you have paid are not refundable if they were consistent with the application on file.
- For effecting conversion of the coverage eligible to be converted upon the earlier of termination of employment or at age 65.

- For accurately completing the necessary forms required for continuing benefits while on maternity leave, sick leave, special leave without pay, retirement, etc. It is extremely important these arrangements be made prior to commencing eligible leave. For continuation of group life and health insurance while on temporary lay-off or on unpaid leave you are responsible for the payment of the full premium amount (Employer/employee contributions) and failure to remit will result in termination of coverage. You are also responsible for the payment of the full premium amount (Employer/employee contributions) if you are a temporary/hourly employee and you maintain benefits during a pay period when you have not worked and have not received pay. Failure to remit premiums will result in the termination of coverage.
- For providing appropriate claim information necessary to process LTD and/or Waiver of Premium claims as well as to ensure notice of claim/proof of claim where necessary has been provided within appropriate time frames as required under the contract.
- For providing appropriate medical information necessary to add a dependant as functionally impaired to continue their coverage beyond the age a dependant would otherwise terminate based on contract guidelines
- For completing the appropriate forms accurately, completely, and within applicable timeframes for such things as change of address, addition or deletion of a dependent, and other significant matters that can change or otherwise affect your coverage.
- To register overage student dependents between age 21 and 25 at the beginning of each school year. Failure to do so may impact coverage.
- Reviewing the online employee benefit booklet, contacting the insurance carrier and/or your organization's plan administrator to ensure you have a sound knowledge of the benefits available, extent of coverage, eligibility criteria, exclusions, restrictions, medical underwriting requirements, conversion options, continuation of benefits, predeterminations and other important requirements of the program.
- Providing proof of the purchase of pension service that may reduce LTD premiums. Premiums will only be adjusted when the plan administrator has been notified and received verification despite the date the purchase may have occurred.
- For notifying your plan administrator if the deletion of an overage dependant requires a change in your premiums from family to single coverage.

*SCHEDULE H



Labrador Benefits Agreement

Between

Her Majesty The Queen In Right of Newfoundland (Represented Herein by the Treasury Board)

College of the North Atlantic

Labrador-Grenfell Regional Health Authority

Municipal Assessment Agency Inc.

Newfoundland and Labrador Housing Corporation

Newfoundland and Labrador School Boards Association

Newfoundland Liquor Corporation

And

Canadian Union of Public Employees

Newfoundland and Labrador Association of Public & Private Employees

Registered Nurses' Union Newfoundland and Labrador

Newfoundland and Labrador Teachers Association

Royal Newfoundland Constabulary Association

Signed: February 10, 2020 Expires: March 31, 2022

ARTICLE 1

SCOPE

1.1 This Agreement is applicable to all employees in Labrador whose Employers are signatory to this agreement, represented by the Canadian Union of Public Employees, the Newfoundland and Labrador Association of Public & Private Employees, the Registered Nurses' Union Newfoundland and Labrador, the Newfoundland and Labrador Teachers Association and the Royal Newfoundland Constabulary Association. The terms of the agreement will be considered to form an integral part of all collective agreements.

ARTICLE 2

DURATION

*2.1 This agreement shall be effective from the February 10, 2020, and shall remain in full force and effect until March 31, 2022. It shall be renewed thereafter through the normal process of collective bargaining utilized by each of the employee groups, or, with the consent of the parties, will be renewed through joint negotiations. At the request of either party negotiations shall commence six (6) months prior to the expiry date of this agreement.

ARTICLE 3

LABRADOR ALLOWANCE

*3.1 Labrador Allowance for employees covered by this agreement shall be paid in accordance with Schedule "A."

	LABRADOR	ALLOWANCE	
	DATE	SINGLE	DEPENDENT
GROUP 1	10-Feb-20	2939	5878
	1-Apr-20	2979	5918
	1-Apr-21	3019	5958
GROUP 2	10-Feb-20	3402	6793
	1-Apr-20	3442	6833
	1-Apr-21	3482	6873
GROUP 3	10-Feb-20	3558	7089
	1-Apr-20	3598	7129
	1-Apr-21	3638	7169

In the case of spouses who are both employed by Her Majesty the Queen in Right of Newfoundland and Labrador as represented by Treasury Board, or a Board, Agency or Commission, the total amount paid to both employees shall not exceed the dependent rate for the allowance contained in this article. This allowance shall be paid to employees on a pro-rated basis in accordance with their hours of work excluding overtime.

*3.2 Labrador Benefits will be paid to employees for periods of maternity, parental and adoption leave.

ARTICLE 4

TRAVEL ALLOWANCE

*4.1 Employees covered by this agreement shall receive a travel allowance to help offset the costs of travel to areas outside of Labrador based on the following rates per employee and his/her dependent(s). The travel allowance shall be paid out during the pay period following April 15th at the rate in effect on April 15th of the year in which the allowance is to be paid.

	TRAVEL .		
	DATE	EMPLOYEE	DEPENDENT
GROUP 1	10-Feb-20	911	702
	1-Apr-20	950	702
GROUP 2	10-Feb-20	962	754
	1-Apr-20	1001	754
GROUP 3	10-Feb-20	1015	806
	1-Apr-20	1054	806

- *4.2 *(a) This allowance shall be paid to employees in the first pay period following April 15th of each year on a pro-rated basis in accordance with their hours of work in the previous twelve (12) month period, excluding overtime. The amount of travel allowance to be paid shall be based on the number of dependents at March 31st of each year.
 - *(b) An employee retiring, resigning or otherwise terminating employment shall be entitled to a proportional payment of travel allowance as determined in 4.2 (a) based on the employee's hours of work in the current fiscal year. In the case of death the payment shall be made to the employee's beneficiary or estate.

- 4.3 (a) For the purpose of calculating this benefit the following leaves shall be considered as hours of work:
 - (i) Maternity Leave/Parental Leave/Adoption Leave
 - (ii) Injury-on-Duty/Worker's Compensation Leave
 - (iii) Paid Leaves
 - (iv) Any other period of unpaid leave for which the employee is eligible to accrue service under the respective collective agreement
 - (b) The provisions of 4.3 (a) will not apply when the employee would 'otherwise have been laid off.
 - (c) The provision of 4.3(a) (iv) will apply only to employees who have worked or have been credited with hours of work under 4.3(a) (i), (ii) or (iii) for a period of 20 days in the aggregate in the qualifying period.
- 4.4 In the case of spouses who are both employed Her Majesty the Queen in Right of Newfoundland and Labrador as represented by Treasury Board, or a Board, Agency or Commission, each spouse shall receive the employee travel allowance, but only one spouse shall claim the benefit for dependents.
- 4.5 The travel benefit available to the Royal Newfoundland Constabulary Association under their Collective Agreement and to teachers under Article 25 of the NLTA Labrador West Collective Agreement shall continue to apply except in cases where Article 4 of this joint agreement provides a greater benefit. E.g. Members of the RNCA would continue to receive the employee travel benefit under their collective agreement unless the employee travel benefit in this joint agreement is greater. In addition to the employee benefit under the RNCA collective agreement, members of the RNCA shall also receive the dependent benefit under the joint agreement.

ARTICLE 5

LEAVE

5.1 Employees covered by this agreement shall receive three (3) non-cumulative, paid leave days in the aggregate per year. This leave will only be utilized when the employee is delayed from returning to the community due to interruptions to a transportation service occurring within Labrador. This article shall also apply where there has been an interruption to a transportation service occurring at the last departure point directly to Labrador.

ARTICLE 6

EXISTING GREATER BENEFITS

6.1 No provision of this agreement shall have the effect of reducing any benefit for any employee which exists in each applicable employee group collective agreement outlined in Article 1.

ARTICLE 7

DEFINITIONS

- 7.1 **Dependent** for the purpose of this Agreement, dependent means a spouse, whether of the same or opposite gender, and children under eighteen (18) years of age, or twenty-four (24) years of age if the child is in full time attendance at a school or post-secondary institution or any child that remains in the direct care of the parent in the same household because the dependent is medically verified as disabled and under twenty-four (24) years of age.
- *7.2 **Spouse** for the purpose of this agreement, spouse means a person to whom an employee is legally married or a person in a common law relationship with whom the employee has lived with for more than one (1) year.

SCHEDULE A

COMMUNITY GROUPING

The employee's community grouping shall be determined by the location of their headquarters.

GROUP 1

Happy Valley/Goose Bay North West River Sheshatshiu Wabush Labrador City Churchill Falls

GROUP 2

Red Bay
L'Anse au Loup
L'Anse au Clair
Forteau
Pinware
West St. Modest
Mud Lake
Cartwright
Mary's Harbour
Port Hope Simpson
St. Lewis
Charlottetown
Lodge Bay
Paradise River

GROUP 3

Rigolet
William's Harbour
Norman's Bay
Black Tickle
Pinsent's Arm
Makkovik
Postville
Hopedale
Davis Inlet/Natuashish
Nain

MEMORANDUM OF UNDERSTANDING

Re: Nurses Committee

The parties acknowledge that the Registered Nurses' Union Newfoundland and Labrador (RNUNL) have indicated that they have issues of concern unique to Nurses who live and work in Labrador and that the RNUNL will attempt to address these concerns through a committee which will be established subsequent to these negotiations.

MEMORANDUM OF UNDERSTANDING

Re: Labrador Benefits Agreement - Interpretation

In an effort to clarify the interpretation of certain items contained in the Labrador Benefits Agreement the parties agree to the following:

- Article 4.2(b) refers to employees who terminate employment, (i.e. are not on layoff status and do not have recall rights). These employees have their Travel Allowance paid out based on the hours worked in the current year and it shall be paid out at the rate in effect on the date employment is terminated.
- 2) For the purposes of Article 4.4 it is agreed that an employee may refuse to claim the employee benefit if it is to their benefit to have their spouse claim them as a dependant. Employees who exercise this option will not be entitled to any portion of the Employee Travel Allowance. It is incumbent on the employee to communicate this choice to their respective Employer(s).
- 3) Notwithstanding Schedule A, the following employee shall be entitled, on a without prejudice basis, to the rates applicable to Mud Lake as long as they remain within the employ of their current Employer and continue to permanently reside in Mud Lake:

Vyann Kerby, Health Labrador Corporation

This agreement is effective from April 1, 2013 and shall expire upon the renewal of the Labrador Benefits Agreement expiring March 31, 2016.

4) For the purposes of clarification and in accordance with Article 3.1 (Labrador Allowance), Article 4.1 (Travel Allowance) and Article 9.1 (Definitions), benefits are applicable for the fiscal year (April 1 to March 31) in which a dependent reaches 18 years of age or 24 years of age, if the dependent is in full time attendance at a school or post-secondary institution. Full time attendance shall be determined by the educational institution in which a dependent is registered.

For example:

If a dependent reaches 18 years of age on January 1, 2011 and is no longer enrolled as a full time student in a post secondary institution, they would be eligible for the travel benefit payable April 15, 2011.

Earl Hann

On behalf of the Unions

Lisa Curran

On Behalf of the Employers

Jon Cunan

Date Schriding 10, 2020

February 10, 2020

Mr. Steve Brooks Executive Director Newfoundland and Labrador Teachers' Association 3 Kenmount Road St. John's, NL AlB 1W1

Dear Mr. Brooks:

This letter is to confirm that for teachers in Labrador, the payment of the travel allowance provided under Article 4 - Travel Allowance, of the Joint Agreement on Labrador Benefits shall be calculated for the school year, September to June, but shall be paid in accordance with the provisions of Article 4 of the Joint Agreement.

Yours truly, Genilunan

Lisa Curran

Chief Negotiator Collective Bargaining Division

Original letter dated December 20, 1999

February 10, 2020

Mr. Jerry Earle President NAPE 330 Portugal Cove Place St. John's, NL, A1B 3M9

Re: Labrador Benefits Allowance and Travel Allowance for School Board Student Assistants (NAPE) and School Board Support Staff (NAPE and CUPE)

Within three (3) months of the date of signing of the Labrador Benefits Agreements, officials from the Human Resource Secretariat (HRS) of the Government of Newfoundland and Labrador, the Newfoundland and Labrador English School District (NLESD) and Newfoundland and Labrador Association of Public and Private Employees (NAPE) will meet to discuss the application of the Labrador Benefit Allowance and the Travel Allowance for employees who work less than full time hours as outlined in the above respective collective agreements.

Yours truly, Juni Cuuran

Lisa Curran

Chief Negotiator

Collective Bargaining Division

IN WITNESS WHEREOF the parties hereto have executed this Agreement the day and year

first before written. SIGNED on behalf of Treasury Board representing Her Majesty the Queen in Right of Newfoundland by the Honourable Tom Osborne, President of Treasury Board, in the presence of the witness hereto subscribing: President of Treasury Board SIGNED on behalf of the College of the North Atlantic College of the North Atlantic SIGNED on hehalf of Labrador-Grenfell Regional Health Authority Labrador-Grenfell RHA SIGNED on behalf of the Newfoundland and Labrador Housing Corporation

SIGNED on behalf of the Newfoundland and Labrador School Boards Association

${\bf SIGNED}$ on behalf of the Newfoundland Liquor Corporation

Witness	Newfoundland Liquor Corporation
SIGNED on behalf of the Municipal Assessmen	t Agency
Witness	Municipal Assessment Agency
SIGNED on behalf of the Canadian Union of Pu	ablic Employees
Eart an Witness	CUPE
SIGNED on behalf of the Registered Nurses' U	Inion Newfoundland and Labrador Debbue Loward RNUNL
SIGNED on behalf of the Newfoundland and La	abrador Teachers' Association

SIGNED on behalf of the Royal Newfoundland Constabulary Association

Witness RNCA

 ${\bf SIGNED}$ on behalf of the Newfoundland & Labrador Association of Public and Private Employees

Saltann
Witness
NAPE

SCHEDULE I

<u>DEFERRED SALARY LEAVE PLAN HEALTH CARE</u> (effective January 1, 1995)

- 1. Definitions
- 2. Introduction
- 3. Administration
- 4. Eligibility
- 5. Enrolment
- 6. Salary Benefits During Deferral and Leave Periods
- 7. Withdrawal from the Plan
- 8. Returning from Leave

DEFERRED SALARY LEAVE PLAN

It is the Employer's policy that, where possible, all employees have access to a deferred salary leave plan, designed to help plan and finance a leave of absence for periods of 6 - 12 months. The deferral period may not exceed six years.

1. <u>Definitions</u>

Review Committee: a committee, comprised of members representing the Employer and

employee groups, which will review issues related to the Plan

Deferral period: that period of time during which participating employees defer a portion

of their salary

Leave Period: that period of time, immediately following the deferral period, during

which participating employees are on a leave of absence from work

Taxation year: based on a calendar year January 1 to December 31 (inclusive)

2. Introduction

The Deferred Salary Leave Plan permits employees to finance a leave of absence by deferring a maximum of 33 1/3% of their salary (before deductions) in a taxation year. The period over which employees may defer their salaries may not exceed six years.

The deferred portions of employees' salaries are deposited into an account with the Government of Newfoundland and Labrador. This account accrues interest which must be paid at the end of each calendar year. Because this interest is considered to be income from employment, it is subject to income tax for the year in which it was earned. The interest rate, on the deferred portion of employees' salaries will be the rate of interest earned by Government on its bank accounts.

The minimum leave period is six (6) months, except where the employee is attending a designated educational institution on a full time basis; in this case the minimum period of leave is three months. The maximum period of leave is twelve (12) consecutive months, starting immediately after the salary deferral period.

Revenue Canada regulates all deferred salary leave programs and has the ability to cancel any Employer plan if the regulations are violated. For taxation purposes Revenue Canada requires that at the end of the leave period, employees return to the Employer under whom they participated in the deferred salary leave program for, at least the same amount of time as the leave period. As such, the Deferred Salary Leave Plan cannot serve as an early retirement program.

3. Administration

The Deferred Salary Leave Plan will be administered by the Government Accounting Division of Treasury Board and the Personnel Department of the Employer. The Chief Executive Officer will be responsible for approving deferred salary leave for employees who wish to use the plan. Approval forms will be sent to the Payroll Department where employees will be set up on the deferred salary payroll. The system will calculate the deferred portion of employees regular pay cheques. This deferred portion will then be transferred into an interest bearing account.

Employees who wish to opt out of the plan must inform their Chief Executive Officer in writing. The Chief Executive Officer will then forward this information to the Review Committee who will review the case and decide if it can be approved. The committee will consist of representation from:

Government Accounting Division, Treasury Board Personnel Policy Division, Treasury Board Collective Bargaining Division, Treasury Board Debt Management Division, Finance The Employer NLHBA Union Involved

The Committee will meet on a regular basis to approve requests to withdraw from the plan and/or address any other issues that are not directly dealt with in this policy.

4. Eligibility

Permanent employees with a minimum of twelve (12) months continuous service may apply for enrolment in the Deferred Salary Leave Plan.

5. <u>Enrolment</u>

Employees must make written application to their Chief Executive Officer, providing as much advance notice as possible (preferably at least three (3) months).

Employees will be informed of their approval to participate in the Plan within three months of their request. Approval of requests is dependent on the operational requirements of the department during the employee's anticipated absence.

Table I demonstrates the various Plan options: the amount of salary an employee would have to defer; the period of time over which the salary is deferred; the year in which the leave of absence is to be taken; and the employee's average wage over the entire period of Plan participation.

TABLE I

PLAN OPTION	PERCENTAGE OF SALARY DEFERRED	DEFERRAL PERIOD	LEAVE PERIOD	% OF NORMAL SALARY DURING PLAN
1 over 1.5	33 1/3%	1 year	2nd year (6 months.)	66 2/3%
2 over 3	33 1/3%	2 years	3rd year	66 2/3%
3 over 4	25%	3 years	4th year	75%
4 over 5	20%	4 years	5th year	80%
5 over 6	16 2/3%	5 years	6th year	83 1/3%
6 over 7	14%	6 years	7th year	86%

Over the period of Plan participation, employees' salaries plus the percentage of salaries deferred must equal 100% of employees' pre-plan salaries. For example, employees who choose the 3 over 4 option must defer 25% of their salaries over a 3 year period resulting in a net salary, over 4 years, of 75% of their pre-plan salaries.

6. <u>Salary And Benefits During Deferral and Leave Periods</u>

During the deferral period, employees continue to receive their normal salary less the amount they have chosen to contribute to the Plan.

During the leave period, employees' gross annual salaries will consist of the sum of the contributions made to the Deferred Salary Leave Plan during the deferral period, plus interest. Salary will be received through the normal payroll procedures.

Income tax information slips (T4) for the completion of participating employees' tax return will reflect that portion of salary actually received in the taxation year.

The interest earned on the deferred portion of employees salaries will be considered to be employment income. This interest amount is taxable as employment income and will be included on T4 slips. Income tax information slips will be issued yearly as the interest is paid.

Employment status will be that of leave without pay. The provisions and cost-sharing arrangements for employee benefits will be consistent with the appropriate sections of the Employer policies and Collective Agreements. Appendix I outlines the effect the Deferred Salary Leave Plan will have on all benefit/deduction programs during both the deferral and leave periods.

```
While on leave, no payments will be made for:

overtime;
call back;
stand-by;
automobile allowance;
Labrador allowance;
sick leave;
family responsibility leave;
annual leave;
paid leave;
statutory holidays; or,
any other monetary compensation provided to employees who are at work.
```

No annual leave, sick leave or paid leave credits will be issued by the Employer while employees are on the leave period portion of the Deferred Salary Leave Plan. This time will not be counted toward the requirements for service to achieve additional annual leave or paid leave credits. The leave period, however, will not be considered a break on service.

Employees may purchase their leave period for pension credits upon their return to work.

7. Withdrawal From The Plan

Once approved for participation in the Deferred Salary Leave Plan, employees may withdraw from the Plan, prior to the leave period, only under exceptional circumstances such as:

extreme financial hardship;

death;

total and permanent disability;

transfer to another position where Plan participation is not approved; and/or

lay-off, termination or resignation.

Employees who withdraw from the Plan will receive a refund of their contributions plus the accrued interest on the contributions. Refunds will be made within 50 days of the approval to withdraw from the Plan.

The lump sum payment refunded to the employee is subject to Canada Pension Plan contribution (retroactive to the beginning of the deferral period).

Unless specifically requested by the employee, the Employer will apply the lump sum income tax rate to the refund amount.

8. Returning From Leave

Employees returning form their leave periods will be:

assigned to the same or equivalent position held prior to going on leave; and eligible for the same step in the salary scale that they were paid prior to going on leave.

APPENDIX I

Benefit/Deduction Program	Benefits During the Deferral Period	Benefits During the Leave Period
Provincial Pension	Contribution will be based on normal gross salary, not salary received	No pension payments will be deducted during the leave period
Group Life Insurance/Health Insurance	Coverage and premiums will be based on normal gross salary not salary received; Employer cost sharing willcontinue	Coverage continuation is optional; employee is required to pay full cost of premiums while on leave
Voluntary Accident Insurance/Dental Insurance	Participation optional. employee pays full cost of premium	Participation optional employee pays full cost of premium
Long Term Disability Insurance	Coverage and premiums will be based on normal gross salary not salary received	No coverage available during the leave period
Union Dues (Where applicable)	Payable on that portion of salary actually received or based on Union Constitution	Payable on that portion of salary actually received or based on union constitution
Income Tax	Payable on that portion of salary actually received	Payable on that portion of salary actually received
Canada Pension Plan	Payable on that portion of salary actually received	Payable on that portion of salary actually received
Employment Insurance	Payable on normal gross salary	No deductions are taken as there are no insurable earnings during the leave period
Annual Leave	Normal accrual rates apply	No accrual during leave period. Time accrued prior to the leave period may be taken, subject to approval, before the leave period begins or after the leave period ends
Sick Leave	Normal accrual rates apply	No accrual during leave period. Sick leave will not be paid during leave period
Paid Leave (Where Applicable)	Normal accrual rates apply	No accrual during leave period. Time accrued prior to the leave period may be taken, subject to approval, before the leave period begins or after the leave period ends
Pay Increments	Normal application	Step progression delayed by the length of the leave period

DEFERRED SALARY LEAVE PLAN APPLICATION AND APPROVAL REQUEST FORM

1.	Name: SIN:				
			(Home)		
		(Work)		
 3. 	Work Location: Department: Work Area: Job Title: Annual Salary: Date of Entry into	\$ Employment with the Empl	ployer <u>//.</u> Year Mt Day		
	Plan Option	Percentage of Salary Deferred	Deferral Period	Leave Period	
	1 over 1.5	33 1/3%	1 year	2nd year (6 months)	
	2 over 3	33 1/3%	2 years	3rd year	
	3 over 4	25%	3 years	4th year	
	4 over 5	20%	4 years	5th year	
	5 over 6	16 2/3%	5 years	6th year	
	6 over 7	14%	6 years	7th year	
perio	d noted above. I hav	to/_/	dicated in Section 3 a Deferred Salary Leav		
Emp	lovee Signature		Date		

5.(a)	Approval is hereby granted for the deferral of	f salary as outlined in Section 3.
Adm	inistrator's Signature	Date
Appr	oval cannot be granted as requested for the follo	owing reason(s):
Adm	inistrator's Signature	Date
b)	Approval for salary deferral period: / Year M	to // / Year Mt Day
	Percentage of salary deferred/year: Amount Deferred: \$	
	Percentage of salary maintained: Amount Maintained: \$	
	Payroll Deduction: \$	
	Approval for leave period: / / Year Mt Day	to / / Year Mt Day
c.c.	Employee Debt Management (Finance)	

SCHEDULE J

JOB SHARING

In accordance with the letter regarding job sharing on page 136 of the RNUNL Collective Agreement signed January 13, 1995, below are guidelines to assist Employers and nurses in job sharing arrangements. The guidelines in no way obligate either party to enter into a job sharing arrangement, but rather provide guidance where the parties mutually agree to enter into such an arrangement. Nothing in these guidelines shall be interpreted to limit the right of any Employer and the union to vary these guidelines, and nothing shall be interpreted so as to alter any existing job sharing agreements between the Union and any Employer. It is hoped that job sharing will improve the quality of life for nurses who wish to avail of it.

The suggested guidelines are as follows:

- 1) that job sharing be voluntary employee initiated arrangement;
- 2) that a job sharing arrangement be defined as one normal full-time position shared by two nurses each working 50% of the position;
- 3) that the employees have the ability to self-schedule based on one normal full-time rotation;
- 4) that all the benefits of the collective agreement be shared equally between the nurses. These benefits not be any less than the part-time benefits currently outlined in the collective agreement, (ie. all benefits applicable to part-time be extended to job sharing as well).

Employees who occupy a full-time position would initiate a potential job sharing arrangement and would enter into discussion with the Employer to establish the arrangement. Such an arrangement would have to be agreeable to the Employer, and the Employer would be permitted to limit the number of full-time positions available for job sharing.

Once the job sharing arrangement has been approved by the Employer for a particular full-time position, a job sharing partner would have to be located for the initial incumbent. The selection of a job sharing partner would have to be conducted via an agreed upon mechanism. Such a mechanism could include job posting as per the collective agreement, or via a "job sharing register". The selection of a job sharing partner would be as per Article 24 of the RNUNL Collective Agreement.

Once a job sharing partner has been selected, the full-time position would be shared between the two employees on a 50-50 basis. The employees shall work 50% of the normal regular hours over a 4 to 6 week period. The employee shall receive a cheque each pay period as if they worked 50% of the normal regular hours in that pay period, regardless that the employee's self-scheduling may result in one job sharing partner working greater than 50% of the normal regular hours in the pay period being paid. This will ensure consistency for payroll and benefits. Benefits would be applied as would be for a part-time employee (ie. on a pro-rated basis).

Employees engaged in a job sharing arrangement would be permitted to engage in self-scheduling. The Employer would post the applicable work schedule as they would for a full-time employee, however, the job sharing employees would decide amongst themselves whom shall work which particular shifts on a 50-50 basis. Once the work schedule has been decided upon, the employees would notify and seek the approval of the Employer one week in advance of the commencement date of the work outlined, provided the schedule is posted two (2) weeks in advance.

The employees engaged in a job sharing arrangement would be permitted to have a trial period. The trial period would be a specified period of time agreed upon between the Employer and the employees engaged in the arrangement, during which time either employee would be permitted to revert to their former position. For reasons other than an unsuccessful trial period, should one of the employees engaged in job sharing wish to discontinue the arrangement the following options may be offered:

- 1. The full time position may be offered to the remaining employee.
- 2. The remaining employee may request to have the job sharing arrangement continued, and a new job sharing partner selected.

NOTE: These options should be discussed, between the Employer and the job sharing partners, and agreed upon prior to the approval of the job sharing arrangement. A notice period to the Employer for the discontinuance of a job sharing arrangement should also be pre-determined.

Should there be a need for annual leave or sick leave replacement for either of the employees engaged in the job sharing arrangement, there will be no requirement for the other job sharing partner to fill in as a replacement. However, the option should be given to the employees to replace one another for annual leave purposes. Should the employees agree to this practice, the Employer must be notified. In the case of sick leave replacement, the Employer would be expected to replace as normal for short term sick leave. For long term sick leave, the job sharing partner could be given the option to cover the shifts.

SCHEDULE K

SENIORITY NUMBERS

The following schedule outlines a pilot project for the development of a seniority number system for employees who are members of RNUNL.

PART A

- 1. In order to simplify the method of calculating seniority, each employee of each Regional Board will be assigned a "seniority number" based upon a formula outlined in this agreement. The assignment of this number in no way affects any part of the collective agreement or transition agreement(s) except as specifically outlined in this schedule. The introduction of the seniority numbering system is intended only as a means of recording seniority.
- 2. Each Employer agrees to post a seniority list showing seniority recorded in hours as well as dates of hire. If the employee protests their seniority, they must submit a written request for a review of their seniority within two (2) months of posting of the seniority list. This review must be completed by the Employer within six (6) months from date of posting of the seniority list. This list will show seniority up to and including September 30, 1999, (December 31st, 1996 for employees of The Western Health Care Corporation Institutions Board and Community Health Western), or the last pay day in September, 1999, depending on which date the individual Employer has traditionally used to calculate seniority for the purposes of meeting the Employer's obligations under Clause 23.02 of the collective agreement. Seniority for all employees will be adjusted, where necessary, to ensure that no employee has an advantage by reason of different schedules/pay periods etc. as of the applicable date.
- 3. In addition to all permanent employees in permanent positions, for the purposes of the initial assignment of seniority numbers, "permanent employees" shall include:
 - (a) all permanent employees on layoff status who have retained recall rights;
 - (b) all permanent employees occupying temporary positions while retaining their permanent status:
 - (c) all permanent employees on approved leaves of absence;
- 4. As of effective date of new agreement, the seniority of all employees, excluding employees of The Western Health Care Corporation Institutions Board and Community Health Western, shall be considered capped as of September 30th, 1999, or the last pay day in September, 1999, depending on which date the individual Employer has traditionally used to calculate seniority for the purposes of meeting the Employer's obligations under Clause 23.02 of the collective agreement. All employees other than permanent employees shall be considered to have seniority based on their respective dates of hire. All references to seniority in the collective agreement and transition agreement(s) shall be considered to be references to an employee's seniority as of September 30th, 1999 (December 31st, 1996 for employees of the

Western Health Care Institutions Board and Community Health Western), or the last pay day in September, 1999, depending on which date the individual Employer has traditionally used to calculate seniority for the purposes of meeting the Employer's obligations under Clause 24.02 of the collective agreement, or an employee's date of hire, as applicable, until such date as seniority numbers are actually assigned by the particular Employer. The introduction of seniority numbers will not be used retroactively.

- 5. Following the protest period, all permanent employees will be issued a seniority number. The permanent employee with the most seniority hours will be issued number "1" and so on down the applicable seniority list. Once all permanent employees have been issued seniority numbers, employees holding casual or temporary status as of September 30th, 1999 (December 31st, 1996 for employees of The Western Health Care Institutions Board and Community Health Western), or the last pay day in September, 1999, depending on which date the individual Employer has traditionally used to calculate seniority for the purposes of meeting the Employer's obligations under Clause 24.02 of the collective agreement, will be issued a seniority number based on date of hire and will retain this number even if they eventually obtains permanent status after the applicable date. All new employees hired after the applicable date will then be issued seniority numbers based on their respective dates of hire. The lower the seniority number, the higher will be the employee's seniority.
- 6. Two or more Employers in a geographic region may elect to issue consecutive numbers to make provisions to accommodate existing clauses in transition agreements, or to simplify possible future amalgamations. Should two or more Employers issue consecutive numbers, separate seniority lists shall be maintained at any site where the collective agreement or transition agreement(s) require separate seniority lists. The seniority list(s) for those sites shall only reflect the names, dates of hire, seniority by classification and seniority numbers of those employees employed at that site.
- 7. Notwithstanding Paragraph 6 above, should the Union and the Employer make future agreement(s) to amalgamate seniority lists, the amalgamations can be accomplished by subdividing existing seniority numbers. (E.g. 16A, 16B, etc.)
- 8. When employees have the same number of seniority hours, or the same dates of hire, the order of seniority will be determined by drawing names from a hat based on procedures established in consultation with the union.
- 9. The Union agrees that after the "protest period", there will be no further requests to check seniority hours. The Union and the Employer(s) will work jointly to ensure that all employees are made aware of the importance of verifying their seniority/date of hire before the issuance of seniority numbers.
- 10. Issues concerning private agreements relating to seniority for leaves of absence applicable to one or more sites under a particular Employer will be discussed between the local Employer and the Union, with the intention that all Employees be treated equally.
- 11. This agreement may be terminated by either party with the provision of six (6) months written notice. The parties agree to meet during the first two (2) months of the notice period to attempt to resolve any differences that may arise.

- 12. Should the Employer or the union elect to revert to a system based upon hours:
 - 1. All employees who held seniority as of the applicable date (December 31st, 1996 for employees of The Western Health Care Institutions Board and Community Health Western), will be credited with additional seniority based upon full time hours from the applicable date (December 31st, 1996 for employees of The Western Health Care Institutions Board and Community Health Western), for the duration of time the seniority numbering system was in effect.
 - 2. All other employees hired after the applicable date will be credited with seniority based upon full time hours from their respective dates of hire, for the duration of time the seniority numbering system was in effect.

This will ensure that all employees who held seniority as of the applicable date (December 31st, 1996 for employees of The Western Health Care Institutions Board and Community Health Western), will be placed in the same relative positions on the seniority list(s) as they would have been in had the seniority numbering system not been introduced.

3. Subject to the provisions of Clause 10 of Appendix D, the Seniority Number Agreement in the Western Transition Agreement, employees who have accepted part-time or job sharing positions in reliance on the seniority numbering system, will be given the opportunity to revert to their previous employment status in accordance with the layoff and recall provisions in effect with their particular Employer, until June 30th, 2001.

If either party serves notice of terminating the seniority by number pilot project, the employees who wish to revert to their previous employment status shall notify the Employer in writing of their desire to do so within one (1) month of the formal notice being served. Should the Employer give notice to terminate the agreement in the 30 days prior to June 30th, 2001, employees will still have 30 days in which to give notice of their desire to revert to their previous employment status.

13. The Union and the Employer may extend this pilot project (excluding Clause 12.3) into the next collective agreement or may agree to make seniority numbering a permanent method of recording seniority.

PART B

Language will be added to the applicable clauses of the collective agreement and will supercede the existing clauses while the seniority numbering system is in effect. The new language will replace references to the word "seniority" with the words "seniority number".

The intent is to outline both the old and the new clauses in the body of the collective agreement. This will eliminate the need to refer to the schedule to find the applicable clauses.

SCHEDULE L

TRANSITION AGREEMENTS

Eastern Regional Health Authority (including Faith Based Homes) Central Regional Health Authority Western Regional Health Authority Labrador-Grenfell Regional Health Authority

TRANSITION AGREEMENT

BETWEEN

HER MAJESTY THE QUEEN IN RIGHT OF NEWFOUNDLAND

(Represented herein by the Treasury Board)

AND

THE NEWFOUNDLAND AND LABRADOR HEALTH BOARDS ASSOCIATION

REPRESENTING
THE EASTERN REGIONAL INTEGRATED HEALTH AUTHORITY
(Referred herein as the Employer)

AND

THE NEWFOUNDLAND AND LABRADOR NURSES' UNION

DATE February 26, 2010

This agreement made this 26th day of two thousand and nine in accordance with Clause 40.02 of the Collective Agreement between Her Majesty the Queen in Right of Newfoundland and Labrador (represented herein by the Treasury Board), the Newfoundland and Labrador Health Boards Association, and the Newfoundland and Labrador Nurses' Union (NLNU), herein referred to as the Collective Agreement. It is recognized that if there is any conflict with the Collective Agreement the Transition Agreement shall apply unless otherwise specified herein. This Agreement governs all members of the NLNU who are employees of the Eastern Regional Integrated Health Authority (referred herein as the Employer).

It is further recognized that effective the date this Transition Agreement comes into effect, the three transition agreements and their associated letters of understanding with each of the Employer's Legacy Organizations, as outlined below will be repealed and replaced with this Transition Agreement:

- Transition Agreement between Avalon Health Care Institutions Board and The Newfoundland and Labrador Nurses' Union; dated May 8, 1997.
- Transition Agreement between The Health Care Corporation of St. John's and The Newfoundland and Labrador Nurses' Union; dated June 7, 1996.
- Transition Agreement between Peninsulas Health Care Corporation and The Newfoundland and Labrador Nurses' Union; dated July 2002.

1. Bargaining Unit Structure

- 1.1. As per the order of the Labour Relations Board (LRB) dated July 16, 2008 (the LRB Order), a copy of which is attached as Schedule "A", it was determined that there be established a single Nurses (NS) bargaining unit in the classifications covered by the NLNU and the St. John's Home Care Program Collective Agreements, but not including Nurses employed at the Agnes Pratt Home, Pentecostal Senior Citizens Home, St. Luke's Home, St. Patrick's Mercy Home and the Salvation Army Glenbrook Lodge; and that the seniority lists for the existing twelve (12) Nurses (NS) bargaining units be dovetailed into a single seniority list for the Nurses (NS) bargaining unit.
- 1.2. The reference throughout the Collective Agreement to bargaining units refers to employees who are NLNU members as defined by the Collective Agreement at all sites operated by the Employer as one single bargaining unit.
- 1.3. The Employer will post the new single seniority list in all work sites prior to January 1st, 2010. Subject to protests and grievances the new single seniority list will be effective January 1st, 2010.

Seniority will be transferred to the new single bargaining unit recognizing the following principles:

a. Employees whose names appear on more than one existing seniority list may combine their seniority providing that no employee will be credited with more than equivalent full time hours in any calendar year.

- b. Article 38.04 (a) and (b) of the Collective Agreement between the parties executed June 20th, 2009 (the Collective Agreement) will be applied. Employees who dropped a Step(s) on the salary scale by reason of the resignation(s) referenced in Article 38.04(a) and (b) shall have their former step reinstated with full retroactivity.
- Article 38.04 of the Collective Agreement will not apply to employees who resign after January 1st, 2010.
- d. The parties agree that within two (2) months of the date of signing of this Transition Agreement, the Employer and the Union will review grievances related to these issues and where the only issue was the loss of seniority which is being reinstated by this agreement, the grievance(s) shall be withdrawn.
- 1.4. Employees will have two (2) months from the date the seniority list is posted to submit a protest in writing to the Human Resources Department outlining the specific reason why they feel their seniority to be incorrect. The Human Resources Department will have five (5) months from when the seniority protest was received to provide the employee with a written response outlining the outcome of their seniority protest and the detail of what adjustments, if any, were required. A copy of the response will be copied to the employees' personal file.

A revised seniority list incorporating corrections will be posted by August 31st, 2010.

Once the initial protest period has elapsed, all future protests will be completed in accordance with Article 24.03 of the Collective Agreement. Subject to the mutual agreement to the contrary, where displacement or staff changes have occurred, it will not be altered due to the discovery of errors not brought to the attention of the Employer during the protest period.

- 1.5. If the situation should develop whereby two (2) or more employees have the same seniority, the seniority shall be broken by random draw, i.e. drawing names from a hat. A Union representative, together with the individuals whose names are in the hat, or designates, may be present with management when the draw takes place.
- 1.6. As of April 1, 2005, the employees referenced in the LRB Order were transferred to the Employer. The Employer recognizes the service of each of the employees as it was recognized by the previous Employers prior to the new organization assuming governance.

2. Staff Changes

2.1. Effective January 1st, 2010, when filling permanent vacancies they shall be filled on the basis of one region wide bargaining unit and one single seniority list in accordance with Article 25 of the Collective Agreement.

- 2.2. Effective January 1st, 2010, when filling temporary vacancies which exceed or are expected to exceed 6 months they shall be filled on the basis of one region wide bargaining unit and one single seniority list in accordance with Article 25 of the Collective Agreement.
- 2.3. All employees will be assigned a primary site of work. It is recognized that certain employees may be required to work on a multi-site basis or have region-wide responsibilities. These individuals will be given written notification of this requirement. All other situations will be dealt with in accordance with Article 25:09 of the Collective Agreement.

2.4. Permanent Employees in Temporary Positions

Subject to the approval of the Employer, a permanent employee who applies for and is accepted for a temporary position in accordance with 2.2 above, may revert to their former position upon completion of the temporary position. When Employer approval is obtained, permanent employees obtaining temporary employment shall maintain their permanent positions. The Employer shall notify the employee in writing of its decision to grant or deny position protection before the employee accepts the temporary work.

If there is a need to fill the individual's permanent position while the employee is in the temporary position, it will be filled on a temporary basis only and the Union agrees to waive the definition of "Temporary Employee" found in Clause 2.01(w) of the Collective Agreement for this purpose only.

3. Transfer of Services Within Eastern Health

- 3.1. For the purpose of this Article, a service transfer is the transfer of a service from one physical location to another within Eastern Health. Service Transfer does not include a change in the management reporting structure or a location change within the same facility or site operated by the Employer.
- 3.2. Eastern Health reserves the right to transfer staff as a result of service transfer. Employees who are transferred shall retain seniority, service, and other recognized earned and portable benefits.
- 3.3. Employees who are affected by service transfer shall not have access to lay-off or displacement options unless:
 - 3.3.1. there is an associated downsizing of positions within the service, or
 - 3.3.2. the service is being transferred to a worksite located 85 kms (driving distance) or greater from the employee's current worksite.
- 3.4. Where there is an associated downsizing of positions within the service the Employer will offer nurses, on the basis of seniority, the choice of transferring within the service or accessing lay-off / displacement. If there are not enough nurses interested in transferring

with a service the Employer reserves the right to transfer nurses commencing with the most junior nurse.

Employees considering accepting layoff should fully explore the Employment Insurance implications before exercising this option.

- 3.5. Eastern Health will select employees on the basis of seniority when examining transfers to comparable positions within the service.
- 3.6. Eastern Health will provide a familiarization period up to but not exceeding two weeks, to the employees transferring with their Department / Program to a new site. This will assist them in becoming acquainted with essential information such as policies and procedures, routines, location of supplies and equipment, fire and disaster plans, etc.
- 3.7. Subject to the bumping process, and subject to 3.3.2 above, temporary employees replacing permanent employees in a department / program that is transferred will transfer with the department / program if the permanent employee they are replacing is transferring with the department / program.
- 3.8. Subject to the bumping process, casual employees and temporary employees on lay-off will be transferred with the program / service they usually work in.
- 3.9. Subject to the bumping process, employees on layoff and employees off work on various forms of leave recognized in the collective agreement shall have the same options available to them in the event of service transfer as though they were working.

4. Grievance Management Procedure

The Proposed Western Pilot referenced on p.303 of the Collective Agreement will be expanded to include the Employer.

5. Statutory Holidays

- 5.1. Subject to Article 18.01(a), employees shall observe Thanksgiving Day and Armistice Day as the two additional statutory holidays.
- 5.2. Subject to Article 18.01(b), Public Health Nurses, shall observe any day in the calendar year that the employee chooses as the one additional statutory holiday. This day must be taken within the calendar year.

6. Professional Practice/Labour Management Committees

The parties agree to the establishment of a joint committee to deal with professional practice and labour management issues functioning on local and sub-regional levels. Where there are both professional practice issues and labour management issues on the agenda, professional practice issues shall be dealt with first.

6.1. Local Committees:

Local committees will be governed by Article 5:02 of the Collective Agreement. The structure of these committees is as follows:

6.1.1.Local Committee Structure by Region:

- i. St. John's Region with one committee for each of the following:
 - a) Adult Acute Care represented by employees of the General Hospital, L.A. Miller Centre and Cancer Care Program
 - b) Adult Acute Care represented by employees of the St. Clare's Mercy Hospital
 - c) Adult Mental Health represented by employees of the Waterford Hospital
 - d) Child Health Program represented by employees of the Janeway Children's Health and Rehabilitation Centre including employees of the Children's Mental Health Division
 - e) Dr. Walter Templeman Hospital
 - f) Centre for Nursing Studies
 - g) Long-term Care represented by employees of the Hoyles-Escasoni Complex and Masonic Park Nursing Home
 - h) Community Health Services represented by employees within the St. John's region.
- ii. Rural Avalon with one committee for each of the following:
 - a) Health Services represented by employees of the Carbonear General Hospital and Dr. AA Wilkinson Health Centre
 - b) Long-term Care represented by employees of the Harbour Lodge Nursing Home and Interfaith Citizen's Home
 - c) Health Services represented by Placentia Health Centre and Dr. William H. Newhook Community Health Centre
 - d) Community Health Services within Rural Avalon.
- iii. Peninsulas with one committee within each of the following communities:
 - a) Clarenville (Integrated and Community Health Services)
 - b) Bonavista (Integrated and Community Health Services)
 - c) Burin (Integrated and Community Health Services)
 - d) Grand Bank (Integrated and Community Health Services)
 - e) St. Lawrence (Integrated and Community Health Services)
- 6.1.2. The proposed local committee structure is subject to change with mutual agreement of the parties.

6.2. Sub-regional Committees:

6.2.1. Sub-regional Structure:

In addition to the local Professional Practices Committees/Labour Management Committees as per 6.1, sub-regional committees will be established for the St. John's, Rural Avalon and Peninsulas regions.

6.2.2. Function of the Committees:

The sub-regional committees will address issues that are regional in nature and that are not the subject matter of a grievance. Items from the local committees unresolved after three local committee meetings will be automatically forwarded to the sub-regional committee. Urgent issues may be referred to the sub-regional committees immediately, if mutually agreed by the co-chairs.

6.2.3. Frequency of Meetings:

Sub-regional committees will meet on a quarterly basis or at the request of either one of the parties, unless mutually agreed otherwise. Where possible, members will endeavor to attend all meetings in-person, with the understanding that a minimum of every second meeting will be held face-to-face.

Employees who are required to attend these meetings will be compensated at straight time or given equivalent time off work in lieu. Employees who are required to travel to attend these meetings will be compensated in accordance with the Employer's Travel Policy.

6.2.4. Membership of Committees:

Sub-regional committees will be comprised of representatives from both the Union and the Employer as follows:

- A union representative from each of the local committees.
- The Manager of Human Resources Consulting Services.
- A professional practice representative.
- A Program/Corporate Director representative from each of the following acute, longterm and community care.

Committee members may invite other individuals to committee meetings, depending on the issues being discussed. Members are also encouraged to have a delegate who can attend meetings in their absence.

Meetings will be chaired alternately by the Employer and the Union.

6.2.5. Minutes and Agenda

The Agenda will be circulated to committee members seven (7) calendar days prior to each meeting.

Minutes shall be approved and signed by the co-chairs prior to circulation. Minutes shall be forwarded to Executive Management, Committee Members and Local Presidents. Union representatives will be responsible for ensuring that a copy is forwarded to the union.

Approved Minutes will be also posted electronically.

6.2.6. The proposed sub-regional committee structure is subject to change with mutual agreement of the parties.

7. Employees Working with More than One Legacy Organization

7.1. Existing Multi-Position employees

Employees, who, prior to the effective date of the Transition Agreement, occupied more than one position with the Employer, may continue to hold these positions provided the following criteria are met:

- 7.1.1.No employee will be permitted to hold positions whose combined hours of work exceed 75 hours biweekly (i.e. greater than 1.0 FTE); or
- 7.1.2. No employee will be permitted to hold more than one permanent full-time position.

Employees whose combined positions are greater than 1.0 FTE shall advise their employer of which position they wish to drop so their total hours of work do not exceed 75 hours biweekly. Employees may request in writing in accordance with Article 25.10 to reduce their hours in one of their positions so that their total positions do not exceed 1.0 FTE.

7.2. Employees, who, prior to the effective date of this Transition Agreement, held a permanent part-time position with one of the Employer's legacy organizations and worked on a casual basis with another of the Employer's legacy organizations, will no longer be paid as if they were separate employees. All hours worked in excess of their permanent part-time hours will be treated as additional shifts and the employee will be paid as per the Letter of Understanding "Part-time employees working additional shifts".

7.3. Future Multi-site positions

- 7.3.1.Casual Employees may be permitted to work on a casual basis at any of the Employer's sites or in multiple clinical areas, to a maximum of three, subject to the following:
 - 7.3.1.1. Approval is required from each manager responsible for each of the clinical areas where the employee wishes to work.
 - 7.3.1.2. The employee must advise each manager of their work schedule.

- 7.3.1.3. One manager must take the responsibility for the employee's payroll. Employees will be paid in accordance with Article 2.01 (a) of the Collective Agreement.
- 7.3.1.4. Each manager is responsible for ensuring that the employee receives proper orientation to their clinical area.
- 7.3.2. Permanent part-time employees may be permitted to work additional shifts outside their area/unit, subject to the following:
 - 7.3.2.1. Approval of the employee's permanent part-time manager is required.
 - 7.3.2.2. The employee must advise each manager of their work schedule.
 - 7.3.2.3. The employee's permanent part-time manager must take the responsibility for the employee's payroli.
 - 7.3.2.4. If the employee works in excess of his/her normal hours as per Article 9 of the Collective Agreement, the manager of the area where the hours worked were incurred is responsible for the additional premium cost.
 - 7.3.2.5. Extra shifts are to be offered to permanent part-time employees and casual employees in the work area/unit prior to offering additional work to employees outside their area/unit.
 - 7.3.2.6. The above criteria are to be read in conjunction with those outlined in the Letter of Understanding "Part-time employees working additional shifts", with the exception of clause (b) of this letter.
 - 7.3.2.7. The manager of the area where the employee wishes to work additional shifts is responsible for ensuring that the employee receives proper orientation to this clinical area.
- 7.3.3. Permanent part-time employees may be permitted to hold a second permanent part-time position with the Employer, subject to the following:
 - 7.3.3.1. The combined hours of work must not exceed 75 hours biweekly (i.e. greater than 1.0 FTE).
 - 7.3.3.2. Approval of the employee's current and future permanent part-time manager is required.
 - 7.3.3.3. The manager of each area reserves the right to change the employee's hours of work in accordance with Article 8 of the Collective Agreement. If this change creates a scheduling conflict, the employee will be required to forfeit one of their permanent part-time positions.
 - 7.3.3.4. The employee must advise each manager of their work schedule. The employee's initial permanent part-time manager must take the responsibility for the employee's payroll.
 - 7.3.3.5. If the employee works in excess of his/her normal hours as per Article 9 of the Collective Agreement, the manager of the area where the hours worked were incurred is responsible for the additional premium cost.
 - 7.3.3.6. The manager of the area where the employee is seeking a second permanent part-time position is responsible for ensuring that the employee receives proper orientation to this clinical area.

7.3.4. In exceptional circumstances only, permanent employees may be permitted to work in excess of their permanent full-time hours in an area outside their clinical area. Prior approval from the employee's manager is required and all hours worked in excess of his/her normal hours will be considered overtime and the employee will be compensated at the applicable overtime rate in accordance with Article 9 of the Collective Agreement.

The employee must inform their manager of all excess hours worked in another clinical area.

7.4. Review of Multi-site Arrangements

All multi-site arrangements will be reviewed and evaluated one-year from the effective date of the signing of the Transition Agreement.

7.5. Forfeiture of Multi-site Arrangements

Employees, who hold more than one position with the employer, whether on a permanent or temporary basis, may be permitted to forfeit one of the positions, as follows:

- 7.5.1. An employee who holds multiple permanent positions may forfeit one of their permanent positions and retain their accumulated seniority and status. Employees will be required to provide their manager with three weeks written notice of their intent to forfeit one of their permanent positions.
- 7.5.2. An employee who is working in a permanent and temporary position concurrently and who subsequently forfeits their permanent positions will retain their accumulated seniority but will forfeit their permanent status. Employees will be required to provide their manager with three weeks written notice of their intent to forfeit their permanent position.

October 29, 2009

Mr. John Vivian
Executive Director
Newfoundland and Labrador Nurses' Union
P.O Box 416
St. John's NL
A1C 5J9

Re: St. John's Home Care Program Collective Agreement

Dear Mr. Vivian:

This letter will confirm our understanding that this transition agreement will also apply to those employees covered by the St. John's Home Care Program Collective Agreement.

Additionally, this letter will also confirm our understanding that once the terms of this transition agreement are implemented; a committee with the appropriate management and union representatives from the community sector will be struck to deal with the issues not dealt within this transition agreement. However, the parties also agree that this committee may be waived if a provincial process were put in place to address the same issues.

Sincerely,

Stephen C. Dodge Vice-President, People and Information Services Eastern Health

The Eastern Regional Integrated Health Authority /NLNU Transition Agreement, February 18, 2010

Signed on behalf of the Eastern Regional Integrated Hea in the presence of the witness hereto subscribed	Ith Authority by its proper officers
Many Nichels Witness	Louise Jones
Signed on behalf of the Newfoundland and Labrador He officers in accordance with the constitution and in the subscribed	presence of the witness hereto
Witness	Elizabeth San
Signed on behalf of the Newfoundland and Labrador Nurs the presence of the witness hereto subscribed	ses' Union by its proper officers in
Witness	Jebal bu.
Signed on behalf of Treasury Board by its proper officer hereto subscribed	•
Witness	Marshell



Labour Relations Board

5170

IN THE MATTER OF

Public Service Collective Bargaining Act

- and -

Eastern Regional Integrated Health Authority

Applicant

- and -

Newfoundland and Labrador Nurses' Union

Respondent

WHEREAS pursuant to Sections 44, 44.1 and 45 of the Public Service Collective Burgaining Act, the Labour Relations Board has received an application from Eastern Regional Integrated Health Authority for orders consolidating existing classifications of Nurses of the Employer who are members of separate bargaining units represented by the Newfoundland and Labrador Nurses' Union into a single Nurses (NS) bargaining unit of employees of the Employer, with dovetailing of seniority lists and consequential relief;

AND WHEREAS the Board, following consideration of the application and the representations of the interested parties; and noting that the Respondent consents to the orders requested in the application, has decided to grant the application;

NOW THEREFORE it is hereby ordered by the Labour Relations Board that:

- Eastern Regional Integrated Health Authority is now the Employer and is a party to or is bound by Collective Agreements with the Newfoundland and Labrador Nurses' Union which are as follows:
 - (a) a Collective Agreement made among Newfoundland and Labrador Nurses' Union, Her Majesty the Queen in Right of Newfoundland and Labrador, represented by Treasury Board and Newfoundland and Labrador Health Boards Association in effect from 26 January 2007 to 30 June 2008; and including a Transition Agreement dated 8 May 1997 for Avalon Health Care Institutions Board; a Transition Agreement dated 7 June 1996 for Health Care Corporation of St. John's; a Sub-Agreement dated 7 June 1996 for Health Care Corporation of St. John's and St. John's Regional Health and Community Services Board; and a Transition Agreement dated July 2002 for Peninsules Health Care Corporation (the NLNU Collective Agreement);
 - (b) and a Collective Agreement made among Newfoundland and Labrador Nurses* Union, Her Majesty the Queen in Right of Newfoundland, represented by Treasury Board, St. John's Home Care Program and Newfoundland and Labrador Hospital and Nursing Home Association dated 7 February 1991 (the St. John's Home Care Program Collective Agreement)

5º Floor, Beothusk Bidg., 35 Crossie Place, P. C. Box 8703, St. John's, N.L. Conece, A19 4Lif Yersphone: (703) 729-2707 Facelraide: (700) 729-5730 Wew.gov.hicalifb

- and that there be established a single Nurses (NS) hargaining unit in the classifications
 covered by the NLNU and the St. John's Home Care Program Collective Agreements,
 but not including Nursea employed at the Agnes Pratt Home, Pentecostal Senior
 Citizens' Home, St. Luke's Home, St. Pantek's Mercy Home, and the Salvation Army
 Glenbrook Lodge;
- and that the seniority lists for the existing twelve (12) Nurses (NS) bargaining units be dovetailed into a single seniority list for the Nurses (NS) bargaining unit;
- and that the Newfoundland and Lahrador Nurses' Union continues to represent the members of the Nurses (NS) bargaining unit;
- and that the Collective Agreements currently in effect will remain in effect during the transition period or until a new Nurses (NS) Collective Agreement is signed between Eastern Regional Integrated Health Authority and Newfoundland and Labrador Nurses' Union;
- and that the Labour Relations Board retains jurisdiction to deal with any further
 questions or problems which may arise with respect to this application and to provide
 for the resolution of same.

Camello Executive Officer

THE official seal of the Board was hereunto affixed and attested to by the Chief Executive Officer of the Board at the City of St. John's in the Province of Newfoundland and Labrador this 16th day of July 2008

This Private Agreement is made pursuant to Article 3.03 of the RNUNL Provincial Collective Agreement between Eastern Regional Integrated Health Authority ("Eastern Health"), St. Patrick's Mercy Home, The Agnes Pratt Home, Saint Luke's Home, The Salvation Army Glenbrook Lodge (together the "faith-based" homes") and the Registered Nurses' Union Newfoundland and Labrador ("Union").

This Agreement is made upon the condition and understanding that it will not affect the parties' subsequent interpretation of the terms of the Collective Agreement except as provided herein.

Whereas there is a Labour Relations Board Order between Eastern Health and the Union dated July 16, 2008;

And, whereas registered nurses represented by the Union who are employed at the faithbased homes are excluded from the Labour Relations Board Order, and are part of separate seniority lists;

And, whereas the faith-based homes, Eastern Health, and the Union wish to merge the seniority lists of the faith-based homes with the seniority list of Eastern Health, the parties hereby agree that:

- The parties agree that the purpose of this Agreement is not meant to be interpreted
 in any manner whatsoever as an acknowledgement by any of the signatories as to
 the issue of who is the "Employer". Further, the parties agree and acknowledge
 that any party to this Agreement reserves the right to make representation in a
 formal setting at that party's sole discretion, be it through the Labour Relations
 Board or otherwise, as to the designation of who is the "Employer".
- Eastern Health and the faith-based homes will merge and dovetail their respective seniority lists for the Union, effective October 25, 2015, into a single seniority list for the Union with Eastern Health.
- All bargaining unit positions at Eastern Health and at the faith-based homes posted after October 25, 2015, shall be posted for all employees on the merged single seniority list for the Union.
- 4. The seniority of Union members employed at the faith-based homes will be transferred to the Eastern Health Union seniority list, and vice-a-versa, recognizing the following principles:
 - a. Employees whose names appear on more than one existing seniority list may combine their seniority providing that no employee will be credited with more than equivalent full time hours in any calendar year.
 - b. Article 38.04 (a) and (b) of the Collective Agreement between the parties signed on September 4, 2014 (the Collective Agreement), will be applied.

RNUNL and Eastern Health and Faith-Based Homes - Merger of Seniority Lists and Service

Employees who dropped a Step(s) on the salary scale by reason of the resignation(s) referenced in Article 38.04(a) and (b) shall have their former step reinstated with full retroactivity. Employees who would have advanced on the salary scale sooner, following April 1, 2005, if there had been a merger of seniority at that time, shall have the appropriate step reinstated with full retroactivity.

- Article 38.04 of the Collective Agreement will not apply to employees who resign after October 25, 2015.
- d. The parties agree that within two (2) months of the date of signing of this Agreement, Eastern Health, the faith-based homes, and the Union will review grievances related to these issues and where the only issue was the loss of seniority which is being reinstated by this agreement, the grievance(s) shall be withdrawn.
- 5. Subject to paragraph 6, employees will have two (2) months from the date that the merged single seniority list is posted to submit a protest in writing to the Human Resources Department outlining the specific reason why they feel their seniority to be incorrect. The Human Resources Department will have two (2) months from when the seniority protest was received to provide an employee with a written response outlining the outcome of their seniority protest and the detail of what adjustments, if any, were required. A copy of the response will be copied to the employees' personal file.

A revised seniority list incorporating corrections will be posted by January 31,

Once the initial protest period has elapsed, all future protests will be completed in accordance with Article 23.03 of the Collective Agreement. Subject to the mutual agreement of the parties to the contrary, where displacement or staff changes have occurred, it will not be altered due to the discovery of errors not brought to the attention of the Human Resources Department during the protest period.

- 6. The parties agree that, following the expiration of the protest period noted in paragraph 5 of this Agreement, if an employee comes forward with a request for an adjustment of seniority that has not been previously considered, the parties may consider that request and the parties may mutually agree in writing to make an appropriate adjustment of seniority.
- 7. If the situation should develop whereby two (2) or more employees have the same seniority, the seniority shall be broken by random draw. A Union representative, together with the individuals whose names are in the draw, or designates, are entitled to be present with management when the draw takes place.

RNUNL and Eastern Health and Paith-Based Homes - Merger of Seniority Lists and Service

- 8. As of April 1, 2005, Eastern Health recognizes the service of each of the Union employees at the faith-based homes for the purposes of the accumulation of paid benefits (annual leave, sick leave, service for severance pay, etc.) and the faith-based homes recognize the service of each of the Union employees at Eastern Health for the purpose of the accumulation of paid benefits.
- Any disputes arising out of this Agreement, failing agreement by the parties, shall be determined by the Grievance and Arbitration process and/or through the Labour Relations Board process.
- The parties agree that there will be no retroactive application whatsoever of any adjusted seniority hours to past job competitions, transfers of positions, layoffs, bumping or other issues involving seniority that arose prior to October 25, 2015.

These terms and conditions have been read and agreed upon by:

mi	Oct 20 2015
Registered Nurses' Union NL	Date
//	Judy Boone
	Witness
Marily Nichols	Oct 20, 2015
Eastern Regional Integrated Health Authority	Date 2. / /
	Jose Whele
1 14	Witness
14	Oct. 20, 2015
St. Patrick's Mercy Home	Date
	Witness
maybuland	Oct 22, 2015
St. Patrick's Mercy Honde	Date (201)
St. I Black S Mility Hydro	CHAILS

RNUNL and Eastern Health and Faith-Based Homes - Merger of Sentority Lists and Service

Witness

	V
Et III	Oct 23/11
The Agnes Pratt Home	Date
	- amelo
4	Witness
I dunitte Margo	Oct 23/15
The Agnes Pratt Home	Date Out A
	grand
	Witness
Galos.	22-007-2015
Saint Luke's Home	Date
	dela 'e
9	Witness
Welly Manie	Oct 22, 2015
Saint Luke's Home	Date
	Witness
	witness
-dtulias =	Coroku 21/2015
The Salvation Army Glenbrook Lodge	Date
1	Witness
1 1 1/	Williess
per Stelle	Oct 21/15
The Salvation Army Glenbrook Lodge	Date O
Ker Colo	Witness
	TI MINOR

RNUNL and Eastern Health and Faith-Based Homes - Merger of Sentority Lists and Service

TRANSITION AGREEMENT

BETWEEN

HER MAJESTY THE QUEEN IN RIGHT OF NEWFOUNDLAND

(Represented herein by the Treasury Board)

AND

THE NEWFOUNDLAND AND LABRADOR HEALTH BOARDS ASSOCIATION

REPRESENTING CENTRAL REGIONAL HEALTH AUTHORITY

AND

THE NEWFOUNDLAND AND LABRADOR NURSES' UNION

This agreement made this <u>J</u> day of <u>April</u> two thousand and ten in accordance with Clause 40.02 of the Collective Agreement between Her Majesty the Queen in Right of Newfoundland and Labrador (represented herein by the Treasury Board), the Newfoundland and Labrador Health Boards Association, and the Newfoundland and Labrador Nurses' Union (NLNU), herein referred to as the Collective Agreement. This Agreement governs all members of the NLNU who are employees of the Central Regional Health Authority.

Subject to the <u>Flexible Hours of Work Community Health</u>, Letter of Understanding dated June 19th, 2009 found in the Collective Agreement, it is further recognized that effective the date this Transition Agreement comes into effect, the transition agreements with each of the Employer's legacy organizations, as outlined below will be repealed and replaced with this Transition Agreement:

- Transition Agreement between Treasury Board, the Newfoundland and Labrador Health Care Association on behalf of the Central East Health Care Institutions Board and the Newfoundland and Labrador Nurses Union dated September 29th, 1997.
- Transition Agreement between Treasury Board, the Newfoundland and Labrador Health Care Association on behalf of the Central West Health Board and the Newfoundland and Labrador Nurses Union dated March 18th, 1997.
- Transition Agreement between Treasury Board, the Newfoundland and Labrador Health Care Association on behalf of the Health and Community Services-Central Region and the Newfoundland and Labrador Nurses Union dated January 1st, 1999.

1. Transfer of Business - Successor Rights

1.1 In accordance with Section 44 of the Public Service Collective Bargaining Act the parties to this agreement agree that there has been a "transfer of business" from the following health care employers:

Central East Health Care Institutions Board

AND

Central West Health Board

AND

Health and Community Services-Central Region

TO

Central Regional Health Authority

As per the order of the Labour Relations Board (LRB) dated November 12th, 2008 (the LRB Order), a copy of which is attached as Schedule "A", it was determined that there be established a single regional nurses (NS) bargaining unit in the classifications covered by the NLNU Collective Agreement and the Gander and District Continuing Care Program Collective Agreement, within the Central Regional Health Authority.

2. No Loss Of Seniority For The New NLNU Bargaining Unit

NLNU members shall not lose their seniority as a result of the transfer of business as outlined in Clause 1 of this agreement.

- 2.1 The Employer will post a new single seniority list in all work areas prior to June 1st, 2010. Subject to protests and grievances the new single seniority list will be effective June 1st, 2010.
- Employees will have two (2) months from the date the seniority list is posted to submit a protest in writing to the Human Resources department outlining the specific reason why they feel their seniority to be incorrect. The Human Resources department will have five (5) months from when the seniority protest was received to provide the employee with a written response outlining the outcome of their seniority protest and the detail of what adjustments, if any, were required. A copy of the response will be copied to the employee's personal file.
- A revised seniority list incorporating corrections will be posted by January 31st, 2011.
- Once the initial protest period has elapsed, all future protests will be completed in accordance with Article 24.03 of the Collective Agreement. Subject to the mutual agreement to the contrary, where displacement or staff changes have occurred, it will not be altered due to the discovery of errors not brought to the attention of the employer during the protest period.
- 2.5 If the situation should develop whereby two (2) or more employees have the same seniority, the seniority shall be broken by drawing names from a hat. A union representative, together with the individuals whose names are in the hat, or designates, may be present with management when the draw takes place.

3. Transfer of Service - Effective Date

As of April 1, 2005 the employees of the three (3) former employers outlined in Clause 1 of this agreement became employees of the Central Regional Health Authority. Central Regional Health Authority recognizes the service of each of the NLNU members as it was recognized by the previous employers prior to that date. Service for seniority purposes is to be recognized on a single, regional bargaining unit basis.

4. Professional Practice Committees

It is agreed that a minimum of two (2) professional practice committees will be established for community based nurses.

The parties further agree to the establishment of one regional professional practice committee which will meet at least twice per calendar year. These meetings shall be face to face. Additional meetings may occur as required.

The regional committee shall be comprised of representatives from the Union and the Employer and may include a Union representative and an Employer Representative from each of the local committees.

The regional committee will address issues that are regional in nature. Items from the local committees unresolved after three local committee meetings may be forwarded to the regional committee. Urgent issues may be referred to the regional committee immediately by mutual agreement between the parties.

5. Transfer of Services Within the Central Regional Health Authority

- 5.1 For the purpose of this Article, a service transfer is the transfer of a service from one community to another within Central Regional Health Authority.
- 5.2 Central Regional Health Authority reserves the right to transfer staff as a result of service transfer. Employees who are transferred shall retain seniority, service, and other recognized earned and portable benefits.
- 5.3 Employees who are affected by a service transfer shall have access to lay-off or displacement options if:
 - 5.3.1 There is an associated downsizing of positions within the service, or
 - 5.3.2 The service is being transferred to a worksite located 80 kms (driving distance) or greater from the employee's current worksite.
 - 5.3.3 The service is being transferred to a worksite that, while located less than 80 kms (driving distance) from the employee's current worksite, requires travel by ferry.
- Where there is an associated downsizing of positions within the service or the service is being transferred to a worksite located 80 kms (driving distance) or greater from the employee's current worksite, or the service is being transferred to a worksite that, while located less than 80 kms (driving distance) from the employee's current worksite, requires travel by ferry, the Employer will offer nurses, on the basis of seniority, the choice of transferring within the service or accessing lay-off / displacement. If there are not enough nurses interested in transferring with a service the Employer reserves the right to temporarily transfer

nurses commencing with the most junior nurse for a reasonable period of time while the employer is training replacement staff.

Employees considering accepting layoff should fully explore the Employment Insurance implications before exercising this option.

- 5.5 Central Regional Health Authority will provide a familiarization period up to but not exceeding two weeks, to the employees transferring with their Unit/Division to a new site. This will assist them in becoming acquainted with essential information such as policies and procedures, routines, location of supplies and equipment, fire and disaster plans, etc.
- Subject to the bumping process, temporary employees replacing permanent employees in a Unit/Division that is transferred will transfer with the Unit/Division if the permanent employee they are replacing is transferring with the Unit/Division.
- 5.7 Subject to the bumping process, employees on layoff and employees off work on various forms of leave recognized in the collective agreement shall have the same options available to them in the event of service transfer as though they were working.

6. Grievance Procedure

- Step 1: Grievance submitted to her/his immediate supervisor
- Step 2: Grievance submitted to Human Resources Manager/Consultant
- Step 3: Grievance submitted to Director of Strategic Human Resources Management.
- Step 4: Failing satisfactory settlement being reached in step 3, the referral to arbitration shall be submitted to the Director of Strategic Human Resources Management.

7. Posting of Temporary Positions

Temporary vacancies which exceed or are expected to exceed sixteen (16) weeks shall be posted and filled in accordance with the collective agreement.

8. Community Health Issues

The parties recognize that there are outstanding issues related to community health nursing which are not properly the subject of these transition discussions. The parties agree that a committee with appropriate management and union representatives from the community sector will be struck to deal with the issues not dealt with in this transition agreement. The committee will commence meetings prior to October 31st, 2010.

Schedule A



Labour Relations Board

5172 IN THE MATTER OF

Public Service Collective Bargaining Act

- and -

Central Regional Health Authority

Applicant

- and -

Newfoundland and Labrador Nurses' Union

Respondent

WHEREAS pursuant to Sections 44, 44.1 and 45 of the Public Service Collective Bargaining Act, the Labour Relations Board has received an application from Central Regional Health Authority for orders consolidating existing classifications of Nurses employed by the Employer who are members of separate bargaining units represented by the Newfoundland and Labrador Nurses' Union into a single Nurses (NS) bargaining unit, with dovetailing of seniority lists and consequential relief;

AND WHEREAS the Board, following consideration of the application and the representations of the interested parties; and noting that the Respondent consents to the orders requested in the application, has decided to grant the application;

NOW THEREFORE it is hereby ordered by the Labour Relations Board that:

- Central Regional Health Authority is now the Employer and is a party to or is bound by the existing NLNU Collective Agreement and the Gander and District Continuing Care Program Collective Agreement;
- and that there be established a single Nurses (NS) Bargaining Unit in the classifications covered by the NLNU Collective Agreement and the Gander and District Continuing Care Program Collective Agreement;
- and that the seniority lists for the existing three (3) Nurses (NS) bargaining units be dovetailed into a single seniority list for the Nurses (NS) bargaining unit;
- and that the Newfoundland and Labrador Nurses' Union continues to represent the members of the Nurses (NS) bargaining unit;
- and that the Collective Agreements currently in effect will remain in effect during the transition period or until a new Nurses (NS) Collective Agreement is signed between Central Regional Health Authority and Newfoundland and Labrador Nurses' Union;
- and that the Labour Relations Board retains jurisdiction to deal with any further
 questions or problems which may arise with respect to this application and to provide
 for the resolution of same.

5th Floor, Beothuck Bidg., 20 Crosble Place, P. O. Box 8700, St. John's, Nt., Canada, A1B 4J6 Telephone; (709) 729-2707 Facsimile; (709) 729-5738 www.gov.nl.ca/lrb THE official seal of the Board was hereunto affixed and attested to by the Chief Executive Officer of the Board at the City of St. John's in the Province of Newfoundland and Labrador this 12th day of November 2008

Canallo d Chief Executive Officer

DATED THIS _36th_ DAY OF _ Capa	2010
Debou Jonard Newfoundland and Labrador Nurses union	WIDNESS
CENTRAL REGIONAL HEALTH AUTHORITY	WITNESS
TREASURY BOARD	WITNESS
Elizabeth Lane NEWFOUNDLAND AND LABRADOR HEALTH BOARDS ASSOCIATION	<u>CPasons</u> WITNESS

TRANSITION AGREEMENT

BETWEEN

HER MAJESTY THE QUEEN IN RIGHT OF NEWFOUNDLAND

(Represented herein by the Treasury Board)

AND

THE NEWFOUNDLAND AND LABRADOR HEALTH BOARDS ASSOCIATION

REPRESENTING WESTERN REGIONAL HEALTH AUTHORITY

AND

THE NEWFOUNDLAND AND LABRADOR NURSES' UNION

DATE: 31-MARCH-2010

This agreement made this _______ day of _______ two thousand and ten in accordance with Clause 40.02 of the Collective Agreement between Her Majesty the Queen in Right of Newfoundland and Labrador (represented herein by the Treasury Board), the Newfoundland and Labrador Health Boards Association, and the Newfoundland and Labrador Nurses' Union (NLNU), herein referred to as the Collective Agreement. This Agreement governs all members of the NLNU who are employees of Western Health (referred herein as the Employer).

It is further recognized that effective the date this Transition Agreement comes into effect, the transition agreement with each of the Employer's Legacy Organizations, as outlined below will be repealed and replaced with this Transition Agreement:

 Transition Agreement between Western Health Care Corporation and Western Regional Health and Community Services Board and The Newfoundland and Labrador Nurses' Union; dated June 2, 1997

1. Transfer of Business - Successor Rights

1.1 In accordance with Section 44 of the Public Service Collective Bargaining Act the parties to this agreement agree that there has been a "transfer of business" from the following health care employers:

Western Health Care Corporation

AND

Western Regional Health and Community Services Board

TO

Western Regional Health Authority

1.2 As per the order of the Labour Relations Board (LRB) dated January 17, 2007 (the LRB Order), a copy of which is attached as Schedule "A", it was determined that there be established a single regional Nurses (NS) bargaining unit in the classifications covered by the NLNU Collective Agreement within the Western Regional Health Authority.

2. No Loss Of Seniority For The New NLNU Bargaining Unit

NLNU members shall not lose their seniority as a result of the transfer of business as outlined in Clause 1 of this agreement.

3. Transfer of Service - Effective Date

As of April 1, 2005 the employees of the two (2) former employers outlined in Clause 1 of this agreement became employees of the Western Regional Health Authority. Western Regional Health Authority recognizes the service of each of the NLNU members as it was recognized by the previous employers prior to that date. Service for seniority purposes is to be recognized on a single, regional bargaining unit basis.

4. Professional Practice/Labour Management Committees

The parties agree to the establishment of joint committees to deal with professional practice/labour management issues functioning on a local level. Where there are both professional practice issues and labour management issues on the agenda, professional practice issues shall be dealt with first.

One local professional practice/labour management committee will be established for:

- Bay St. George Long Term Care Centre
- Sir Thomas Roddick Hospital (including Population Health Bay St. George area)
- Port aux Basques
- Burgeo
- Norris Point
- Port Saunders
- Corner Brook/Bay of Islands

The parties further agree to the establishment of one regional professional practice/labour management committee which will meet at least three times per calendar year. At least one of these three meetings will be face to face.

The regional committee shall be comprised of representatives from the Union and the Employer and shall include at least one Union representative and one Employer Representative from each of the local committees.

The regional committee will address issues that are regional in nature. Items from the local committees unresolved after three local committee meetings may be forwarded to the regional committee. Urgent issues may be referred to the regional committee immediately by mutual agreement between the parties.

5. Statutory Holidays

- 5.1 The parties agree that there exists an established practice in which all Public Health Nurses receive the same statutory holidays as all other nurses in the bargaining unit. These holidays are outlined in Clause 18.01(a) of the NLNU Collective Agreement.
- 5.2 Subject to Clause 18.01(a)(h), the two additional statutory holidays to be observed shall be:
 - Thanksgiving Day
 - Armistice Day

6. Transfer of Services Within the Western Regional Health Authority

- 6.1 For the purpose of this Article, a service transfer is the transfer of a service from one physical location to another within Western Regional Health Authority. Service Transfer does not include a change in the management reporting structure or a location change within the same facility or site operated by the Employer.
- 6.2 Western Regional Health Authority reserves the right to transfer staff as a result of service transfer. Employees who are transferred shall retain seniority, service, and other recognized earned and portable benefits.
- 6.3 Employees who are affected by service transfer shall have access to lay-off or displacement options if:
 - 6.3.1 There is an associated downsizing of positions within the service, or
 - 6.3.2 The service is being transferred to a worksite located 80 kms (driving distance) or greater from the employee's current worksite.
- 6.4 Where there is an associated downsizing of positions within the service or the service is being transferred to a worksite located 80 kms (driving distance) or greater from the employee's current worksite, the Employer will offer nurses, on the basis of seniority, the choice of transferring within the service or accessing lay-off / displacement. If there are not enough nurses interested in transferring with a service the Employer reserves the right to temporarily transfer nurses commencing with the most junior nurse for a reasonable period of time while the employer is training replacement staff.

Employees considering accepting layoff should fully explore the Employment Insurance implications before exercising this option.

- 6.5 Western Health will provide a familiarization period up to but not exceeding two weeks, to the employees transferring with their Unit/Division to a new site. This will assist them in becoming acquainted with essential information such as policies and procedures, routines, location of supplies and equipment, fire and disaster plans, etc.
- 6.6 Subject to the bumping process, and subject to 6.3.2 above, temporary employees replacing permanent employees in a Unit/Division that is transferred will transfer with the Unit/Division if the permanent employee they are replacing is transferring with the Unit/Division.
- 6.7 Subject to the bumping process, employees on layoff and employees off work on various forms of leave recognized in the collective agreement shall have the same options available to them in the event of service transfer as though they were working.

DATED THIS 31 ST DAY OF MA	, 2010
Debbie Jonuard. NEWFOUNDLAND AND LABRADOR NURSES UNION	WITNESS DAVID CONNAY
Susa Sule WESTERN REGIONAL HEALTH AUTHORITY	WITNESS
Elizabeth Lane NEWFOUNDLAND AND LABRADOR HEALTH BOARDS ASSOCIATION	WITNESS
TREASURY BOARD	WITNESS

SCHEDULE "A"

LABOUR RELATIONS BOARD ORDER # 5062 JANUARY 17, 2007



Labour Relations Board

5062 IN THE MATTER OF

Public Service Collective Bargaining Act

- and -

Western Regional Integrated Health Authority

Applicant

- and -

Newfoundland and Labrador Nurses' Union

Respondent

WHEREAS pursuant to Section 44 of the Public Service Collective Bargaining Act, the Labour Relations Board has received an application from Western Regional Integrated Health Authority for the resolution of a question or problem that has arisen as a result of the sale, lease, transfer, disposition of a business or the operations of the business or a part of either;

AND WHEREAS the Applicant and Respondent agree that there has been a sale, lease, transfer or other disposition of a business or the operations of the business or a part of either as envisaged under Section 44 of the Act;

AND WHEREAS following investigation and consideration of the representations of the interested parties, the Board has decided that there are questions or problems which have arisen with respect to the consolidation of a number of health institutions under the Western Regional Integrated Health Authority and these questions or problems require resolution;

AND WHEREAS Western Regional Integrated Health Authority is now the Employer in the place of Western Health Care Corporation and Western Regional Health and Community Services Board;

NOW THEREFORE it is hereby ordered by the Labour Relations Board that:

- Western Regional Integrated Health Authority is now the Employer in the place of Western Health Care Corporation and Western Regional Health and Community Services Board;
- Western Regional Integrated Health Authority is now the Employer and is a party to or is bound by the Collective Agreement with Newfoundland and Labrador Nurses' Union which is as follows:
 - (a) a Collective Agreement between Newfoundland and Labrador Nurses' Union and Her Majesty the Queen in Right of Newfoundland (represented by The Treasury Board) and The Newfoundland and Labrador Health Boards Association on behalf of Employers listed in the Collective Agreement, including Western Health Care Corporation and Western Regional Health and Community Services Board, and including a Transition Agreement dated 2 June 1997, which Collective Agreement was in effect from 22 July 2002 to 30 June 2005 (Nurses' Collective Agreement);

- 3. there be established a single Nurses' bargaining unit in the classifications set out in the Nurses' Collective Agreement;
- 4. the Newfoundland and Labrador Nurses' Union is hereby declared to be the sole and exclusive bargaining agent for employees of Western Regional Integrated Health Authority in the Nurses' bargaining unit;
- 5. the seniority list in reference to the above-noted Collective Agreement be compiled such that each Nurse is credited with his or her best seniority number so that each Nurse's name appears once only on the seniority list;
- 6. the Collective Agreement currently in effect will remain in effect during the transition period or until a new Collective Agreement is signed between Western Regional Integrated Health Authority and Newfoundland and Labrador Nurses' Union; and
- 7. the Board retains jurisdiction to deal with any further questions or problems which may arise with respect to this application and to provide for resolution of same.

THE official seal of the Board was hereunto affixed and attested to by the Chief Executive Officer of the Board at the City of St. John's in the Province of Newfoundland and

Labrador this 17th day of January 2007

TRANSITION AGREEMENT

BETWEEN

HER MAJESTY THE QUEEN IN RIGHT OF NEWFOUNDLAND (Represented herein by the Treasury Board)

AND

THE NEWFOUNDLAND AND LABRADOR HEALTH BOARDS ASSOCIATION

REPRESENTING

LABRADOR-GRENFELL REGIONAL HEALTH AUTHORITY

AND

THE NEWFOUNDLAND AND LABRADOR NURSES' UNION

This agreement made this ________ day of _________ two thousand and ten in accordance with Clause 40.02 of the Collective Agreement between Her Majesty the Queen in Right of Newfoundland and Labrador (represented herein by the Treasury Board), the Newfoundland and Labrador Health Boards Association, and the Newfoundland and Labrador Nurses' Union (NLNU), herein referred to as the Collective Agreement. This Agreement governs all members of the NLNU who are employees of the Labrador-Grenfell Regional Health Authority.

It is further recognized that effective the date this Transition Agreement comes into effect, the transition agreements with each of the Employer's legacy organizations, as outlined below will be repealed and replaced with this Transition Agreement:

- Transition Agreement between Treasury Board, the Newfoundland and Labrador Health Care Association on behalf of the Health Labrador Corporation and the Newfoundland and Labrador Nurses Union dated February 11th, 1998, and
- Transition Agreement between Treasury Board, the Newfoundland and Labrador Health Care Association on behalf of the Grenfell Regional Health Services Board and the Newfoundland and Labrador Nurses Union dated March 19th, 1997.

1. Transfer of Business - Successor Rights

1.1 In accordance with Section 44 of the Public Service Collective Bargaining Act the parties to this agreement agree that there has been a "transfer of business" from the following health care employers:

Health Labrador Corporation

AND

Grenfell Regional Health Services Board

TO

Labrador-Grenfell Regional Health Authority

1.2 As per the order of the Labour Relations Board (LRB) dated July 16th, 2008 (the LRB Order), a copy of which is attached as Schedule "A", it was determined that there be established a single regional nurses (NS) bargaining unit in the classifications covered by the NLNU Collective Agreement, within the Labrador-Grenfell Regional Health Authority.

2. No Loss Of Seniority For The New NLNU Bargaining Unit

NLNU members shall not lose their seniority as a result of the transfer of business as outlined in Clause 1 of this agreement.

- 2.1 The Employer will post a new single seniority list in all work areas prior to June 1st, 2010. Subject to protests and grievances the new single seniority list will be effective May 11th, 2010. Should a new list not be posted prior to that date the Employer will calculate seniority manually for any job competitions in which seniority would be the deciding factor.
- 2.2 Employees will have two (2) months from the date the seniority list is posted to submit a protest in writing to the Human Resources department outlining the specific reason why they feel their seniority to be incorrect. The Human Resources Department will have four (4) months from when the seniority protest was received to provide the employee with a written response outlining the outcome of their seniority protest and the detail of what adjustments, if any, were required. Should the employee dispute the employer's decision the employee may file a grievance and time limits for the filing of such a grievance will not commence until the employee receives the employer's written response to their seniority protest. A copy of the response will be copied to the employee's personal file.
- 2.3 A revised seniority list incorporating corrections will be posted by January 31st, 2011.
- Once the initial protest period has elapsed, all future protests will be completed in accordance with Article 24.03 of the Collective agreement. Subject to the mutual agreement to the contrary, where displacements or staff changes have occurred, it will not be altered due to the discovery of errors not brought to the attention of the employer during the protest period.
- 2.5 If the situation should develop whereby two (2) or more employees have the same seniority, and it becomes necessary to determine who is senior for the purposes of a job competition or layoff/displacement, the seniority shall be broken by drawing names from a hat. A union representative, together with the individuals whose names are in the hat, or designates, may be present with management when the draw takes place.

3. Transfer of Service - Effective Date

As of April 1, 2005 the employees of the two (2) former employers outlined in Clause 1 of this agreement became employees of the Labrador-Grenfell Regional Health Authority. Labrador-Grenfell Regional Health Authority recognizes the service of each of the NLNU members as it was recognized by the previous employers prior to that date. Service for seniority purposes is to be recognized on a single, regional bargaining unit basis.

4. Professional Practice Committees

One local Professional Practices Committee will be established for each of the following areas:

Labrador City/Wabush Happy Valley-Goose Bay St. Anthony Health Centres (Forteau, Roddickton, Flowers Cove) Community Clinics

It is agreed that the parties will review the committee structure after one year at which time either party may give notice to revert to the Collective Agreement

The parties further agree to the establishment of one regional professional practice committee which will meet at least twice per calendar year. At least one of these meetings shall be face to face. Additional meetings may occur as required.

The regional committee shall be comprised of representatives from the Union and the Employer and may include a Union representative and an Employer Representative from each of the local committees.

The regional committee will address issues that are regional in nature. Items from the local committees unresolved after three local committee meetings may be forwarded to the regional committee. Urgent issues may be referred to the regional committee immediately by mutual agreement between the parties.

5. Transfer of Services Within the Labrador-Grenfell Regional Health Authority

- 5.1 For the purpose of this Article, a service transfer is the transfer of a service from one physical location to another within Labrador-Grenfell Regional Health Authority.
- 5.2 Labrador-Grenfell Regional Health Authority reserves the right to transfer staff as a result of service transfer. Employees who are transferred shall retain seniority, service, and other recognized earned and portable benefits.
- 5.3 Employees who are affected by a service transfer shall have access to lay-off or displacement options if:
 - 5.3.1 There is an associated downsizing of positions within the service, or
 - 5.3.2 The service is being transferred to a worksite located 80 km (driving distance) or greater from the employee's current worksite.
 - 5.3.3 The service is being transferred to a worksite that, while located less than 80 km (driving distance) from the employee's current worksite, requires travel by air or ferry.

- Where there is an associated downsizing of positions within the service or the service is being transferred to a worksite located 80 km (driving distance) or greater from the employee's current worksite, or the service is being transferred to a worksite that, while located less than 80 km (driving distance) from the employee's current worksite, requires travel by air or ferry, the Employer will offer nurses, on the basis of seniority, the choice of transferring with the service or accessing lay-off / displacement.
- 5.5 If there is a service transfer in circumstances where employees do not have access to lay/off or displacement options and there are not enough nurses interested in transferring with a service, the Employer reserves the right to transfer nurses commencing with the most junior nurse first.
- 5.6 Employees considering accepting layoff should fully explore the Employment Insurance implications before exercising this option.
- 5.7 Labrador-Grenfell Regional Health Authority will provide a familiarization period up to but not exceeding two weeks, to the employees transferring with their Unit/Division to a new site. This will assist them in becoming acquainted with essential information such as policies and procedures, routines, location of supplies and equipment, fire and disaster plans, etc.
- 5.8 Subject to the bumping process, and subject to 5.3 above, temporary employees replacing permanent employees in a Unit/Division that is transferred will transfer with the Unit/Division if the permanent employee they are replacing is transferred with the Unit/Division.
- 5.9 Subject to the bumping process employees on layoff and employees off work on various forms of leave recognized in the collective agreement shall have the same options available to them in the event of service transfer as though they were working.

6. Grievance Procedure

Step 1

Grievance submitted to her/his immediate non-unionized supervisor.

Labrador-Grenfell Health will supply to the Union a list of designates for each step (above Step 1) of the grievance procedure. This list will be updated as required.

7. Posting of Temporary Positions

Temporary vacancies which exceed or are expected to exceed sixteen (16) weeks shall be posted and filled in accordance with the collective agreement.

SCHEDULE "A"

LABOUR RELATIONS BOARD ORDER JULY 16, 2008



Labour Relations Board

5167

IN THE MATTER OF

Public Service Collective Bargaining Act

- and -

Labrador-Grenfell Regional Health Authority

Applicant

- and -

Newfoundland and Labrador Nurses' Union

Respondent

WHEREAS pursuant to Sections 44, 44.1 and 45 of the Public Service Collective Bargaining Act, the Labour Relations Board has received an application from Labrador-Grenfell Regional Health Authority for orders consolidating existing classifications of Nurses of the Employer who are members of separate bargaining units represented by the Newfoundland and Labrador Nurses' Union into a single Nurses (NS) bargaining unit of employees of the Employer, with dovetailing of seniority lists and consequential relief;

AND WHEREAS the Board, following consideration of the application and the representations of the interested parties; and noting that the Respondent consents to the orders requested in the application, has decided to grant the application;

NOW THEREFORE it is hereby ordered by the Labour Relations Board that:

- Labrador-Grenfell Regional Health Authority is now the Employer and is a party to or
 is bound by a Collective Agreement in effect from 26 January 2007 to 30 June 2008
 made among Newfoundland and Labrador Nurses' Union, Her Majesty the Queen in
 Right of Newfoundland and Labrador, represented by Treasury Board and
 Newfoundland and Labrador Health Boards Association; and including a Transition
 Agreement dated 19 March 1997 for Grenfell Regional Health Services Board and a
 Transition Agreement dated 11 February 1998 for Health Labrador Corporation (the
 NLNU Collective Agreement);
- and that there be established a single Nurses (NS) bargaining unit in the classifications covered by the NLNU Collective Agreement;
- and that the seniority lists for the existing two (2) Nurses (NS) bargaining units be dovetailed into a single seniority list for the Nurses (NS) bargaining unit;
- and that Newfoundland and Labrador Nurses' Union continues to represent the members of the Nurses (NS) bargaining unit;

5th Floor, Beothuck Bidg., 20 Crosble Place, P. O. Box 8700, St. John's, NL, Canada, A1B 4J8 Telephone: (709) 729-2707 Facsimile: (709) 729-5738 www.gov.nl.calirb

- and that the Collective Agreement currently in effect will remain in effect during the transition period or until a new Nurses (NS) Collective Agreement is signed between Labrador-Grenfell Regional Health Authority and Newfoundland and Labrador Nurses' Union;
- and that the Labour Relations Board retains jurisdiction to deal with any further
 questions or problems which may arise with respect to this application and to provide
 for the resolution of same.

THE official seal of the Board was hereunto affixed and attested to by the Chief Executive Officer of the Board at the City of St. John's in the Province of Newfoundland and Labrador this 16th day of July 2008

Chief Executive Officer

DATED THIS DAY OF _August	, 2010
Debbe Toward. NEWFOUNDLAND AND LABRADOR NURSES UNION	WITNESS WITNESS
LABRADOR GRENFELL REGIONAL HEALTH AUTHORITY	WITNESS
TREASURY BOARD	WITNESS
PLIZABETH LAND NEWFOUNDLAND AND LABRADOR HEALTH BOARDS ASSOCIATION	WITNESS

*SCHEDULE M

1. Alternative Scheduling

It is recognized that the quality of nurses' work and personal lives are affected by their requirements to work a rotational schedule. Both parties recognize the benefits of alternative scheduling models and will endeavor to explore options with respect to innovative scheduling during the term of the collective agreement.

Schedules other than those outlined in the collective agreement may be developed in order to improve quality of working life, support continuity of patient care, ensure adequate staffing resources, and support cost-efficiency. The parties agree that such innovative schedules may be developed locally by the Employer in consultation with the Union and subject to the following principles:

- (a) such schedules shall be established by mutual agreement of the Employer and the Union
- (b) the introduction of such schedules shall be determined by the Employer and the Local union representatives.
- (c) the parties may agree to amend provisions of the collective agreement to accommodate any innovative unit schedules.
- (d) the request to try an alternative schedule must have the support of at least seventy five (75) percent of the employees in the work area whose schedules are affected by the alternative schedule.
- (e) such schedules may be discontinued by either party giving sixty (60) days written notice.
- (f) only employees who are impacted by the alternative schedule and hold permanent positions in the work area, including those permanent employees on leaves of absence, will be eligible to vote. Casual employees, employees holding temporary positions in the work area and employees occupying permanent float positions shall not be eligible to vote.
- (g) the trial period for an alternative schedule shall not be less than twelve (12) weeks.

Jobs sharing, job sharing and/or part-time work for a full weekend and every second weekend may be considered.

The complementary weekday worker continues to work 75 hours bi-weekly but no weekend work is scheduled. All shifts are scheduled between Monday and Friday.

1. Four (4) Shift Twelve (12) Hour Schedule

- Requires a six (6) week rotational schedule
- Requires mutual agreement between the Employer and the employee
- Normally works two (2) days and two (2) nights consecutively subject to mutual agreement to the contrary

- Self-scheduling is optional with the agreement of the parties
- Normally works a maximum of 20 shifts in 42 calendar days
- Union waives 8.05 "Consecutive Shifts" and 8.09 "Weekends Off" for the employees working a four (4) shift twelve (12) hour schedule.

2. Weekend Nurse

The Employer proposes piloting this schedule at Eastern Health. Location to be determined.

- 1. The union will agree to exceptions to the definition of Article 2.01(y) "Temporary Employee" to permit the hiring of temporary weekend nurses for the duration of the pilot(s).
- 2. The following parameters shall apply:
 - (a) The weekend nurse positions shall be full time nurse positions and will be normally integrated into core staffing. The union would not be opposed to an Employer introducing the concept other than as part of core staffing.
 - (b) The rotation of a weekend nurse shall normally be comprised of two (2) shifts of 11.25 hours and one (1) shift of 7.5 hours per week. However an option to scheduling and rotations may include a six (6) week cycle of two (2) weekends of three (3) shifts of 11.25 hours and one (1) weekend of two (2) shifts of 11.25 hours. In any rotation the combination of weekend hours worked will provide an average of thirty (30) hours worked (excluding meal breaks) each weekend.
 - (c) Weekend nurses shall be scheduled to work eighty (80) percent of the hours of a full-time nurse and be paid one hundred (100) percent pay of a full-time nurse.
 - (d) Weekend nurses shall be entitled to switch shifts with other nurses including other weekend nurses to a maximum of two shifts per employee in a six week rotation.
 - (e) Weekend nurses will earn the same benefits as a full time nurse and pension contributions and group insurance premiums will be the same as those of a full time nurse.
 - (f) Although weekend nurses will accumulate leave as a full time employee, they will be deducted leave at an accelerated rate if leave is taken while working a weekend schedule. For example, a nurse who requests two (2) weeks annual leave will be deducted seventy-five (75) hours, the same amount they would have been paid to work a sixty (60) hour schedule; or if a nurse is sick on a twelve (12) hour shift, they will be deducted the eleven point two five (11.25) hours at an accelerated rate of 1.25 times equaling 14.06 hours.

- (g) Statutory holidays for weekend nurses shall be administered in the same manner as those for part time nurses including being able to bank their stats.
- (h) Compensation for extra shifts or overtime shall be on the same basis as for full time employees, after the weekend worker has worked 75 hours, including shift differential and weekend premiums where applicable.
- (i) Any remaining shifts may be filled at the discretion of the Employer.

2. Provincial Nursing Self-Scheduling Guiding Principles

The parties agree to review the self-scheduling guiding principles on an annual basis and update as required.

The Employer and unit/facility staff may enter into a self-scheduling arrangement. The following guiding principles apply:

- 1. Fifty (50) per cent plus one of permanent and temporary staff must agree to a self-scheduling model.
- 2. Self-schedules must comply with relevant collective agreements, however, exceptions may be established in unit specific guidelines acknowledging that overtime premiums are not applicable (e.g. employees may self-schedule greater than seven (7) consecutive eight (8) hour shifts or greater than three (3) consecutive twelve (12) hour shifts).
- 3. Existing employee accommodations will be honoured; new requests for accommodations will be assessed on an individual basis.
- 4. Unit-specific self-scheduling committees will be established, comprised of frontline staff members. The unit manager (or designate) will provide oversight to the committee and engage with staff, as indicated, to resolve issues that may arise, and approve the final work schedule.
- 5. The duration of work schedules will be determined by staff majority and operational requirements (i.e. six (6), eight (8), or twelve (12) week work schedules).
- 6. Staff members may choose to work outside of a regular rotation and not always follow a master rotation A or B team; however, all staff must work an equal amount of weekend shifts per work schedule, as per unit specific guidelines. Staff may choose to work additional weekend shifts per work schedule.
 - *Part-time employee weekend shifts are pro-rated as per letter of employment unless otherwise agreed.

- 7. Staffing levels per shift are based on patient/resident/client acuity, applicable standards of nursing care, and skill mix. Units will develop a process to ensure the rotation of shifts (i.e. day/nights) is equitable.
- 8. Staff may develop unit specific guidelines to allocate shifts on statutory holidays, including Christmas and New Year's period, and a method for selecting annual vacation dates. Otherwise, relevant Collective Agreement provisions will apply.
- 9. Units will determine the number of special scheduling requests per employee, per work schedule, and implement a process to equitably allocate priority for special scheduling requests among the staff. Staff submitting more than one request per work schedule will indicate the priority of requests. Unit staff will develop guidelines for special scheduling request approvals when multiple requests for the same shift are submitted.
 - *Note: Special scheduling request examples include preference for day of rest and preference for shift rotation.
- 10. The "blank" work schedule will be available in the format utilized by the clinical area (hard copy; electronic) for a minimum of two weeks to allow everyone to fill in requests. Unit staff are to determine timelines and rotation for staff to complete self-schedule per each timesheet.
- 11. The final approved work schedule will be posted two weeks prior to start of the work schedule term.
- 12. Any unfilled shifts thereafter will be filled as per the applicable collective agreement language.
- 13. Prior to approval and posting of the work schedule, staff may be requested to make changes to self-scheduled shifts to ensure the unit's operational staffing requirements are met.

Responsibilities of Self Scheduling Committee

- Develop and periodically review fair and equitable self-scheduling guidelines and processes with staff; provide orientation to new staff.
- Ensure staffing numbers meet core staffing requirements; develop a process to staff unfilled shifts as per the Collective Agreement prior to submitting the draft schedule to the manager.
- Act as a resource to staff and to assist with interpretation of the guidelines.
- Ensure adherence to timelines for submission of requests and posting the approved work schedule.
- Develop a process to resolve scheduling conflicts between employees including managing changes and escalating issues to manager for resolution.
- Keep manager or designate informed of any issues that arise during the self-scheduling process.
- Ensure the work schedule is posted within two weeks in advance of start date.
- *All staff members will have an opportunity to serve on the self-scheduling committee on a rotational basis.

Responsibilities of Staff Member:

- Submit schedule according to the posted deadline.
- Schedule shifts according to the guidelines, i.e. indicate specific scheduling preferences, special requests.
- Follow unit-specific scheduling guidelines when indicating shifts where you prefer not to be changed.
- Ensure the number of scheduled hours per work schedule are correct as per letter of employment.
- Assist the scheduling committee in problem solving if required.

Responsibilities of Unit Manager:

- Assist in scheduling problem solving as required.
- Respond to leave requests.
- Review and approve the completed schedule prior to posting.

*SCHEDULE N

LAYOFF AND RECALL

This agreement replaces all other agreements regarding layoff and bumping

BUMPING:

1) Application

An Employee shall be entitled to access the bumping provisions outlined in this Agreement when:

- 1. A layoff notice has been issued in accordance with Article 23.02 and/or
- 2. There has been a permanent transfer of positions within the same Employer in the department/program to which the employee is permanently assigned, and no vacant permanent position exists in the same department/program within 80 kilometers of where the restructuring occurred excluding positions within float pools. Employees may elect to waive the 80 kilometer restriction; and/or
- 3. There has been a permanent reduction or elimination of positions in the unit/division to which the employee is permanently assigned and no vacant position in the same department/program exists within the employee's site excluding float pools and no vacant position exists in the same department/program within 80 kilometers of the employee's site. Employees may elect to waive the 80 kilometer restriction; and/or
- 4. There has been a temporary reduction in the work available in the unit/division to which the employee is permanently assigned necessitating the displacement of the employee to other units/divisions and that reduction exceeds or is expected to exceed nine (9) months; and/or
- 5. The Employer and the Union agree that they may mutually agree to the issuance of layoff notices in circumstances other than those described above.

2) General Principles

In the event of bumping the following general principles shall apply:

- a) An employee has the right to have a shop steward or other union representative in attendance when their options are explained.
- b) An employee who occupies either of the three classifications of Nurse I, Nurse I Long Term Care or Nurse I Specialist shall be permitted to bump into or assume a position in either of the three classifications.
- c) Except as otherwise noted in these principles, an employee shall be permitted to bump into or assume a position at their own or lower salary (classification level).
- d) For the purposes of Article 23, Psychiatric Nurse Is (N.S.-29), and Nurse Is (N.S.-28) shall be considered at the same classification level for the purposes of this Agreement.

- e) An employee working in a rotational position may bump the most junior employee in a rotational position, as per Clause 4 below or may select any vacant rotational position, as per Clause 5 below.
- f) An employee working in a Monday through Friday non-rotational day position may bump the most junior employee in any position, rotational or non-rotational as per Clause 4 below or may select any vacant position as per Clause 5 below.
- g) A employee working in a permanent full time or permanent part-time position shall be entitled to maintain their hours of work but may, at the employee's option, bump into or assume a position that involves a reduction of hours or an increase in hours including up to full time hours.
- h) An employee bumping into a position, being recalled or assuming a vacancy for which they requires orientation will be provided orientation in line with Clause 27.02 of the Collective Agreement.
- i) Where the employee does not have the option to accept a permanent vacancy or does not have a permanent bumping option at their classification level, they may choose a temporary option pursuant to Clause 4(c). Provided no equivalent options at their classification level are available, permanent employees who accept a temporary option shall again execute their rights under this agreement when the temporary position ends.
- j) Where float pools exist, an employee may choose not to bump into the float pool.
- k) A permanent employee who has no available bump or vacancy at their own classification level within their geographic region, and who accepts a vacant position, or who bumps into a position at a lower classification level shall have their salary red circled.
- k) In all instances, employees bumping into positions or accepting vacancies must be qualified, able and fit for the position.

3) Notification

As per Article 23.05, RNUNL Provincial Office and the Branch shall be given a minimum of forty eight (48) hours' notice of any circumstances requiring the use of this Agreement prior to the issuance of any individual notices.

Individual notices will be given to affected employees in the circumstances outlined in Clause 1 (Application).

4) **Bumping Options**

It is agreed that each Employer will provide a comprehensive list of all applicable unit/divisions and department/programs for each respective Employer within three (3) months of the date of signing of this agreement to be included as Schedule A to this agreement. It is understood by the parties that as required, organizational changes may necessitate a change in this list and the Employer agrees to update the Union in writing of these changes as they occur.

- a) For the purposes of this article the following definitions apply:
 - i) "Unit/Division" means the nursing unit or immediate work area to which the employee is assigned.
 - ii) "Department/Program" means the work/clinical area to which the employee is assigned.
 - iii) "Site" means the physical location where the employee works. For employees in Regional Positions "site" means any of the sites at which the employee regularly works.
 - iv) "Geographic Region" means all work sites within a 80 kms driving distance from the employee's current work site as outlined in Appendix "A".
- b) A permanent employee who receives notice of layoff or notice that their position has been identified for a bump, shall have the following rights:
 - i) Accept the layoff (where applicable).
 - ii) Bump the most junior permanent employee in either the same or lower bargaining unit classification at either of the following:
 - 1. Any unit/division within their site or
 - 2. Any unit/division within their department/program within their geographic region
 - iii) If a displacing employee at the NS-28 or NS-29 classification level has no permanent option within a 80 km driving distance of their current work site they shall be permitted to access positions at the Community Health Nurse I (NS-30) and Public Health Nurse I (NS-30) classification levels. When it is apparent that an employee(s) will have to displace up pursuant to this provision, the opportunity to do so shall first be offered to the senior employees impacted by the layoff/displacement, subject always to the proviso that the displacing employee(s) must be qualified, able and fit for the position.
 - iv) If no permanent option exists under Clauses 4(b)(ii) (1) and (2), bump the most junior permanent employee in any site within the organization
- c) i) If no permanent options exist as outlined in Clause b) or if a permanent employee has no permanent option at their own classification level, the permanent employee may choose to bump any junior temporary employee in either the same or lower bargaining unit classification with a remaining work assignment greater than sixteen (16) weeks, at either of the following:
 - 1. within their site
 - 2. within their unit/division
 - 3. within their department/program
 - 4. in their department/program within their geographic region
 - 5. within their geographic region
 - 6. in their department/program within the organization
 - 7. within the organization.

The sixteen (16) week requirement may be waived by mutual agreement between the Employer and the Union where there is minimal orientation required.

ii) While the permanent employee is working in the temporary position their permanent status will be maintained.

5) Options

In consultation with the Branch, the Employer will meet with the affected employee and provide them with a list of vacant positions as well as an outline of their bumping options. The employee shall communicate their preference within three (3) days (excluding weekends and statutory holidays) of the meeting. No employee shall have less than three (3) days (excluding weekends and statutory holidays) of their individual meeting in which to communicate their preference to the Employer. Employees who have no permanent or temporary options shall be given an additional fourteen (14) calendar days notice of layoff following the bumping process. An employee shall not be compelled to accept a vacant position where they have bumping options.

6) Date to Determine Seniority

The date to determine seniority for employees whose seniority is calculated in hours shall be the last pay day before the Employer gives Notice to the Union under Clause 3. If a centralized payroll is not in place, the seniority list will be adjusted to the same end date for all employees.

7) Equal Seniority

If a situation should develop whereby two (2) or more employees have the same seniority, the seniority shall be broken by random draw. A union representative, together with the individuals whose names are in the draw, or their designate, are entitled to be present with management when the draw takes place.

8) Loss of Seniority

Under Clause 23.07, an employee would lose their seniority if they are laid off for a period of two (2) years or more. The two-year period will recommence each time such an employee works a shift, regardless of whether they work that shift as a permanent, casual or temporary employee.

9) Employees Reverting Back to Previous Positions

Provided the employee is not on lay-off, if the position of the bumped employee becomes vacant within the next two (2) years of them being bumped from their position, they will be given the right to revert back to their previous position.

The Union agrees to waive posting requirements for such realignments. The two (2) year period shall run from the original date of bumping and shall only apply to the initial position the employee is bumped from and shall not be extended by reason of casual employment or recalls of any nature.

10) Recall

Upon layoff, permanent employees shall have recall rights to permanent and temporary positions for a period of two (2) years from their original date of layoff unless otherwise agreed between the Employer and Union.

Permanent employees shall have recall rights to permanent positions and temporary positions of greater than sixteen (16) weeks anticipated duration provided the position is a position that the employee would have been able to bump into had it been available at the time of layoff but no employee shall be obligated to accept recall to:

- (i) A site other than the site at which they worked prior to layoff
- (ii) A position requiring more or less hours of work than the position at which they worked prior to layoff
- (iii) A position at a lower classification level

11) Casual Employment for Permanent Employees on Layoff

Permanent employees, who are being laid off, should advise the Employer in writing of their desire to do casual work.

12) Layoff and Recall of Employees on Leave of Absence and Workers Compensation

Such employees shall be laid off and recalled in the same manner as any other employee. If they are unavailable or unable to return to work because of Workers Compensation or leave of absence, they will be reinstated as employees for the purposes of any benefits that they may be entitled to under the collective agreement as of the date of recall and the next qualified permanent employee on lay-off or a temporary employee shall be hired in their place for the remainder of their leave of absence or Workers Compensation.

SCHEDULE O

This MOA, and contents within, refers to the Collective Agreement signed September 4, 2014

MEMORANDUM OF AGREEMENT made at St. John's, in the Province of Newfoundland and Labrador, this day of November. 2015.

BETWEEN

Her Majesty the Queen in Right of Newfoundland and Labrador (hereinafter referred to as the "Employer")

AND

The Registered Nurses Union of Newfoundland Labrador (hereinafter referred to as the "Union")

(hereinafter collectively referred to as the "Parties")

WHEREAS the parties entered into a Collective Agreement (the "Collective Agreement") effective September 4th, 2014 and expiring on June 30, 2016, which contains the terms and conditions of employment for all employees covered by the Collective Agreement; and

WHEREAS the parties recognize the previously exchanged correspondence dated January 17, 2014 signed by Geoff C. Williams, former Deputy Minister of the Human Resources Secretariat, addressed to John Vivian Executive Director of the Union (appended as Appendix A);

NOTHWITHSTANDING THE ABOVE and in accordance with Article 40.02 of the Collective Agreement, and to add further clarity to the January 17, 2014 letter referenced above, the parties agree to the following:

- For the purposes of the Collective Agreement, Registered Nurses employed in the classifications of Registered Nurse I, Registered Nurse IB and Registered Nurse IC shall be considered Level I.
- 2. For further clarity to paragraph 1, and as it relates to the movement of Registered Nurses between the classifications in Level I, it is understood by the parties that, in the event that a Registered Nurse obtains employment in a higher paying classification within the Level I nursing classifications, including any new classifications which may be added, he/she shall be placed on the higher paying salary scale in accordance with his/her salary in his/her current position, and in accordance with the JES Implementation Formula as outlined in the January 17th, 2014 letter on page 185 of the current Collective Agreement and as appended to this MOU in Appendix B.
- It is further understood by the parties that the reference to the movement of employees in paragraph 2 includes movement arising from the terms and

conditions of the Collective Agreement and includes temporary assignments, displacement, bumping, the movement of casual registered nurses between positions, and movement arising from positions obtained through the job competition process in the Collective Agreement

- 4. Additionally, it is understood by the parties that any future reclassifications within the Level I classifications that results in the reclassification of Registered Nurses to a higher paying Level 1 classification will result in the employee been placed on the higher paying salary scale in accordance with his/her salary prior to reclassification, and in accordance with the JES Implementation Formula.
- Movement to a higher paying Level 1 classification shall not require a Trial Period under Article 24.06 of the Collective Agreement.
- Any disputes arising from this agreement will be resolved in accordance with the grievance and arbitration process outlined in the Collective Agreement.

Signed on behalf of:

Her Majesty the Queen in Right of Newfoundland And Labrador, represented herein by Treasury Board

Registered Nurses Union of Newfoundland and Labrador

Appendix A

January 17, 2014

Mr. John Vivian Executive Director Newfoundland and Labrador Nurses Union P.O Box 416 St. John's, N L A1C 5J9

Dear Mr. Vivian:

Upon implementation of the new Job Evaluation System, effective July 15, 2015, Registered Nurses included in the Nurse I, Nurse I Long Term Care and Nurse I Specialist classifications shall have the following conditions apply;

- 1) For the purpose of Clause 25.04 (a) each of the three classifications, Nurse I, Nurse I Long Term Care and Nurse I Specialist, shall be considered Level I,
- 2) Schedule N 2 (a) shall have added, the following; "An employee who occupies either of the three classifications of Nurse I, Nurse I Long Term Care or Nurse I Specialist shall be permitted to bump into or assume a position in either of the three classifications".

This letter will remain in effect until June 30, 2016 or until a new collective agreement is negotiated.

Regards,

Geoff C Williams

Appendix B

January 17, 2014

Mr. John Vivian
Executive Director
Newfoundland and Labrador Nurses Union
P.O Box 416
St. John's, N L
A1C 5J9

Dear Mr. Vivian:

Re: Job Evaluation System (JES) Implementation

This will confirm employees will be implemented on to the Government Job Evaluation System effective July 15, 2015 as follows:

- If current salary falls below Step 1 of the new salary scale, employees will advance to step 1 on implementation; or
- If current salary falls between 2 steps of the new salary scale, employees will advance to the next highest step; or
- If current salary is above the maximum hourly rate on the new salary scale, employees will be red-circled at that rate.

Sincerely:

Elizabeth J Lane

Mr. John Vivian Executive Director Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Mr. Vivian:

Re: Annual Leave and Vacation

This is to confirm our understanding that the reference to annual leave and vacation in the Agreement all mean the same; the words are interchangeable for the purposes of earning and being awarded annual leave and vacation.

Sincerely,

Lisa Curran Chief Negotiator

From Curan

Mr. John Vivian Executive Director Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Mr. Vivian:

Re: Loss of Seniority

This will confirm our understanding that a person who holds a position outside the bargaining unit and who has retained seniority in accordance with clause 23.09, shall not be able to use their seniority to displace an employee of the bargaining unit.

Employees who have left the bargaining unit prior to or after the implementation of a seniority number system, and who return to the bargaining unit, will receive no greater or lesser benefit than that currently outlined in clause 23.09.

Sincerely,

Lisa Curran Chief Negotiator

Kin Curan

Mr. John Vivian Executive Director Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Mr. Vivian:

Re: Statutory Holiday and Weekends Off

This will confirm our intention on the above; where possible the Employer will endeavour to give an employee her day off in lieu of a statutory holiday on Monday following the employee's weekend off.

Sincerely,

Lisa Curran Chief Negotiator

From Curan

Mr. John Vivian Executive Director Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Mr. Vivian:

Re: Hiring Practices

This will confirm our understanding that the local branch of the RNUNL shall have the option of placing on the Professional Practices Committee's agenda, the practices of the Employer with regard to the use of and hiring of casual registered nurses. The committee shall make recommendations in the normal manner and the local branch shall be provided with detailed statistics on the utilization of casual registered nurses within thirty (30) days of their request.

Sincerely,

Lisa Curran

Chief Negotiator

Mr. John Vivian
Executive Director
Registered Nurses' Union NL
P.O. Box 416
St. John's, NL A1C 5J9

Dear Mr. Vivian:

Re: Organization of Nursing Work Studies

The Employer, Government and the Union are committed to implementing organization of work studies. The studies will arise out of and be specific to individual workplaces. The studies will consider alternative ways of organizing nursing work, respecting the input of nurses and the tasks they perform each day. The studies demonstrate support for the need to address registered nurses' workload and the provision of quality patient care.

Principles

- 1. Studies to be conducted in each of the following areas: tertiary care unit, rural centre nursing unit, community site and long term care unit.
- 2. Sites to be chosen by mutual agreement between the Employer and the union.
- 3. Studies to be conducted by "unit based working committees" consisting of an equal number of union and Employer representatives. Committees to be co-chaired by a unit representative and an Employer representative.
- 4. Cost of the studies to be the responsibility of the Employer to a maximum cost of \$50,000.00 in total for all four (4) studies inclusive of any paid time to be provided to employees who serve on the unit based working committees.
- 5. Provincial Union President or a member of the Provincial Board of Directors to be an ex officio member of each committee in a non-voting non decision making capacity.
- 6. Department of Health and Community Services to make available the services of a consultant to assist the unit based working committees with the development and implementation of an evaluation process. Evaluation criteria to be jointly agreed upon.
- 7. Duration of projects to be determined by unit based working committees. Suggested time lines include: (1) two (2) months to gather data on current work organization practices and to develop strategies for any change identified and (2) the impact of any agreed upon changes to be evaluated three (3) months after implementation. The four (4) studies to be

completed by the date of expiration of the collective agreement. All projects to be completed within the financial allocation identified in number four (4) above.

8. The unit based working committees will provide information to the union and the Employer for communication purposes.

Sincerely,

Lisa Curran

Chief Negotiator

Mr. John Vivian
Executive Director
Registered Nurses' Union NL
P.O. Box 416
St. John's, NL A1C 5J9

Dear Mr. Vivian:

Re: B.N. 2000

Effective April 1, 1999:

- 1. No employee will be laid off as a direct result of the standardization of the entry level qualifications (to practice) for new graduates to a Bachelor of Nursing to be introduced in the year 2000. The Employer shall retain the existing right to lay off in accordance with the collective agreement.
- 2. Should the Union allege that the Employer has acted in an arbitrary or discriminatory manner in adding a Bachelor of Nursing as a required or preferred qualification for any position not currently requiring a B.N., the Union shall have the right to file a grievance in accordance with the collective agreement. Once the grievance has been referred to arbitration either party shall have the option of referring the dispute to Grievance Mediation, in which case the mediation shall take place within six (6) months of the filing of the grievance.

Sincerely,

Lisa Curran Chief Negotiator

Mr. John Vivian
Executive Director
Registered Nurses' Union NL
P.O. Box 416
St. John's, NL A1C 5J9

Dear Mr. Vivian:

Re: Part-time Employees Working Additional Shifts

Effective October 1, 1999, (First notice to be posted September 1, 1999)

a) Each six months, on dates to be determined by the Employer and posted in all applicable workplaces a minimum one month in advance, part-time employees may advise the Employer in writing of their desire to work additional shifts. An employee may elect at any time to advise the Employer in writing that they will no longer be available to work additional shifts. In such a case, the Employer will not be obligated to offer additional shifts to the employee for the duration of the six months in which they served the notice.

Note: letter also applies to job sharing employees.

- b) Subject to clauses (a), (c), (d) and (e), casual work (extra shifts) known to be available at least one week in advance shall be first offered to part-time employees in the work area/unit to which they are assigned, up to equivalent full-time hours, before it is offered to casual employees. Part-time employees shall have the option of having their names added to a list for casual work (extra shifts) not known to be available one week in advance. Part-time employees will not be given priority for such work. Part-time employees shall continue to receive pro-rated benefits for all extra shifts. Part-time employees who accept an extra shift(s) will be required to work the additional shift(s) in accordance with the collective agreement. This clause will not apply in institutions providing a greater benefit to part-time employees in which case those greater benefits will continue for the life of the collective agreement.
- c) Part-time employees who have requested extra shifts under paragraph (a) above will not receive the premium rate outlined in clause 32.05 **Part-time Employees Additional Shifts**.
- d) The Employer is not obligated to offer extra shifts to part-time employees if this results in the Employer having to pay a contract premium rate, unless the Employer would incur a contract premium rate regardless of which employee worked the shift (eg. Statutory Holiday Pay).

e) Part-time employees who do not have a position on a specific unit will be given priority for extra shifts in any areas where they have received orientation and regularly work.

Sincerely,

Lisa Curran

Chief Negotiator

Lion Curan

Mr. John Vivian
Executive Director
Registered Nurses' Union NL
P.O. Box 416
St. John's, NL A1C 5J9

Dear Mr. Vivian:

Re: Gander and District Continuing Care Program and St. John's Home Care Program

This will confirm our understanding that the Agreements covering employees of these former two Employers will continue to remain in effect for these employees or until amended by Transition Agreements. The changes to the Provincial Registered Nurses' Agreement will apply to these Agreements and the employees covered by these Agreements.

For the purposes of clause 11.05 of the Provincial Nurses' Agreement, the "start time" for the employees of the former St. John's Home Care shall be 0800 hours.

Sincerely,

Lisa Curran Chief Negotiator

Mr. John Vivian
Executive Director
Registered Nurses' Union NL
P.O. Box 416
St. John's, NL A1C 5J9

Dear Mr. Vivian:

Re: Schools of Nursing

- 1. Employer agrees to "grandparent" the payment of Master's Educational Allowance to: Existing Nursing Instructors as of effective April 1, 1999, who are required to have a Master's Degree and have a Master's Degree in disciplines other than nursing, and
- 2. Existing Nursing Instructors who are enrolled and accepted into a Master's program as of effective April 1, 1999, in disciplines other than nursing will receive the Master's Educational Allowance upon receiving their Master's Degree and assuming a position that requires a Master's Degree with the Schools of Nursing.

Sincerely,

Lisa Curran Chief Negotiator

Mr. John Vivian
Executive Director
Registered Nurses' Union NL
P.O. Box 416
St. John's, NL A1C 5J9

Dear Mr. Vivian:

Re: Healthy Workplace Pilot

The parties agree that there may be a need to discuss initiatives necessary to the maintenance of healthy workplaces. To that end, the parties further agree to take a consultative interest based approach to the issue during the life of the collective agreement.

Sincerely,

Lisa Curran Chief Negotiator

Kin Curan

Mr. John Vivian Executive Director Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Mr. Vivian:

Re: Annual Paid Sick Leave – Article 18.01

All employees including casual employees hired prior to December 1, 2006 will be subject to the language in effect prior to December 1, 2006.

Sincerely,

Lisa Curran

Chief Negotiator

August 4, 2023

Ms. Tracey Trahey Chief Negotiator Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Ms. Trahey:

*Re: Enhanced Senior Level Collaboration between RNUNL, Government and Newfoundland and Labrador Health Services

- The parties agree to continue the tripartite committee titled Senior Joint Quality Work Life Committee.
- The Committee will address priority nursing practice and patient care issues affecting the provincial, system wide nursing work environment.
- Membership to be determined by the Terms of Reference.
- A Terms of Reference will be developed and updated appropriately.

Sincerely,

Fiona Ellis

Director of Collective Bargaining

Mr. John Vivian Executive Director Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Mr. Vivian:

Re: Market Adjustment

This will confirm our understanding reached during negotiations whereby if the Employer (Treasury Board Committee of Cabinet) determines that it is unable to recruit employees in specific positions at a particular geographic site, the Employer (Treasury Board Committee of Cabinet) may provide benefits to employees excluding those outlined in the collective agreement (for example, bonuses).

Sincerely,

Lisa Curran Chief Negotiator

Lion Curan

Mr. John Vivian Executive Director Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Mr. Vivian:

Re: Flexible Hours of Work Community Health

The parties agree that those employees currently covered under a flexible hours of work arrangement with their respective Employer, whether in terms of a private agreement, pilot project or under collective agreement language, will be grandparented under that current work arrangement. New employees hired after the date of signing of this collective agreement will be subject to the flexible hours of work.

Sincerely,

Lisa Curran

Chief Negotiator

Mr. John Vivian
Executive Director
Registered Nurses' Union NL
P.O. Box 416
St. John's, NL A1C 5J9

Dear Mr. Vivian:

Re: Nurse Uniforms

The parties agree that, with the exception of areas where OR scrubs are provided, the colour combination of white top and black bottom as chosen by the RNUNL will be reserved for nurse uniforms. In areas where the Employer provides uniforms to nurses, the Employer agrees to order the black and white colour combination in the event that future purchasing agreements are negotiated.

Notwithstanding the foregoing, the parties further agree that the colours of black and white are reserved for nurse uniforms and such is reflected in the Employers' global dress code policies. The Employer is not obligated or required to monitor, police or enforce compliance and/or non-compliance or to take any action in the event of non-compliance. The Employer agrees to encourage nurses to wear the agreed upon colours.

Additionally, this letter of understanding and/or the issues contained therein or arising from shall not be subject to the grievance and arbitration process under this collective agreement. Furthermore, this letter of understanding in no way obligates the Employer to continue to provide uniforms in areas where uniforms are currently provided to nurses.

Sincerely,

Lisa Curran Chief Negotiator

Mr. John Vivian Executive Director Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Mr. Vivian:

Re: Review of Relief Staffing

The parties agree to a one time review of relief staffing (permanent float, temporary and casual positions) to commence two years from the date of signing of the collective agreement. The parameters of the review are to be agreed upon by the parties. The review is to be completed within six (6) months of commencement.

Sincerely,

Lisa Curran Chief Negotiator

From Curan

Mr. John Vivian Executive Director Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Mr. Vivian:

Re: Sick Leave

The parties agree to the establishment of a provincial committee comprised of an equal number of Union and Employer representatives. The committee will explore sick leave utilization of the unionized workforce across the respective regional health authorities and make non-binding recommendations to RNUNL, the regional health authorities and the Government of Newfoundland and Labrador.

Sincerely,

Lisa Curran Chief Negotiator

Mr. John Vivian
Executive Director
Registered Nurses' Union NL
P.O. Box 416
St. John's, NL A1C 5J9

Dear Mr. Vivian:

Re: Workload Measurement

Provincial Workload Measurement Committee (Nurse Educators, Nurse Instructors)

TERMS OF REFERENCE

Mandate:

To research and develop a workload measurement system for Nurse Educators and Nursing Instructor I's and to make evidence-based recommendations to the Working Committee's at The Centre for Nursing Studies (CNS) and the Western Regional School of Nursing (WRSON) on the implementation of this system. These recommendations will also be distributed to both the Registered Nurses' Union of Newfoundland and Labrador (RNUNL) and the Employers (Western Health, Eastern Health).

Membership:

Chair - Representative from HRS

Director of Nursing - CNS

Director of Nursing - WRSON

Associate Director of Nursing - CNS

Associate Director of Nursing - WRSON

- 2 Nurse Educators (1 from CNS and 1 from WRSON)
- 2 Nursing Instructors (1 from CNS and 1 from WRSON)

Subject Matter Expert mutually agreeable by RNUNL, CNS and WRSON

the subject matter expert will act as an Advisor to the Committee

Meetings:

The committee shall meet once per month or at the call of the Chair.

Timelines:

The committee shall meet within 60 days of signing of the Collective agreement and provide recommendations to the Employer within six (6) months of the first meeting; this timeline may be adjusted by mutual agreement of the parties.

Decision Making:

Decision making shall be by consensus.

Minutes:

The Chair of the Committee will maintain a document outlining the actions and status of the Committee. This document will be provided to all Committee members.

Centre for Nursing Studies - Workload Measurement Working Group

TERMS OF REFERENCE

Mandate:

To assess and apply the workload measurement system to both Nurse Educators and Nursing Instructors as developed and recommended by the Provincial Workload Measurement Committee. This Working Group will also make recommendations to the Employer on consistent and reasonable work assignments across all programs.

Membership:

Director of Nursing

Associate Directors of Nursing

HR Business Partner

1 Nurse Educator

1 Nursing Instructor

1 Faculty Member from Continuing Education

RNUNL Representative

Chair – The Chair will alternate between the HR Business Partner and the RNUNL

Representative

Timelines:

The working group shall meet within 60 days of being provided the recommendations of the Provincial Committee; this timeline may be adjusted by mutual agreement of the parties.

Meetings:

The working group shall meet once per month or at the call of the Chair.

Decision Making:

Decision making shall be by consensus.

Minutes:

The Employer will provide an administrative support member to act as recording secretary. Minutes shall be reviewed by co-chairs prior to distribution to the working group for approval.

Western Regional School of Nursing - Workload Measurement Working Group

TERMS OF REFERENCE

Mandate:

To assess and apply the workload measurement system to both Nurse Educators and Nursing Instructors as developed and recommended by the Provincial Workload Measurement Committee. This Working Group will also make recommendations to the Employer on consistent and reasonable work assignments across all programs.

Membership:

Director of Nursing

Associate Director of Nursing

HR Manager

1 Nurse Educator

1 Nursing Instructor

RNUNL Representative

Chair - The Chair will alternate between the HR Manager and the RNUNL Representative

Timelines:

The working group shall meet within 60 days of being provided the recommendations of the Provincial Committee; this timeline may be adjusted by mutual agreement of the parties.

Meetings:

The working group shall meet once per month or at the call of the Chair.

<u>Decision Making</u>:

Decision making shall be by consensus.

Minutes:

The Employer will provide an administrative support member to act as recording secretary. Minutes shall be reviewed by co-chairs prior to distribution to the working group for approval.

Sincerely,

Lisa Curran

Chief Negotiator

LETTER OF UNDERSTANDING

Re: Other Post-Employment Benefits ("OPEB") Eligibility

The parties hereby confirm and acknowledge:

- 1. Former employees who are deferred pensioners within the meaning of the *Other Post-Employment Benefits Modification Act*, S.N.L. 2014 c.O-9 (the "Act") shall, as of the coming into force of the Act, only be entitled to OPEB in accordance with the Act.
- 2. Current employees as of the date of signing of the collective agreement who retire not later than December 31, 2019, with a minimum of five (5) years' pensionable service shall qualify for OPEB.
- 3. Current employees as of the date of signing of the collective agreement who retire after December 31, 2019, shall qualify for OPEB only where such employees are:
 - a. Pension eligible;
 - b. Have a minimum of ten (10) years' pensionable service; and
 - c. Retire and commence receipt of a pension immediately on ceasing active employment in the public service.
- 4. Employees who are hired subsequent to the date of signing of the collective agreement ("Newly Hired Employees"), shall qualify for OPEB only where such employees are:
 - a. Pension eligible;
 - b. Have a minimum of fifteen (15) years' pensionable service; and
 - c. Retire and commence receipt of a pension immediately on ceasing active employment in the public service.
- 5. Former employees who are rehired following loss of seniority subsequent to the date of signing of the collective agreement shall be considered to be Newly Hired Employees for the purpose of this Letter of Understanding.
- 6. Notwithstanding clause 5 above:
 - a. Employees with service prior to the date of signing of the collective agreement who are hired subsequent to the date of signing of the collective agreement who retain portability of benefits under the collective agreement; or
 - b. Employees with service prior to the date of signing of the Collective Agreement who are employed outside the bargaining unit in the public service and are reemployed in a RNUNL Public Service bargaining unit position subsequent to the date of signing of the Collective Agreement without a break in service in the Public Service shall not be considered to be Newly Hired Employees for the purpose of the this Letter of Understanding.

For the purposes of this clause the definition of RNUNL Public Service shall be limited to those Employers covered by the RNUNL collective agreement.

- 7. Employees who do not meet the criteria noted in clauses 2, 3 or 4 above shall not be entitled to OPEB on ceasing active employment in the public service.
- 8. Employees who become entitled to OPEB pursuant to clauses 2 or 3 above shall pay 50% of the premiums of the plan and the Employer shall pay fifty percent (50%).
- *9. Newly Hired Employees who become entitled to OPEB pursuant to clause 4 above shall pay premiums of the plan on the basis of their date of hire as follows:
 - a) employees hired up to March 31, 2020 shall pay 50% of the premiums of the plan and the Employer shall pay 50%; and
 - b) employees hired subsequent to March 31, 2020 shall pay 60% of the premiums of the plan and the Employer shall pay 40%.
- *10. Former employees who are rehired subsequent to March 31, 2020 and who become entitled to OPEB pursuant to clause 4 above shall be required to pay premiums in accordance with clause 9b) above.
- 11. This Letter of Understanding, made pursuant to s.3(2) of the Act, shall prevail where any term herein conflicts with a provision of the collective agreement, one of its Schedules, Letters or Memoranda of Agreement, including, without limitation, any practice, settlement of dispute, agreement or arbitration award arising from events prior to the coming into force of the Act.
- 12. Nothing in this Letter of Understanding shall have the effect of waiving or negating, in whole or in part, any requirement, procedural or substantive, under a Group Health and Life Insurance program or policy sponsored by the Employer, e.g., the filing of continuation or other required forms, provision of proof of insurability, etc....
- 13. This Letter of Understanding may be executed in any number of counterparts, each of which will be considered an original of same, and which together will constitute one and the same instrument. A facsimile signature or an otherwise electronically reproduced signature of any party shall be deemed to be an original.

July 30, 2019

Mr. John Vivian Executive Director Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Mr. Vivian:

Re: Annual Vacation and Holiday Provisions for Employees Employed by the Government of Newfoundland and Labrador

The parties agree that employees hired to work in Departments of the Government of Newfoundland and Labrador, after the effective date of signing of this agreement, shall be entitled to annual vacation in accordance with the following:

- (i) Less than one (1) year one and one quarter (1 1/4) working days for each month of service;
- (ii) One (1) year or more but less than ten (10) years three (3) weeks;
- (iii) More than ten (10) years but less than twenty-five (25) years of service -four (4) weeks;
- (iv) More than twenty five (25) years of service five (5) weeks.

The parties agree that employees hired to work in Departments of the Government of Newfoundland and Labrador, after the effective date of signing of this agreement, shall receive one (1) working day paid leave for each of the designated fourteen (14) holidays as follows:

- (a) New Year's Day
- (b) St. Patrick's Day
- (c) Good Friday
- (d) St. George's Day
- (e) Victoria (Commonwealth) Day
- (f) Discovery Day
- (g) Memorial Day

- (h) Orangeman's Day
- (i) Labour Day
- (j) Thanksgiving Day
- (k) Armistice (Remembrance) Day
- (l) Christmas Day
- (m) Boxing Day

One civic holiday, at a time to be determined by the Employer, in the area in which employees reside.

Sincerely,

Lisa Curran Chief Negotiator

Ms. Tracey Trahey Chief Negotiator Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Ms. Trahey:

*Re: Provincial Nursing Network

The Parties agree to the creation of a Provincial Nursing Network to include all regulated nurses, Employers, educators, professional regulatory bodies, unions and others identified as appropriate. The terms of reference are to be discussed. The mandate shall include, amongst other mutually agreed items, recruitment and retention issues.

Sincerely,

Fiona Ellis

Director of Collective Bargaining

Ms. Tracey Trahey Chief Negotiator Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Ms. Trahey:

*Re: Job Postings

The Parties agree to establish a working group with equal representation to discuss the standardization of job postings. This working group shall be established within six (6) months of signing of the Collective Agreement.

Sincerely,

Fiona Ellis

Director of Collective Bargaining

Ms. Tracey Trahey Chief Negotiator Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Ms. Trahey:

*Re: Severance Pay

Effective March 31, 2018, there shall be no further accumulation of service for severance pay purposes.

Employees who qualified for and elected to defer the payment of severance pay beyond March 31, 2018 cannot extend beyond the end of the period of employment and the redemption value shall be frozen as of March 31, 2018, for both accumulated service and weekly salary.

Severance pay that was deferred shall be paid upon an Employee's request or at the end of the period of employment.

Sincerely,

Fiona Ellis

Director of Collective Bargaining

Ms. Tracey Trahey Chief Negotiator Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Ms. Trahey:

*Re: Statutory Holiday Replacement

Each year, an employee shall be entitled to designate replacement statutory holiday(s) that are days of cultural or religious significance to the employee in place of any or all of the statutory holidays outlined in the collective agreement. The Employer will endeavor to accommodate such requests and such requests shall not be unreasonably denied.

For the purposes of this letter, cultural or religious significance shall be defined as: A day in which a religious observation is held or a day that celebrates the culture of a particular nation, people, or other social group.

The Employee shall inform the Employer of their choice(s), in writing, prior to November 15 in the calendar year before the new designations take effect. Such notice shall state clearly which statutory holiday(s) the employee is replacing and which day(s) of cultural or religious significance, including the dates on which they occur, that they are designating in the stead of the replaced statutory holiday(s).

Where the specific date(s) of cultural or religious significance are not yet confirmed on or before November 15 in the calendar year before the new designations take effect, the Employee will notify the Employer of the day(s) of significance and will provide date(s) as soon as they become available. The Employer will endeavor to accommodate such requests received after November 15 in the calendar year before the new designations take effect, subject to operational requirements and availability of replacement staff. Requests will not be unreasonably denied.

Such statutory holiday replacement, once designated, will not be amendable for the applicable calendar year. The Employer will grant the newly designated holiday(s) as paid day(s) off. Once designated per the above process, the newly named holiday(s) shall be the day(s) to which all rights which are normally associated with the specific statutory holiday being replaced are now applied:

- (a) The newly designated holiday days will attract all benefits of the collective agreement as if that day were the actual statutory holiday that they are designated to replace.
- (b) The replaced statutory holiday(s) will become a regular day, whether it be a workday or a day of rest, and will not attract any additional benefit previously

- attributable to it as a statutory holiday: all such benefits will have transferred to the designated replacement day(s).
- (c) Where the Employer does not provide service on a day described under b) above, and where the employee is scheduled to work on that day of the week, the Employer will make reasonable efforts to provide meaningful work to the employee on that day.
- (d) Where the Employer is unable to provide meaningful work on that day, the employee may access vacation, banked overtime, or other like paid banks to cover the missed day of work.

Sincerely,

Fiona Ellis

Director of Collective Bargaining

Ms. Tracey Trahey Chief Negotiator Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Ms. Trahey:

*Re: Other Post-Employment Benefits ("OPEB")

The parties recognize the mounting liabilities associated with the Other Post-Employment Benefits (OPEBs), and the resulting potential impact to the future sustainability of the Group Insurance Plan. Therefore, RNUNL agrees to engage in a joint collaborative process with the Employer and other interested stakeholders to meet within ninety (90) days of the signing of the Collective Agreement to address those challenges with the Employer and find solutions to ensure the long-term sustainability of the benefit plan.

Sincerely,

Fiona Ellis

Director of Collective Bargaining

Ms. Tracey Trahey Chief Negotiator Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Ms. Trahey:

*Re: Temporary Overtime Rate of Pay

Effective date of signing, and for a one-year period thereafter, a temporary overtime rate of pay will be implemented.

Prior to posting the complete schedule, unfilled shifts shall first be offered to part-time employees in the work area/unit to which they are assigned, up to equivalent full-time hours, before they are offered to casual employees. The Employer is not obligated to offer extra shifts to part-time employees where it results in the Employer having to pay a contract premium rate, unless the Employer would incur a contract premium rate regardless of which employee worked the shift (eg. Statutory Holiday Pay). Part-time employees who do not have a position on a specific unit will be given priority for extra shifts in any areas where they have received orientation and regularly work. Where Part-time employees accept additional shifts, they shall be paid the sum of thirty (30) cents per hour in addition to their regular hourly rate as per Article 32.05. Part-time or casual employees who accept these shift(s) will be required to work them in accordance with the collective agreement. Once all attempts to schedule the unfilled shifts have been exhausted, the remaining shifts will be offered at double the regular hourly rate for each hour worked.

With the exception of the conditions referenced above, all other overtime hours of work shall be applied and paid in accordance with Article 9 – Overtime.

Unless otherwise agreed to in writing by the parties to the Collective Agreement, this Letter of Understanding will not be extended or renewed.

Sincerely,

Fiona Ellis

Director of Collective Bargaining

Ms. Tracey Trahey Chief Negotiator Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Ms. Trahey:

*Re: Portability of Seniority

All employees of the RHAs shall have full portability of seniority without conditions after the amalgamation of the RHA's on April 1, 2023.

Effective April 1, 2023, employees who resigned from a permanent or temporary position with a Regional Health Authority to take a permanent or temporary position with the Provincial Health Authority and who forfeited seniority as a result of that resignation, shall be credited with their forfeited seniority, provided they commenced their new position within one hundred and twenty (120) days of resignation.

Casual employees who take another casual position with the Provincial Health Authority will maintain their seniority. Permanent or temporary employees who resigned from their position with a Regional Health Authority to take a casual position with the Provincial Health Authority shall forfeit their seniority as a result of that resignation, subject to the *Human Rights Act*, 2010.

There shall be no retroactive application of seniority.

Sincerely,

Fiona Ellis

Director of Collective Bargaining

Ms. Tracey Trahey Chief Negotiator Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Ms. Trahey:

*Re: Conversion of Casual and Part-time Employees

The Employer shall post expressions of interest, open to casual and part-time employees, to fill vacant permanent full-time or permanent part-time positions. The Employer may consider the option of filling one (1) full-time position with two (2) employees if there is an interest in a job sharing arrangement. In instances where a permanent full-time position is converted to two part-time positions to accommodate a job sharing arrangement, the Employer shall maintain the position as permanent full-time. Expressions of interest shall reflect the time limits for posting of positions outlined in the collective agreement or such greater time periods as the Employer and the Union may agree.

Within sixty (60) days from the date of signing this Collective Agreement, the Parties shall conduct a review of all current permanent vacancies and mutually agree to the specific positions and number of same that will be offered through conversion.

These identified opportunities shall be first offered on a seniority basis to employees who are orientated to and currently working in the work area/program. This shall include employees on any leave of absence or employees absent due to maternity, adoption or parental leave or because of a workplace injury for which the employee is in receipt of workers' compensation benefits.

Sincerely,

Fiona Ellis

Director of Collective Bargaining

Ms. Tracey Trahey Chief Negotiator Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Ms. Trahey:

*Re: Permanent Full-time (PFT) Incentive Pilot Project

Effective date of signing and for the life of this collective agreement, employees who currently hold or accept a full-time permanent position shall receive five thousand dollars (\$5000.00) annually to be paid on a bi-weekly basis for each bi-weekly pay period that an employee works full-time hours.

The parties agree to assess the effectiveness of the PFT Incentive Pilot Project on an annual basis.

The parties agree the pilot project will not be automatically renewed in a subsequent collective agreement.

Sincerely,

Fiona Ellis

Director of Collective Bargaining

Ms. Tracey Trahey Chief Negotiator Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Ms. Trahey:

*Re: Health Sector Safety Council

Within sixty (60) days of date of signing of the Collective Agreement, the Employer and the Union will request a joint meeting of Workplace NL to discuss the efficacy of forming a Health Sector Safety Council.

Sincerely,

Fiona Ellis

Director of Collective Bargaining

Ms. Tracey Trahey Chief Negotiator Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Ms. Trahey:

*Re: Preceptorship/Mentorship

Preceptorship Program

The parties recognize the importance of the role of the preceptor. Prior to the commencement of a preceptorship, the Employer and Employee will discuss work assignment adjustments, if required.

Preceptorship is the period of time designated formally by the Employer for a Preceptor to be working with a student (e.g. NP, RN, LPN, Paramedic).

Employee(s) who are assigned as a preceptor shall be paid seventy five cents (\$0.75) per hour, in addition to all other pay, for all assigned hours spent preceptoring up to a maximum of seven hundred and fifty dollars (\$750) per annum.

Mentorship Program Pilot Project

The Parties recognize the importance of mentorship as a means to support nurses as they adapt to new practice environments. The establishment of a nursing mentorship program is a priority for the Parties.

Mentoring is a voluntary, mutually beneficial professional relationship. A nurse mentor provides guidance and shares clinical experience and knowledge to support nurses new to an area gain confidence in their roles.

The Parties agree to establish a pilot nursing mentorship program by April 1, 2024.

A pilot will be implemented throughout NL Health Services. The Employer shall post annual Expressions of Interest for nurses interested in mentoring within their practice environment.

The maximum ratio shall be one (1) Mentor to three (3) Mentees. The duration of the mentoring relationship will be dependent on the needs of the mentee and will be evaluated by the manager and clinical educator, where applicable. The relationship will not be less than three (3) months in duration and will not exceed twelve (12) months.

The Parties commit to conducting an evaluation of the pilot to gauge its effectiveness in improving mentees' transition to new practice environments, and its impact on nursing recruitment and retention six (6) months prior to the expiration of the Collective Agreement.

Educational materials will be presented to mentors and mentees, in a mutually agreed format, prior to their engagement in the Mentorship program.

Employee(s) who are assigned as a mentor shall be paid a premium of nineteen hundred and fifty dollars (\$1950) per year and will be paid quarterly for each three (3) month period the employee is engaged as a mentor.

Employees assigned as mentors agree to share nursing knowledge and provide clinical support after the conclusion of the formal mentorship term.

Sincerely,

Fiona Ellis

Director of Collective Bargaining

Ms. Tracey Trahey Chief Negotiator Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Ms. Trahey:

*Re: Ten (10) Hour Shifts

The Parties recognize the requirement for a ten (10) hour shift within the Provincial Health Authority. A ten (10) hour shift will only be implemented in a work area by mutual agreement between the Parties. Once a 10-hour shift is implemented in an area, subsequent job postings will be posted with those hours of work. This is not a recognized day shift.

A ten (10) hour shift is defined as:

The normal hours of work for employees working a ten (10) hour shift schedule shall be seventy-five (75) hours a fortnight divided into 9.375 hour shifts excluding the unpaid meal period but including a rest period of fifteen (15) in the first half and second half of the shift. The meal period and rest period may be combined by mutual agreement between the employee and their supervisor.

Sincerely,

Fiona Ellis

Director of Collective Bargaining

Ms. Tracey Trahey Chief Negotiator Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Ms. Trahey:

*Re: Flexible Twelve (12) Hour Shift

The Parties recognize the need to develop modified start and end times, as an exception to Article 2.01 (v) (ii) "Twelve Hour Shifts", to respond to changing healthcare needs requiring the efficient delivery of services.

The Parties therefore agree:

- (a) Flexible 12-hour shifts shall be utilized only where required for service delivery and/or hours of operation.
- (b) The Provincial Health Authority shall notify the Union in writing of its intent to introduce shifts that are not in line with start times in Article 2.01 (v) (ii). Such notice shall identify the purpose and location of the flexible schedule.
- (c) The Provincial Health Authority and the Union shall meet at least thirty (30) days prior to implementation of the schedule. The shifts to be implemented shall be subject to mutual agreement. Such agreement shall not be unreasonably withheld or denied by the Union.

Sincerely,

Fiona Ellis

Director of Collective Bargaining

Ms. Tracey Trahey Chief Negotiator Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Ms. Trahey:

*Re: Expiration of Paid Sick Leave

The Employer commits to reviewing the interpretation of Article 18.03 as it applies to employees that are requesting to use their annual leave for occasional or day to day absences that is not associated with extended periods of sick leave.

Sincerely,

Fiona Ellis

Director of Collective Bargaining

Ms. Tracey Trahey Chief Negotiator Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Ms. Trahey:

*Re: Casual Employees

All existing obligated employees will have the option of moving onto the new obligated language or reverting to article 33.01.

Sincerely,

Fiona Ellis

Director of Collective Bargaining

Ms. Tracey Trahey Chief Negotiator Registered Nurses' Union NL P.O. Box 416 St. John's, NL A1C 5J9

Dear Ms. Trahey:

*Re: Nurses' Automobile Expenses

The Employer commits to reviewing job postings for nurses and nurse practitioners who are required to work at multiple sites to assess if a vehicle is required as a condition of employment.

Sincerely,

Fiona Ellis

Director of Collective Bargaining

Pilot Project

Granting of Leave of Less Than One Week after Schedule has been Posted

The parties recognize the overriding mandate to protect and maintain patient safety and agree that while the parties will endeavour to give effect to all terms herein, any scheduling requirements necessary for patient safety or care purposes will be implemented notwithstanding this agreement.

- 1. The parties may agree to conduct pilot projects.
- 2. Each pilot project will be three months in duration with the option to mutually agree to extend to six months at the end of the three month period, and will commence upon an agreed upon date.
- 3. Each project will be overseen by a committee composed of three union representatives and three Employer representatives.
- 4. The above noted committees will endeavour to communicate prior to and during the course of the pilot projects in an effort to ensure consistency in process.
- 5. Either party may terminate the pilot process with fourteen (14) calendar days' written notice to the other party.
- 6. The criteria for the projects will be developed by the appropriate committee but will be premised on:
 - a. Leave requests shall be granted upon request on:
 - i. Any shift where extra staff are scheduled in excess of core staffing levels or operational requirements on the pilot units (eg. Patient demand and/or mandatory in-servicing) or;
 - ii. Any shift where it is determined by the Employer that relief employees are available.
 - b. The Employer is not obligated to incur any overtime costs in granting such leave.
 - c. Joint information sessions on the pilot projects shall be completed with staff in the pilot units.
 - d. There will be evaluation before, during and after the pilot to gauge:
 - i. Whether there has been a reduction in sick leave and overtime costs in the pilot units over the duration of the project;
 - ii. Whether there has been a change in leave requests being granted;
 - iii. Whether there has been an improvement in morale and employee engagement in the pilot units; and
 - iv. Such other indicators as the parties may mutually agree upon.
 - e. In the event that a relief employee cancels a pre-booked shift, the Employer may cancel any approved leave that the relief employee was scheduled to provide relief for, subject to the availability of other qualified relief staff.

Pilot Project

Provincial Temporary Employees

The parties agree that, notwithstanding the definition of "Temporary Employee" in Article 2.01(y), the Employer may create temporary position(s) for special projects that do not have a permanent incumbent when:

- (i) The Employer has received external funding from a third party that can be used to create temporary RNUNL position(s) for specified projects for specified periods of time;
- (ii) The temporary position(s) are posted and awarded in accordance with the Collective Agreement for a maximum of 2 cumulative years (Extensions beyond 2 cumulative years require a private agreement with the RNUNL);
- (iii) The successful applicant(s) shall have their position(s) protected for the duration of the special project;
- (iv) RNUNL Provincial Office is to be given written notification by the Employer of the need for the temporary position(s), the source of the third party funding, the duration of the temporary position(s) and the qualifications associated with the temporary position(s);
- (v) Should the special project lead to the creation of permanent position(s) they shall be posted and awarded in accordance with the collective agreement.

The RNUNL shall have the right to grieve the qualifications associated with the temporary position(s) and individuals shall have the right to grieve the awarding of the temporary position(s), all in accordance with the Collective Agreement. No grievances shall be filed on the basis that the temporary position(s) do not meet the definition of "Temporary Employee" in Article 2.01(y).

The RNUNL shall have the right to terminate this pilot project upon 30 days notice to the Employer but such termination will not impact special projects introduced up to the date of termination.

AGREEMENT ON PENSIONS

The Parties agree to the following:

- 1. Introduction of a formal indexing program for those pensioners and survivors who have reached age 65, as follows:
 - 60% of the annual change in the national CPI as published by Statistics Canada (Catalogue 62-001), in the calendar year immediately preceding the anniversary date, to a maximum annual increase of 1.2%;
 - a) For those pensioners and survivors who have attained age 65 from October 1, 2002; and
 - b) For those pensioners and survivors who are not age 65, from the next anniversary date after the date they reach age 65.

Cost: 2% of salary to be shared equally by both parties.

Anniversary Date: October 1, 2002 and every October 1 thereafter.

- 2. Government will pay \$982 million into the Public Service Pension Plan (PSPP) with \$400 million being paid on March 15, 2007 and the remaining balance of \$582 million will be paid by June 30, 2007.
- 3. This memorandum of agreement will not take effect unless all participants, the Newfoundland and Labrador Association of Public and Private Employees, the Canadian Union of Public Employees, the Newfoundland and Labrador Nurses' Union, the Association of Allied Health Professionals, the Canadian Merchant Service Guild, the International Brotherhood of Electrical Workers and Her Majesty the Queen in Right of Newfoundland (represented by the Treasury Board) agree to its terms.
- 4. It is agreed that the payment outlined in Clause 2 above is full settlement of Government's share of the unfunded liability of the PSPP as established on December 31, 2000 and outlined section 2 of the Memorandum of Understanding 2004, Agreement on Pensions and there shall be no further special payments.
- 5. A committee of the parties will be established to identify and resolve any matters required to implement joint trusteeship by April 1, 2008.
- 6. All unions representing Public Service Pension Plan members must indicate, in writing, acceptance of this proposal.
- 7. For the duration of the Collective Agreement the Employer agrees to maintain the Public Service Pension Plan as an independent pension plan.

CLASSIFICATION PLAN

It is agreed that a new classification system would be implemented and that the plan used would be gender neutral. It is also agreed that RNUNL would have input into the selection and implementation of the system. This will be accomplished through a joint steering committee which would be advisory to Government in nature. It is also agreed that the current classification plan would continue until the new plan is established.

It is agreed that the new plan began implementation on April 1, 2008. However, any wage adjustments necessary for implementation of this plan will not accrue on April 1, 2008. The total cost and the timing of any wage adjustments are to be included in negotiations to commence on Government's finalization of the new classification system.

The Unions require that a Job Evaluation Consultant (as selected by the Unions) would have direct contact with the Plan's consultant and have full access to all relevant information. This individual would also communicate with and have access to all meetings of the Steering Committee. The salary and the expenses of the Advisor would be borne by the Unions.

The ratings of the positions will be conducted by the staff of the Classification, Organization and Management Division of Treasury Board. There will be a Benchmark Committee composed of two-thirds management and one-third union representatives who will review the sampling of the ratings as they are done. The Benchmark Committee would have the authority to refer results back to the raters should they be deemed inconsistent. The final decision making authority rests with Treasury Board.

While the new Job Evaluation system is being implemented, all employees can proceed with individual reviews and appeals under the current plan. However, there will be no further occupational reviews.

LAYOFFS DURING THE TERM OF THE COLLECTIVE AGREEMENT

Notwithstanding any Article or Provision of the Collective Agreement, and for the duration of this collective agreement, the Employer shall not layoff for reasons other than lack of work or abolishment of a position that would be considered business as usual in the course of normal Employer operations. The Employer shall not use layoffs to effect Provincial budgetary expenditure reductions.

ARBITRATION

The Parties agree to the following:

- 1. Within six (6) months from the date of signing this Collective Agreement, there shall be a committee established with equal representation to create a regular schedule and process for mediation/arbitrations and arbitration hearings.
- 2. The terms of reference are to be determined by the Committee.

DISPUTE RESOLUTION

The Parties agree to the following:

- 1. Within one hundred and twenty (120) days from date of signing this Collective Agreement, there shall be a joint committee created with equal representation from the Parties to develop an education session for Article 14 Dispute Resolution.
- 2. The Employer agrees to allow at least one (1) RNU member and volunteer to sit on this committee in accordance with Article 14.02.
- 3. The educational material shall be presented within one hundred and twenty (120) days from the date the committee was established.
- 4. The educational materials will be presented to RNU volunteers, members, managers and HR representatives on an annual basis in a mutually agreed format.

PROFESSIONAL PRACTICE COMMITTEE

The Parties agree to the following:

- 1. Within one hundred and eighty (180) days from date of signing this Collective Agreement, there shall be a joint committee created with equal representation from the Parties to develop an education session on Professional Practice.
- 2. The educational material shall be presented within one hundred and twenty (120) days from the date the committee was established.
- 3. The educational materials will be presented to RNU volunteers, members, managers and HR representatives on an annual basis in a mutually agreed format.

BACHELOR OF NURSING IN COMMUNITY HEALTH

All Community Health Nurse I positions shall be posted with the qualifications identified through the Newfoundland and Labrador Government Job Evaluation System dated July 15, 2015.

The Parties agree that a diploma prepared nurse who is willing to complete a post diploma/certificate program in Public Health/Community Health, shall be considered for Community Health Nurse I positions. Where a diploma prepared nurse is awarded a Community Health Nurse I position, the Employer shall provide funding for the program.

Where a diploma prepared nurse is awarded a Community Health Nurse I position, they shall start in the position upon successful completion of the post diploma/certificate program. They shall complete the post diploma/certificate program in Public Health/Community Health within the program's designated timeframe.

MEMORANDUM OF AGREEMENT Re: Kilometer Rate Adjustment Formula (RNUNL)

General

- 1. The purpose of this Memorandum of Agreement (MOA) is to provide a mechanism for the periodic adjustment of the kilometer rate(s) contained in applicable collective agreements for employees who are either required to provide a vehicle as a condition of employment or who may be authorized to use a personal vehicle on Employer's business.
- 2. The terms of this MOA shall be applicable to employees who are members of a bargaining unit covered by a collective agreement listed in Schedule 'A'.
- 3. Adjustments shall be calculated by the Public Service Secretariat and posted to the Human Resource Policy Manual website: www.gov.nl.ca/hrpm. Should there be any dispute as to the calculated rate; the rate established by the Public Service Secretariat shall prevail.

Adjustment Formula

4. <u>Base Fuel Rate</u>

The 'base fuel rate' for calculating fuel costs is 79.4ϕ per liter.

5. Fuel Price

'Fuel prices' shall be those set by the Petroleum Pricing Office for the Avalon Region (Zone 1).

6. Base Kilometer Rate

The 'base kilometer rate(s)' shall be the reimbursement rate(s) contained in an applicable collective agreement.

- 7. <u>Initial Adjustment October 1, 2005</u>
 - a) The 'base kilometer rate' shall be adjusted effective October 1, 2005 based on the difference in the 'fuel price' on October 1, 2005 and the 'base fuel rate' multiplied by 1/10.

('fuel price' on October 1, 2005 - \$0.794) X 0.10 = km rate adjustment

[km rate adjustment is added to the 'base kilometer rate']

b) Kilometer rates shall be rounded to four decimal places after the dollar (\$0.0000).

8. Adjustment Dates (Quarterly Adjustments)

Effective January 1, 2006, the kilometer rate shall be adjusted, based on the 'Adjustment Formula', on a quarterly basis on the following dates each year:

January 1st April 1st July 1st October 1st

9. Adjustment Formula

a) The 'base kilometer rate(s)' shall be adjusted (up or down) on each of the 'adjustment dates' based on the difference in the 'fuel price' on the 'adjustment date' and the 'base fuel rate' multiplied by 1/10.

('fuel price' on 'adjustment date' - \$0.794) X 0.10 = km rate adjustment

[km rate adjustment is added to the 'base kilometer rate']

b) Kilometer rates shall be rounded to four decimal places after the dollar (\$0.0000).

10. Reimbursement Rate

Reimbursement shall be at the rate(s) in effect on the date of travel.

Effective Date

- 11. The MOA shall be effective October 1, 2005, and in accordance with Clause # 10, shall only be applicable to travel which occurs from that date forward.
- 12. This MOA may be terminated upon thirty (30) days notice from either party.

IN WITNESS WHEREOF the parties hereto have executed this Agreement the day and year first before written.
SIGNED on behalf of the Public Service Secretariat, by its proper officers in the presence of the witness hereto subscribing:
Witness
SIGNED on behalf of the Newfoundland and Labrador Health Boards Association
Witness
SIGNED on behalf of the Newfoundland and Labrador Nurses Union, by its proper officers in the presence of the witness hereto subscribing:
Witness

SCHEDULE 'A'

Collective Agreements

RNUNL Provincial Agreement

Gander & District Continuing Care

St. John's Home Care Program

*MEMORANDUM OF AGREEMENT

Nursing Practice and Leadership Premium

General

1. The purpose of this Memorandum of Agreement (MOA) is to recognize and reward the valuable contribution of those registered nurses (RNs) who demonstrate additional leadership and practice - focused commitment in their work beyond what is normally expected of them in the course of their work.

From a nursing practice perspective the premium is intended to recognize the personal commitment of registered nurses' engagement and involvement in additional professional development opportunities beyond workplace expectations. From a leadership perspective, the premium is intended to recognize and reward nurses RNs who take on leadership roles in pursuit of the advancement of the nursing profession and nursing practice.

- 2. Commencing July 1, 2023, four hundred and seventy five thousand dollars (\$475,000) per year, will be designated as available funding under this MOA. Unused funding in any year is to be carried over and to be available for use in subsequent years.
- 3. An RN nurse who qualifies for the Nursing Practice and Leadership Premium will be paid a maximum annual premium of four hundred dollars (\$400) per year provided the available funding under the MOA is sufficient. In any year that the available funding is not sufficient to provide the four hundred dollar (\$400) premium to each eligible nurse RN, the premium for that year shall be determined by dividing the available funding between the number of eligible nurses RNs.
- 4. In order to qualify for the Nursing Practice and Leadership Premium for a 12 month period from April 1 to the following March 31 each year, a nurse RN must earn 50 points in that 12 month time period when participating in Employer approved activities. The list of approved activities along with the relative point value and weight for each is included as Appendix A. A list of examples is included in Appendix B. It is understood that the lists are not exhaustive but will guide the Employer in determining relative point values for other approved activities. Points will be recognized retroactively to April 1, 2023 for the period April 1, 2023 to March 31, 2024.
- 5. It is the sole responsibility of the nurse RN to maintain a record of recognized activities completed on an ongoing basis and to provide acceptable evidence to the Employer that they qualify for the Nursing Practice and Leadership Premium as per the terms of this MOA. The nurse RN shall provide written proof of these activities on the form provided as Appendix C, attached to this MOA, to the Employer, with the understanding that all proof must be submitted to the Employer for consideration by April 30th of each year. Following proof of eligibility, the premium will be paid in a lump sum on or before June 30th of each year.
- 6. The Premium shall not be prorated for part time and casual nurses RNs.

- 7. The nursing practice and leadership premium is intended to recognize the additional "value added" practical work and leadership initiative a nurse RN may commit to voluntarily. Any duties/activities that a nurse RN is required to perform or participate in as part of their position requirements with the Employer cannot be claimed for points under this MOA.
- 8. This MOA or its contents are no way intended to provide employees with greater access to or opportunity to engage in, any of the activities listed herein. In addition, the Employer is not obligated to provide specific programs in order to assist the nurse RN to qualify for this benefit.
- 9. Where the terms of this MOA conflict with the collective agreement, this MOA shall prevail.
- 10. Points earned with one Employer will be recognized by a subsequent Employer for the purposes of qualifying for this premium. Transfer and Portability requirements of the Collective Agreement must be met.
- 11. Disputes arising under this MOA shall be subject to the grievance and arbitration process under the collective agreement with the exception of disputes relating to a nurse's RNs eligibility under this MOA. In the event a dispute arises with respect to a nurse's RNs eligibility, the merits of the dispute will be reviewed in a timely manner by the appropriate Director in consultation with a union representative.
- 12. The Employer will provide the RNUNL with the following:
 - (1) the number of eligible nurses RNs by Employer and;
 - (2) the amount of money available in that year and;
 - (3) the amount of the premium per eligible nurse RN for that year.

The Employer will endeavor to provide this information two weeks prior to the annual payment of the premium.

Appendix A

Voluntary education, as outlined in the categories below, will qualify towards the Nursing Practice and Leadership Premium. In order for a RN to qualify for this premium, the RN must attain 50 points based on the relative weights assigned to the following approved activities over the time frame of April 1 to the following March 31.

Areas that do not qualify toward this premium:

- 1. Orientation education.
- 2. Compulsory in service programs.
- 3. Education (e.g. Course, Certification, or Diploma) required because of the location or service in which the RN works.
- 4. Education that is compensated under the Educational Differential provisions of the collective agreement.

A. Certification or Diploma in a Speciality (30 Points)

This is defined as a course of study of greater than three months in duration which includes an evaluation component and which may lead to a specialty certification status/or specialty certificate/or diploma. The RN may claim these points if they are enrolled in an eligible specialty certification or diploma program. The points are intended for the study period. Where the course of study spans more than one year these points can be claimed in each study year. In the final year of study the RN will receive either points under this MOA or appropriate compensation under the Educational Differential provisions of the collective agreement; however they cannot receive both. Should a RN have to repeat a course, the points cannot be claimed again.

B. Course in a Speciality (20 Points)

This is defined as a course in a specialty for which there is a required evaluation component to "pass". These points can only be claimed in the year the course is taken. For those courses that require recertification, 5 points for subsequent years while the course certification remains valid.

C. Course, Workshop or Conference (10 – 15 Points as specified)

This is defined as voluntary attendance at a course/learning session, workshop or conference, whether conducted face to face, through webinars or by other means, whereby the skills/theory of topic, is applicable to the practice environment where the RN works. If the course, workshop or conference is a minimum of 3.5 hours in duration, the RN will receive 10 points. If the course, workshop or conference is a minimum of 7.5 hours in duration the RN will receive 15 points.

D. Employer and/or External Education Sessions (5 Points)

This category is applicable when the RN attends an education event which is minimally 1 hour in duration.

E. Electronic/Modular Learning (5 Points)

There are many examples of learning delivered via electronic/modular methods that may be Employer developed or that may be offered through the public domain. The RN must provide proof of having participated and completed the modules.

The education must be a minimum of one (1) hour in duration; however the RN may accumulate time from several sessions to obtain the one (1) hour requirement.

F. Education (10-25 Points as Specified)

If the RN is enrolled in University within the 12 months claim period and then graduates within that same 12 month period, they will now receive the appropriate premium under the Collective Agreement and they cannot also claim for points under this section.

- ✓ Enrolled in PhD in Nursing Program (minimally completing 2 courses per year-25 points)
- ✓ Enrolled in Masters Nursing program (minimally completing 2 courses per year-20 points)
- ✓ Enrolled in Nursing Degree Program (minimally completing 2 courses per year-15 points)
- ✓ Enrolled in Nurse Practitioner Program (minimally completing 2 courses per year-20 points)
- ✓ Instructor-Level Achievement for designated courses (e.g. BLS, ACLS, PALS, NRP, ATLS, ACORN, ALSO, CTAS TNCC, -10 points

G. Committee/Task Force Involvement (for Employer, Province, Union, Professional Organization or Association or Agency)

Many RNs sit on various committees that meet regularly throughout the year which are not part of the regular responsibilities of their position. Some of these committees/task forces are joint, others are internal to the Union, Professional organization or association or agency. In order to claim for the correct number of points the RN must indicate the number of times they have been present at the Committee and provide evidence (i.e. formal minutes of the meeting or confirmation letter of attendance from organization) to support the claim. The RN must attend a minimum of 50% of the yearly meetings in order to claim points.

- Chair of the committee/task force that meets 9-12 times per year (20 points)
- Member of the committee/task force that meets 9-12 times per year (15 points)
- Chair of the committee/task force that meets 4-8 times per year (15 points)
- Member of the committee/task force that meets 4-8 times per year (10 points)
- Chair of the committee/task force that meets 1-3 times per year (10 points)
- Member of the committee/task force that meets 1-3 times per year (5 points)
- Attends the business session of the RNUNL convention, the CFNU Biennium, CRNNL AGM or CNA biennial convention. (5 points).

H. National/Provincial/Branch Union or Professional Association Involvement

This category can only be claimed by a RN that is a member in good standing with their Professional Association on April 1st each year. A RN cannot claim for this if their membership has expired at the time of submission. A RN cannot claim for this by virtue of membership in the College of Registered Nurses of Newfoundland and Labrador (CRNNL) or the Registered Nurses' Union Newfoundland and Labrador (RNUNL). Please note, if a RN is a member of a National or Provincial organization that automatically makes them a member of a local association, the RN can only claim for the parent organization.

	President	Executive/board	Member
		member	
National	20	15	5
Provincial	15	15	5
Branch	15	15	

Points are also assigned as follows:

• Shop Steward of a Branch or CRNNL Workplace Representative (10 points)

I. Publications/Presentations

To claim for publication, the RN must have been published in the past 12 months of the claim period. They will also be required to provide information on where, when and the name of the conference attended. The conference attended must be related to the advancement of the nursing profession and nursing practice.

- Publication in a peer-reviewed professional journal or textbook (15 points)
- Publication in a non-peer-reviewed journal (10 points)
- Speaker at a National Conference (15 points)
- Speaker at a Provincial Conference (10 points)
- Speaker at a Local Conference (10 points)
- Speaker at an Employer-based in-service session (5 points)
- Poster Board Presenter at a National Conference (15 points)
- Poster Board Presenter at a Provincial Conference (10 points)
- Poster Board Presenter at a Local Conference (10 points)
- Poster Board Presenter at an Employer-based Conference (5 points)
- Member of a panel speaking on a topic (5 points)
- Publication of Article in Hospital, Union or Professional Association Newsletter (5 points)

J. Research

To claim in this category the research must either be in progress or completed at any time within the 12 month claim period. The RN can claim for each study only once even if the study spans more than one year. The RN must be "officially" listed as either the Principal Investigator or a Co-Investigator.

- Primary Investigator as part of a multi-site study (15 points)
- Co-Investigator as a part of a multi-site study (10 points)
- Primary Investigator of an Employer/unit based research study (15 points)
- Co-Investigator of an Employer/unit based research study (10 points)
- Develops a unit-specific research proposal (5 points)
- Conducts a literature review as part of a research study (5 points)

Appendix B

Registered Nurses are encouraged to consult with their manager in advance on eligibility of Practice and Leadership initiatives not listed in this Appendix.

It is understood that Appendix B is not exhaustive but will guide the parties in determining relative point values for other approved activities.

- A. Certification/Diploma in a Specialty: 30 Points
 - CNA Certification
 - Lactation consultant
 - Infection control
 - RN First Assist
 - Enterostomal therapist
 - Critical care nursing program CCNP
 - Diabetes Educator
 - Post Basic Perioperative nursing
 - Occupational Health
 - Gerontology post basic
 - More OB
- B. Course in a Specialty: 20 Points
 - Advanced Cardiac Life Support (ACLS)
 - Pediatric advanced Life Support (PALS)
 - Advanced Trauma Life Support (ATLS)
 - Advanced Life Support, Obstetrics (ALSO)
 - Acute Care of the at Risk Newborn (ACORN)
 - Advanced Pediatric Life Support (APLS)
 - Trauma Nursing Core Course (TNCC)
 - Neonatal Transport team Course
 - Adult Transport team Course
 - Hyperbaric Course
 - Chemotherapy Course
- C. Course, workshop or Conference: 10 -15 Points maximum for <u>each</u> conference, workshop or course attended)
 - Any nursing related topic
 - Team Building
 - Conflict resolution
 - Education day(s) at RNUNL convention
 - Education day at CRNNL Convention
 - CFNU biennium education days
 - RNUNL Labour School
 - PRISMA
 - Canadian Triage Acuity Score (CTAS)

D. Employer and/or external Education sessions: 5 Points

- Attendance at grand rounds
- Journal club
- Lunch and learn sessions
- Webinars
- CRNNL teleconferences
- Video conferences

E. Electronic/Modular Learning 5 Points: 5 Points

- Breast feeding module
- Port-a-cath module
- Grievance pilot project e-learning

F. Education

- Enrolled in PhD in Nursing Program (minimally completing 2 courses per year-25 points)
- Enrolled in Masters Nursing program (minimally completing 2 courses per year-20 points)
- Enrolled in Nursing Degree Program (minimally completing 2 courses per year-15 points)
- Enrolled in Nurse Practitioner Program (minimally completing 2 courses per year-20 points)
- Instructor-Level Achievement for designated courses (e.g BLS, CTAS, TNCC, ACLS, PALS, NRP, ATLS, ACORN, ALSO-10 points

APPENDIX "C"

NURSING PRACTICE & LEADERSHIP PREMUIM

Claim for Nursing Practice & Leadership Premiums To be Completed and Submitted by Registered Nurse by May 1st

Eligible 12 Mo Period	nth	April 1, 20				to March 31						
		Name										
			FTE Status (PFT/PPT/TFT/Casual)									
Personal Inform	ation	Job Ti	tle									
		Classi	fication									
		Unit/D worke	ept/Area	a								
		Manag	jer's Na	me								
Points Claimed	Α	В	С	D	E	F	G	Н	I	J	K	Total
by Category (Details Attached)												
Signe	ed					Da	ite:					
For Intern	nal Use	by Mana	agemen	t:								
Points	A	В	С	D	Е	F	G	Н	I	J	K	Total
Approved by Category (Details Attached)												
Premium Approval		F	Premium	Approve	d		Premiur	n Not App	oroved _			
Rationale												
Się	gned:					D	ate:				_	

Description of Nursing Practice & Leadership Points Claimed

Category	Particulars	Date	Points Claimed	Internal Use – Points Approved
A – Certification or				
Diploma in a Specialty (30 Points)				
(30 i dilita)				
		Total		

Category	Particulars	Date	Points Claimed	Internal Use – Points Approved
B – Course in a Specialty				
(20 Points)				
		Total		

Category	Particulars	Date	Points Claimed	Internal Use – Points Approved
C - Course, Workshop				
or Conference (10-15 Points as				
Specified)				
		Total		

Category	Particulars	Date	Points Claimed	Internal Use – Points Approved
D – Employer and/or External Education				
Sessions (5 Points)				
(3 Folits)				
		Total		

Category	Particulars	Date	Points Claimed	Internal Use – Points Approved
E – Electronic/Modular Learning				
(5 Points)				
		Total		
		Total		

Category	Particulars	Date	Points Claimed	Internal Use – Points Approved
F – Education				
(10-25 Points as Specified)				
		Total		

Category	Particulars	Date	Points Claimed	Internal Use – Points Approved
G – Committee/Task Force Involvement (for Employer, Province, Union, Professional Organization or Association or Agency (5-20 Points, as Specified)				Леричи
		Total		

Category	Particulars	Date	Points Claimed	Internal Use – Points Approved
H – National/Provincial/				
Branch Union or Professional Association				
Involvement (5-20 Points, as				
Specified)				
		Total		

Category	Particulars	Date	Points Claimed	Internal Use – Points Approved
I – Publications/ Presentations (5-15 Points, as Specified)		Total		
		Coro S		

Core S

Particulars	Date	Points Claimed	Internal Use – Points Approved
	Total		
	Particulars	Particulars Date Total	

INDEX

		ARTICLE	PAGE
		NUMBER	NUMBER
ARBI	TRATION	15	24
15.01	Composition of Board of Arbitration		24
	Board Procedure		
15.03	Decision of the Board		
15.04	Jurisdiction of the Board		24
15.05			
15.06	Expenses of the Board		24
15.07	Amending of Time Limits		
15.08	Time Off for Witnesses		25
15.09	Expedited Arbitration		25
CALL	.BACK	11	18
11.01	Callback Pay Rate		18
11.02	Transportation Expenses		18
11.03	Transportation Expenses		18
11.04	Additional Duties		18
11.05	Return to Work Following Callback		18
CASU	JAL EMPLOYEES	33	49
33.01	Obligation		49
33.02	Casual Employee Availability		49
33.03	Benefits		50
33.04	List of Casual Employees		51
33.05	Letters of Appointment and Availability		51
33.06	Contact Information		51
33.07	Notification of Unavailability		51
33.08	Obligation Regarding Pre-Booked Shifts		51
33.09	Seniority for Casual Employees		51
33.10	Work on Statutory Holiday		52
33.11	Seniority for Casual Employees While in Temporar	y Positions	52
33.12	Casual Employee Unavailable because of		
	Maternity/Adoption/Parental Reasons		52
COM	PASSIONATE LEAVE	21	36
21.01	Compassionate Leave		36
21.02	Special Leave Without Pay		37
	PLETE AGREEMENT		
41.01	Complete Agreement		59
	Amending or Altering Agreement		
CONT	TRACTING OUT	45	61
45.01	Job Protection		61
45.02	Period of Notice		62

DISCI	PLINE	.30	47
30.01	Employer Actions		47
30.02	Procedure	• • • • • • • • • • • • • • • • • • • •	47
30.03	Adverse Report	• • • • • • • • • • • • • • • • • • • •	47
30.04	Reinstatement	• • • • • • • • • • • • • • • • • • • •	48
30.05	Personal Record		48
30.06	Redress	• • • • • • • • • • • • • • • • • • • •	48
30.07	Work Days		48
30.08	Employee File		48
30.09	Justice and Dignity Provision		48
DISPU	TE RESOLUTION	.14	21
14.01	Prompt Procedure		21
14.02	Assistance from Shop Stewards		21
14.03	Types of Grievances		21
14.04	Employer Representative at Grievance Steps		21
14.05	Effective Date of Grievance Settlement		22
14.06	Technical Objection		22
14.07	Settlement of Dispute		22
14.08	Grievance Presented by Mail		23
14.09	Amending of Time Limits		23
14.10	May Omit Grievance Steps		
14.11	Bypass Steps		23
DRES	SING ROOMS AND PARKING FACILITIES		
31.01	Dressing Rooms		48
31.02	Parking Facilities		49
EDUC	ATIONAL DIFFERENTIAL	.44	61
44.01	Educational Allowance		61
44.02	Psychiatric Classification BN		61
EMPL	OYEE RIGHTS	.5	5
5.01	Employee Right		.5
5.02	Professional Practices Committee		.5
5.03	Employer Shall Not Discriminate		.5
5.04	Posting of Rules and Regulations		.5
5.05	Work of the Bargaining Unit		.5
5.06	Signing Reports		5
5.07	Personal Files		6
5.08	Consultation of File		6
ESCO	RT DUTY	.12	18
12.01	Compensation for Escort Duty		
12.02	8/12 Hour Shifts - Returning Within 7 1/2 Hours and 11	1/4 Hours	18
	No Loss of Days of Rest		
12.04	Return to Work Following Escort Duty		19
	Escort Duty - Travel Allowance		

12.06	Escort Duty Pay		20
12.07	Escort Duty Overtime		20
12.08	Volunteer Escort Duty		20
12.09	No Allowance Within 25 Mile Radius		20
12.10	Travel Insurance		20
12.11	Meal Allowance on Escort		20
12.12	Private Accommodations		
FAMI	LY LEAVE	46	62
46.01	Family Leave		62
	·		
FAMI	LY VIOLENCE LEAVE	48	65
48.01	Family Violence Leave		65
HEAL	TH PROGRAM	28	46
28.01	Staff Health Services		46
28.02	Medical Examinations		46
28.03	Employee's Choice of Physician		46
28.04	Immunizations		46
28.05	Radiological Hazards		46
HOLII	DAYS	17	30
17.01	Designation of Holidays		30
17.02	New Holidays		30
17.03	Work on Designated Holiday		30
17.04	Statutory Holiday During Vacation Period		31
17.05	Statutory Holiday on Day of Rest		31
17.06	Compensation for Holidays Falling on Saturday		31
17.07	Compensation for Holidays Falling on Sunday		31
17.08	Christmas and New Year's Leave		31
17.09	Statutory Holiday During Sick Leave		31
17.10	Statutory Holiday While on Workers' Compensation		31
HOUR	S OF WORK	8	8
8.01	Hours of Work		8
8.02	Meal Period		11
8.03	Working Schedule		
8.04	Rotation of Shifts		
8.05	Consecutive Shifts		
8.06	Days of Rest		12
8.07	Preference for Days of Rest		12
8.08	Exchanging Days of Rest		
8.09	Weekends Off		12
8.10	Rest Between Scheduled Shifts		13
8.11	Double Shift		13
8.12	Differentials		13
8.13	No Split Shift		13
8.14	Hours of Work Not Applicable to Public Health Nurses	J	14

8.15	Two Shift Rotation (Eight (8) Hour Shifts)	14
8.16	Twelve (12) Hour Work Schedule	14
8.17	Permanent Evenings and Nights	14
INITE	ERPRETATIONS AND DEFINITIONS2	1
2.01	Interpretations and Definitions	
2.01	interpretations and Definitions	1
LEAV	VE FOR ATTENDANCE - CRNNL ANNUAL MEETING42	59
	Leave to Attend Meeting	
	TERS:	
	al Leave and Vacation	
	al Paid Sick Leave	201
Annu	al Vacation and Holiday Provisions for Employees Employed	
	by the Government of Newfoundland and Labrador	
B.N.	2000	195
Casua	al Employees	230
Conv	ersion of Casual and Part-time Employees	222
Enhai	nced Senior Level Collaboration between RNUNL, Government	
	and NL Health Services	202
Expir	ation of Paid Sick Leave	229
Flexil	ble Hours of Work Community Health	204
Flexil	ble 12-Hour Shifts	228
Gand	er and District Continuing Care Program	
	and St. John's Home Care Program	198
Healt	h Sector Safety Council	224
Healt	hy Workplace Pilot	200
	g Practices	
Job P	ostings	215
Loss	of Seniority	190
Mark	et Adjustment	203
Nurse	es' Automobile Expenses	231
Nurse	e Uniforms	205
Orgai	nization of Nursing Work Studies	193
	Post-Employment Benefits ("OPEB")	
	Post-Employment Benefits (OPEB) Eligibility	
	Γime Employees Working Additional Shifts	
	anent Full-time (PFT) Incentive Pilot Project	
	bility of Seniority	
	ptorship/Mentorship	
	ncial Nursing Network	
	ncial Workload Measurement Committee	
	ew of Relief Staffing	
	ols of Nursing	
	rance Pay	
	Leave	
	tory Holiday and Weekends Off	
	tory Holiday Replacement	

Temp	orary Overtime Rate of Pay		220
_	our Shifts		
MANI	A CEMENT DICHTS	4	5
	AGEMENT RIGHTS		
4.01	Management Rights	•••••	3
MATI	ERNITY/ADOPTION/PARENTAL LEAVE	19	34
19.01	Request for Maternity/Adoption/Parental Leave		34
19.02	Dismissal		34
19.03	Return to Work		34
19.04	Vacation Before or After Leave		34
19.05	Period of Protection		34
19.06	Service While on Maternity/Adoption/Parental Leave		34
19.07	Group Insurance While on Maternity/Adoption/Parental 1		
19.08	Standby, Callback and Overtime - Pregnant Employees		35
19.09	Standby, Callback and Overtime - Breastfeeding		
19.10	New Positions		35
19.11	Job Postings		35
19.12	Illness Associated with Pregnancy		
	ORANDUM OF AGREEMENT		
	ilometer Rate		
Re: N	ursing Practice and Leadership Premium		245
MEM	ORANDUM OF UNDERSTANDING		
	ment on Pensions		234
_	ation		
	lor of Nursing in Community Health		
	fication Plan		
	te Resolution		
	fs During the Term of the Collective Agreement		
-	ssional Practice Committee		
Tiores	sional Fractice Committee	•••••	237
NURS	SE PRACTITIONERS	34	53
34.01	Nurse Practitioners – MAID		53
	RTIME		
9.01	Overtime Rate		
9.02	Regular Hourly Rate		
9.03	Calculation of Overtime		
9.04	Approval of Overtime		
9.05	Sharing of Overtime		
9.06	Change of Days of Rest		
9.07	Change of Shift		15
9.08	Notification of Changes in Schedule		
9.09	State of Emergency Due to Weather Conditions		
9.10	No Lay-Off to Compensate for Overtime		16
9.11	Consecutive Work Premium		

9.12	Daylight Savings Time		17
9.13	Mandatory Exemption		17
	-TIME EMPLOYEES		
32.01	Hourly Rate		
32.02	Awarding of Benefits to Part-Time Employees		
	Calculation of Seniority		
	Letter of Appointment		
32.05	Part-Time Employees - Additional Shifts		49
PENS	ION PLAN	26	44
	Pension Plan		
20.01	1 chiston 1 lan	••••••	••••
PILO	T PROJECT		
	ng of Leave of Less Than One Week After Schedule h		
Provir	icial Temporary Employees		233
DOCIT	TION DROTECTION	20	25
	TION PROTECTION		
20.01	Position Protection		
20.02	Period of Protection		
20.03	Protection for Education Leave		
20.04	1 2		
20.05	Protection for Permanent Employees Awarded Temp	orary Positions .	30
PROF	ESSIONAL PRACTICE COMMITTEE	49	65
49.01	Professional Practice Committee		
PROM	IOTIONS AND STAFF CHANGES	24	41
24.01	Effective Date		41
24.02	Posting of Vacancies		41
24.03	Contents of Postings		41
24.04	Awarding of Positions		
24.05	Notification of Successful Applicant		42
24.06	Trial Period		
24.07	Return to Former Position		42
24.08	Temporary Vacancies		43
24.09	Displacement		43
24.10	Reduction of Hours of Work		43
DUDD	OSE OF AGREEMENT	1	1
1.01	Purpose of Agreement		1
RECC	GNITION	3	4
3.01	Recognition of Union		
3.02	New Classifications		
3.03	No Private Agreement		
3.04	Agreement Overrules Hospital Policy		
3.05	Future Legislation		

3.06	Union Access		4
3.07	Permission to Hold Meetings		
	OACTIVITY		
50.01	Retroactivity		67
CALA	DIEG	25	4.4
	RIES Receive Salaries as Per Schedule A		
	Unregistered Nurses Isolation Allowance		
	Labrador Benefits		
25.05	Reclassification		44
SCHE	DULES		
	ule A – Step Progression and Contact Allowance – Salari	ies	69
	ule B – Experience Credits		
	ule C – Employers Covered by this Agreement		
	ule D – Classifications of Nurses Covered by this Agreer		
	ule E – Vacation Leave		
	ule F – Classification Review and Appeal Process		
	ule G – Summary of Group Insurance Benefits		
	ule H – Labrador Benefits Agreement		
	ule I – Deferred Salary Leave Plan		
	ule J – Job Sharing Guidelines		
	ule K– Seniority Numbers		
	ule L– Transition Agreements		
	ule M– Alternate Scheduling		
	ule N – Layoff and Recall		
	ule O – MOA of November 2015		
Dellea		•• ••••••	102
SCHC	OOLS OF NURSING	43	59
43.01	Hours of Work		59
43.02	Work Preparation		60
43.03	Attendance at Meetings of Faculty		
43.04	Leave of Absence		
43.05	Text Books		60
43.06	Attendance at Workshops		60
	Condition of Employment		
43.08			
43.09	Instructional Assignments		60
43.10	Graduation		
43.11	Merger and Amalgamation		
SENIC	ORITY, LAYOFF AND RECALL	23	38
23.01	Seniority Defined		38
23.02	Seniority Roster		38
23.03	Protest Period		38
23.04	Layoff and Recall		38

23.05	Advance Notice of Layoff		39
23.06	Loss of Seniority on Suspension		39
23.07	Loss of Seniority		39
23.08	No New Employee(s)		39
23.09	Transfer and Seniority Outside the Bargaining Unit		
23.10	Probation for Newly Hired Employees		
23.11	End of Temporary Employment		
23.12	Temporary Employee Commitment		
	r r v y r v y v v v v v v v v v v v v v		
SICK	LEAVE	18	32
18.01	Annual Paid Sick Leave		
18.02	Proof of Illness		
18.03	Expiration of Paid Sick Leave		
18.04	Sick Leave for Preventative Medical Care		
18.05	Sick Leave Record		
18.06	Group Life and Extended Health Plan		
18.07	Fringe Benefits		
18.08	Sick Leave Credits for the First and Last Month of Emplo		
18.09	Injury on Duty		
18.10	Extension of Sick Leave		
16.10	Extension of Sick Leave	•••••	33
SDEC	AL ALLOWANCE	35	53
	Uniform Allowance		
	Provision of Protective Clothing		
35.02	<u> </u>		
	Areas for Wearing of Protective Clothing Damaged Attire		
35.04	Damaged Aufre	•••••	33
SDECI	AL LEAVE	22	27
22.01	Special Leave Without Pay		
	Paid Jury or Court Witness		
22.02			
	Educational Leave		
22.04	Extended Unpaid Leave		
22.05	Deferred Salary Leave Plan	•••••	38
CT A E	E DEVEL ODMENT	77	15
	F DEVELOPMENT		
27.01	List of Duties and Responsibilities		
	Orientation Program		
	In-Service Program		
	In-Service Program - Specialty Units		
	Interviewing New Employees		
27.06			
27.07	Occupational Health and Safety Committee	••••••	46
OTLANT	DDV	10	17
	DBY		
10.01	Standby Duty Rates		
	Standby Duty Rates on Designated Holiday		
	Failure to Report for Standby Duty		
10.04	Authorization of Standby Duty		18

STRIE	KES - LOCKOUTS	13	20
13.01	No Strikes or Lockouts		20
TECU	NOLOGICAL CHANGE OR NEW		
IECI	METHOD OF OPERATION	40	58
40.01	Advance Notice		
40.02	No Layoff		
40.03	Income Protection		
TEMF	PORARY ASSIGNMENT	36	53
36.01	Assignment to Higher Classification		
36.02	Assignment Exceeding Ten (10) Days		
36.03	No Additional Payment		
36.04	Registered Nurse in Charge of Unit		54
36.05	Temporary Assignment to Lower Classification		55
36.06	Team Leader		
TERM	IINATION OF EMPLOYMENT	29	47
29.01	Vacation Pay on Termination of Service		47
29.02	Letter of Intent		
29.03	Period of Notice		47
29.04	Pay on Termination		47
TERM	IS OF AGREEMENT	51	67
51.01	Duration		67
	Notice of Change		
51.03	•		
TRAN	ISFER AND PORTABILITY	39	57
39.01	Benefits		57
39.02			
39.03	Re-employment After Termination - Casual Employe		
UNIO	N DUES	7	7
7.01	Union Dues		7
7.02	Deductions		8
7.03	T4 Slips		8
UNIO	N REPRESENTATION	6	6
6.01	Names of Representatives/Shop Stewards		
6.02	Leave for Union Business		
6.03	Leave of Absence for Full-Time Union Representative		
USE (OF AUTOMOBILE	37	55
37.01	Reimbursement for Use of Automobile		55
37.02	Nurses' Automobile Expenses		55
	Meal Rates		

37.04	Private Accommodations	56
VACA	ATION161	26
16.01	Length of Vacation	
16.02	Length of Vacation (Public Health Nurses)	
16.03	Vacation Period	
16.04	Selection of Vacation Dates	
16.05	Choice of Vacations	
16.06	Annual Leave Pay	
16.07	Annual Leave Commencement	
16.08	Refund of Overdrawn Vacation Pay	27
16.09	Anticipated Vacation	27
16.10	Maximum Annual Leave Accumulation	28
16.11	Calculation of Length of Vacation	28
16.12	Substitution of Vacation for Sick Leave	
16.13	Substitution of Vacation for Compassionate Leave	29
16.14	Change of Vacation Entitlement	29
16.15	Vacation Leave on Resignation or Transfer	29
16.16	Vacation Credits for the First and Last Month of Employment	29
16.17	Compensation for Work Performed During Vacation	29
16.18	Leave of Less Than One Week – Annual Leave,	
	Compensatory Time Off or Rescheduled Statutory Holiday	29
WORI	KERS' COMPENSATION38	56
38.01	Scope of Workplace Health, Safety and Compensation Act	56
38.02	Pending Settlement	
38.03	Retention of Benefits	56
38.04	Notice of Injury	56
38.05	Rehabilitation	57
	Extended Earnings Loss	57
WOR	KPLACE VIOLENCE AND SEXUAL AND PERSONAL	
	HARASSMENT47	63
47.01	Workplace Violence	
47.02	Sexual and Personal Harassment	64